The Islamic University of Gaza Deanery of Graduate Studies Faculty of Engineering Construction Management Program الجامعة الإسلامية – غزة عمادة الدر اسات العليا كلية الهندسة برنامج ادارة التشييد

A CONSTRUCTION MATERIALS MANAGEMENT SYSTEM FOR GAZA STRIP BUILDING CONTRACTORS

Submitted By Eyad Abed El-Qader Al Haddad

Supervised By Dr. Kamalain Sha`ath

A Thesis Submitted in partial Fulfillment of the Requirements for the Degree of Master Of Science in Construction Management

> Gaza, Palestine December 2006



Dedication

I would like to dedicate this to my wife and family for their infinite and kind support

Eyad Al Haddad



www.manaraa.com

ACKNOWLEDGMENT

- I am grateful to my supervisor Dr. Kamalain Sha`ath for his professional advice, useful guidance, and excellent support through all stages of preparing this thesis. Dr sha`ath`s careful check and useful response have made a great contribution to the production of this thesis in its final form.
- My deep thanks to Prof. Adnan Enshassi and Dr. Rifat Rustom for their sincere support and help.
- Careful acknowledgment to my colleagues Eng. Magdy Mohesen for his advices during the study and for his assistance in preparing the system.
- ✤ My grateful thanks to all contractors who participated in filling questionnaires and provided important information for this study.



Abstract

A Construction Materials Management System for Gaza Strip Building Contractors

Effective construction materials management process is a key to success of a construction project. Nowadays, successful management of construction materials has to be based on thorough and updated information, and processed utilizing a well designed construction materials management software.

The aim of the thesis has been to explore the local practice in construction materials management and develop a construction materials management system to facilitate the management of construction materials mainly in the building construction. Construction materials management related literature has been generally reviewed; meanwhile some construction materials management software packages have been reviewed also.

A survey questionnaire supported by interviews is used to explore the local practice in construction materials management. One hundred and twenty questionnaires were distributed to contractors of first; second, and third class, eighty-four questionnaires were received and analyzed.

The researcher concluded that all contracting companies are interested in using some techniques of managing construction materials such as creating and updating database for materials categories, local and international suppliers. Also, the Israeli closure on Gaza Strip is the main element that affects materials availability and cost and causes increase the cost of main materials such as cement, reinforcement steel and aggregate.

The study shows that most of contracting companies are still managing construction materials manually. Shortage of user-friendly construction materials software packages and lack of qualified personnel in using computer-based materials management systems are considered the main obstacles in using computer in construction materials management.

The researcher explores Microsoft Excel capabilities and utilizes these capabilities in developing a Construction Materials Management Software which he names "construction materials management software" (CMMS).

CMMS is a PC-based software which has been designed to run under Microsoft Windows. Microsoft Excel is used in developing CMMS, as most companies in Gaza strip are familiar with it. Full description of CMMS has been given with detailed implementation procedures. CMMS has been evaluated to test its suitability to local practice. Evaluation of CMMS has addressed both conceptual and practical issues. One of the main recommendations of this research is to encourage local contracting companies to have a construction materials management software package and use it in determining the required quantities of construction materials in order to get materials in time and required quantities, save time and minimize error.



الملخص

نظام إدارة مواد التشييد لمقاولى البناء في قطاع غزة

الإدارة الفاعلة لمواد التشييد لأي مشروع يعتبر احد العوامل المهمة في نجاح هذا المشروع، في الوقت الحالي إدارة مواد التشييد الناجحة هي التي تعتمد على معلومات دقيقة ومحدثة تتم معالجتها بواسطة برنامج محوسب محكم لإدارة مواد التشييد.

غرض الدراسة هو التعرف على الواقع العملي المحلي لإدارة مواد التشييد في مشاريع البناء، وتطوير برنامج حاسوب متخصص في هذا المجال لكي يحسن من الممارسة المحلية في تشييد المباني، خلال البحث جرت مراجعة عامة للأدبيات المتعلقة بإدارة مواد التشييد. كما وتمت أيضا مراجعة لبعض برامج الحاسوب المتخصصة في إدارة مواد التشييد. لإجراء هذا البحث تم استخدام أسلوب الاستبانه المدعم بالمقابلة الشخصية حيث تم توزيع مائة و عشرون استبانه علي مقاولي الدرجة الأولى والثانية والثالثة وقد تم جمع أربع وثمانين استبانه منها.

أوضحت نتائج هذا البحث بان معظم المقاولين مهتمين في استخدام بعض التقنيات لإدارة مواد التشييد لديهم كالقيام بعمل قاعدة بيانات وتحديثها لأسماء المواد والموردين لهذه المواد. كما واستنتج الباحث بأن الاغلاقات الأمنية لقطاع غزة تؤثر على وفرة المواد وأسعارها.

أيضا أوضحت النتائج أن معظم المقاولين ما زالوا يقومون بإدارة مواد التشييد يدويا أو حتى ذهنيا، وأن معظم العوائق التي تحول دون استخدام برامج إدارة المواد المتخصصة هي نقص في البرامج سهلة الاستخدام والمتخصصة في إدارة المواد، وعدم توفر الخبراء في استخدام هذه البرامج. قام الباحث بتطوير برنامج حاسوب لإدارة مواد التشييد وسماه برنامج إدارة مواد التشييد (CMMS) "Construction Materials Management Software" الموجودة في برنامج مايكروسوفت اكسل.

برنامج إدارة مواد التشييد CMMS : مصمم للاستخدام بواسطة الحاسوب الشخصي وهو يعمل قي بيئة النوافذ (ويندوز)، كما تم استخدام برنامج (مايكروسوفت اكسل) في تطويره حيث أن معظم مقاولي قطاع غزة لديهم خبرة جيدة في استخدامات "الإكسل "، وشملت هذه الرسالة وصف كامل للبرنامج مع مثال توضيحي، هذا وتم عمل تقويم للبرنامج للتأكد من ملائمة للواقع المحلي من حيث الفكرة والتطبيق.

كان من التوصيات الرئيسية لهذا البحث، تشجيع المقاولين المحليين لاقتناء برنامج محوسب متخصص في إدارة مواد التشييد واستخدامه في تحديد الكميات اللازمة من المواد للحصول عليها في الوقت المناسب والكميات المناسبة وتوفير الوقت وتقليل الأخطاء.



TABLE OF CONTENTS

Abstract	i
الملخص	ii
TABLE OF CONTENTS	iii
LIST OF TABLES	viii
LIST OF FIGURES	X

CHAPTER 1

INTRODUCTION

1.1	General	1
1.2	Rationale	1
1.3	Research aim	2
1.4	Research objectives	2
1.5	Methodology outline	2
1.6	Thesis contents	3

CHAPTER 2

CONSTRUCTION MATERIALS MANAGEMENT

2.1	Introduction	4
2.2	Background	4
2.3	Importance of materials for a project	6
2.4	What is material management	8
2.5	Need for material management systems	12
2.6	Goals of material management	13
2.7	Benefits of material management	14
2.8	Scope of materials management	15
2.9	Material management steps	16
2.9.1	Request for Quotation (RFO)	17
2.9.2	Purchasing	18



2.9.3	Expediting	18
2.9.4	Transportation	19
2.9.5	Surplus materials	20
2.10	Responsibilities	21
2.11	Legal aspects of document	22
2.12	Just In Time	23
2.12.1	Construction JIT vs. manufacturing JIT	24
	Using JIT to reduce variation and waste: manufacturing vs.	25
2.12.2	construction	
2.12.3	Types of construction buffers	25
2.12.4	Functions of schedule buffers	26

CHAPTER 3

METHODOLOGY

3.1	Introduction	27
3.2	Research strategy	27
3.3	Research design	28
3.4	Limitation of the research	28
3.5	Data collection and questionnaire design	28
3.6	Field work research	29
3.7	Pilot study	31
3.8	Interview	31
3.9	Population and sample	31
3.10	Data coding and data analysis	32
3.11	Developing and evaluating the computerized system	32

CHAPTER 4

SYRVEY RESULTS

4.1	Introduction	34
4.2	Study population characteristics	34



4.2.1	Year of establishment	34
4.2.2	Field of work	35
4.2.3	Classification of contractors	35
4.2.4	Number of employees and their qualifications	35
4.2.5	The number and value of executed projects during the last five years	36
	Distribution of respondent's person in-charge-of managing materials in	
4.2.6	construction projects	37
4.3	Application of construction materials management techniques in	
	construction projects	38
4.3.1	The necessity, degree of usage, and method of use of some techniques in	
	construction materials management	39
4.4	The effect of closure of Gaza strip on materials availability and price	41
4.5	Using software for supporting the construction materials management	42
4.5.1	Respondent's efficiency of using popular computer software	42
4.5.2	Obstacles that are facing the local companies in using a construction	
	materials management software	43
4.5.3	Contractors` opinions regarding willingness to get materials	
	management packages	44
4.6	Implementation of construction materials management systems	44
4.6.1	Benefits of implementation of materials management on construction	
	projects	44
4.6.2	Importance of materials management systems to solve some problems.	45
4.6.3	The responsibility of ordering construction materials	46
4.6.4	The techniques used for ordering materials from suppliers	46
4.6.5	Available covered and open storages	47
4.6.6	Materials that are stored in the company's storages	47
4.6.7	The importance of some factors in qualifying suppliers	48
4.6.8	Frequency of contractor problems at qualifying suppliers	49
4.6.9	The importance of knowing waste percentage for different building	
	materials	49
4.6.10	Material waste on site	50



V

	a	On site practice	50
	b	Materials handling	51
	c	Material transportation	52
	d	Site management	52
	e	Site supervision	53
4.7		Conclusions	55

CHAPTER 5

CONSTRUCTION MATERIALS MANAGEMENT SOFTWARE (CMMS)

5.1	Introduction	58
5.2	Concepts	58
5.2.1	Computerized system	59
5.2.2	Excel environment	59
5.3	Software description and components	60
5.3.1	Start sheet	60
5.3.2	Main menu	61
5.3.2.1	Part one (Input data and basic calculations)	61
5.3.2.2	Part Two (Material Purchase Decisions)	71
5.3.2.3	Part Three (Materials Card)	76
5.3.2.4	Part 4 (Waste Control)	77
5.3.2.5	Help and Method of Use	79
5.4	CMMS Evaluation	80
5.4.1	Evaluation objectives	80
5.4.2	Evaluation methodology	80
5.4.3	Evaluation result and discussions	80
5.5	Software Limitations	83

CHAPTER 6

CONCLOUSION AND RECOMMENDATIONS



6.1	Introduction	84
6.2	Conclusions	84
6.3	Recommendations to the parties involved in construction	86
6.4	Recommendations for further studies	86
	References	88

ANNEXES

ANNEX 1: The questionnaire (In Arabic)	92
ANNEX 2: The questionnaire (English Version)	103
ANNEX 3: A review of Construction Materials Management Software	
(CMMS)	113
ANNEX 4: A sample of Activities List	135
ANNEX 5: System Evaluation Questionnaire (In Arabic)	137
ANNEX 6: System Evaluation Questionnaire (English Version)	142
ANNEX 7: Recording Incoming of Construction Materials Device	147
ANNEX 8: The Average waste percentage of construction material in	
different countries	148
ANNEX 9: Help and Method of Use	150



LIST OF TABLES

Table 2.1: Classification of Materials	6
Table 3.1: Reliability coefficient for the questionnaire	31
Table 4.1: Year of establishment of contracting companies.	34
Table 4.2: Field of company specialization	35
Table 4.3: Degree of classification	35
Table 4.4: Qualification of employees in the contracting companies	36
Table 4.5: Distribution of respondents' number of employee	36
Table 4.6: Distribution of number of executed projects	37
Table 4.7: Distribution of value of executed projects	37
Table 4.8: Distribution of respondents' person in charge of managing	
construction materials	38
Table 4.9: Technique necessity and usage in managing construction materials.	39
Table 4.10: The necessary, degree of usage, and method of use some techniques	
in construction materials management	40
Table 4.11: The effect of closure of Gaza strip on materials availability and	
Price	41
Table 4.12: Using software for supporting the construction materials	
Management	42
Table 4.13: Efficiency of respondents in using popular computer software	42
Table 4.14: The obstacles which are facing contractors in using construction	
materials management systems	43
Table 4.15: Contractors` willingness to get materials management packages	44
Table 4.16: Benefits of implementation of materials management software on	
construction projects	45
Table 4.17: The importance of implementing a materials management system.	46
Table 4.18: The responsible about ordering construction materials	46
Table 4.19: Techniques for ordering construction materials from suppliers	47
Table 4.20: Available covered and open storages	47
Table 4.21: How often contracting companies store construction materials	48
Table 4.22: The factors which are affecting qualifying suppliers	48



Table 4.23: Problems which are facing the companies when qualifying	
suppliers	49
Table 4.24: The importance of knowing materials waste percentage	49
Table 4.25: Factors that increase waste on the construction site (Group a)	51
Table 4.26: Factors that increase waste on the construction site (Group b)	51
Table 4.27: Factors that increase waste on the construction site(Group c)	52
Table 4.28: Factors that increase waste on the construction site (Group d)	53
Table 4.29: Factors that increase waste on the construction site (Group e)	54
Table 5.1: The contractors` responses to performance of CMMS tools and	
Techniques	81
Table 5.2: The contractors` responses to the features of CMMS design and	
structure	82
Table A8: The Average waste percentage of construction material in different	
Countries	149



LIST OF FIGURES

Figure 2.1 Typical material management in construction	11
Figure 2.2: Relationship of Purchasing/Procurement/supply management with	
material management	13
Figure 2.3: Material Management steps	17
Figure 2.4: Relationships and key documents	22
Figure 2.5: Types of construction buffers	26
Figure 3.1: Methodology flow chart	30
Figure 5.1: Construction Materials Management System (CMMS)	59
Figure 5.2: Start sheet	60
Figure 5.3: "Main Menu" sheet	61
Figure 5.4: "Company and project information" sheet	62
Figure 5.5: "Activity Pool" sheet	63
Figure 5.6: "Materials Pool" sheet	64
Figure 5.7: "Project Activities" sheet	65
Figure 5.8: "Activities Materials Quantities' sheet	66
Figure 5.9: "Materials needed" sheet	67
Figure 5.10: "Materials to Order between two Dates" sheet	68
Figure 5.11: "Materials Spaces and Cash Requirements" sheet	69
Figure 5.12: "Earned value S-curve calculation" sheet	70
Figure 5.13: "Earned value S-curve" sheet	70
Figure 5.14: "Materials to order and deliver on site" sheet	71
Figure 5.15: "Materials must be purchased at" sheet	72
Figure 5.16: "Materials Spaces Assignment" sheet	73
Figure 5.17: "Materials Prices Assignment" sheet	74
Figure 5.18: "Materials Cumulative Spaces & Prices at Purchase Order" sheet	74
Figure 5.19: "Purchase Order" sheet	75
Figure 5.20: "Materials must be on the site at" sheet	76
Figure 5.21: "Materials Card" sheet	77
Figure 5.22: "Waste control" sheet	78



Figure 5.23: "Help and method of use" sheet	79			
Figure A3.1: a sample of CMMS Files	114			
Figure A3.2: a sample of Main Menu sheet				
Figure A3.3: a sample of Company and project information Sheet				
Figure A3.4: a sample of Materials Pool Sheet				
Figure A3.5: a sample Activity Pool of Sheet				
Figure A3.6: a sample of Project Activities Sheet				
Figure A3.7: a sample of Activities Materials Quantities				
Figure A3.8: a sample of Materials Needed sheet				
Figure A3.9: a sample of Materials to Order between Two Dates sheet.				
Figure A3.10: sample of Materials Spaces and Cash Requirements				
Sheet	123			
Figure A3.11: a sample of Earned value S-curve calculation sheet	124			
Figure A3.12: a sample of Earned value S-curve.	125			
Figure A3.13: a sample of Materials to order and deliver on site sheet	126			
Figure A3.14: a sample of Materials Must be purchased at sheet				
Figure A3.15: a sample of Materials Spaces Assignment sheet	128			
Figure A3.16: a sample of Materials Prices Assignment sheet				
Figure A3.17: a sample of Materials Cumulative Spaces & Prices at				
P.O.D. sheet	130			
Figure A3.18: a sample of Purchase order sheet	131			
Figure A3.19: a sample of Materials Must be on the site at Sheet	132			
Figure A3.20: a sample of Materials Card sheet	133			
Figure A3.21: a sample of Waste control sheet	134			
Figure A4.1 a sample of Activities List				
Figure A7: Recording incoming of construction materials device	147			



CHAPTER 1 INTRODUCTION

1.1 General

In this research, the existing construction materials management practices of Gaza Strip contracting companies are investigated, and an attempt to improve it is conducted. The proposed improvements are formulated in development of a computerized materials management system. In this chapter, the rationale, research aim, research objectives, and outline methodology of the research are explained. The contents of this thesis are also summarized.

1.2 Rationale

Construction projects can be accomplished utilizing management processes. These processes include planning, organizing, executing, monitoring, and controlling (Ahuja et al 1994). During any construction project the three inter-related factors of time, money, and quality need to be controlled and managed. Successful completion of projects requires all resources to be effectively managed. Materials management is considered as a means to achieve better productivity, which should be translated into cost reduction.

Computer software systems are widely used in construction management in the developed countries. Yet, using computer applications is still in its early phase of development in the developing countries. Therefore, it is logical that computerization has emphasis in the process of improvement of project management practices in the developing countries. The majority of local contractors in Gaza Strip have poor experience in managing construction materials. Materials management software systems are not quite available and there is a shortage in qualified persons to use these software packages. The results obtained from studying the local practices in managing construction materials management system in order to improve the capabilities of local contractors in handling materials management functions.



To the researcher knowledge, a little research effort has been done to investigate the construction materials management practices of Gaza Strip contracting companies. It is important to explore and evaluate these practices, and then offer recommendations and develop practical techniques to improve existing practices.

1.3 Research Aim

The aim of this research is to explore the existing common practices in construction material management for the building construction projects in Gaza strip. Also the research aims at developing a computerized materials management system that suites and, hopefully, improves the local practices.

1.4 Research Objectives

The aim of this research can be broken down into the following objectives:

- to review literature related to the construction material management, and also to review the relevant software packages.
- to investigate the local practices of construction material management in contracting companies in Gaza strip.
- to explore the influence of the Israeli closure on material prices, materials availability, and materials management.
- to assess the impact of computerization on construction materials management.
- to develop a computerized system in construction materials management to improve the common existing practices.
- to evaluate the system by experienced contractors.

1.5 Methodology Outline

The research is conducted through the following stages:

First stage: Literature Review

Materials management related literature was reviewed to identify the main topics and concepts that related to this research. This stage included also a review of available material management software packages. The researcher also explored the Microsoft Excel capabilities.



Second Stage: Field Survey

A survey of the local materials management practices of contracting companies in Gaza Strip was made. A structured questionnaire was used and the person in charge of managing construction materials in the company was interviewed. Statistical analysis for questionnaires was done by using Statistical Package for the Social Sciences (SPSS). Discussion for the obtained results was also made.

Third Stage: Model Formulation and Evaluation

A material management computerized system was developed based on the results obtained from the field survey, the review of material management software packages, and the exploring of the Ms Excel capabilities. This system was evaluated by experienced contractors after applying it on ongoing projects.

1.6 Thesis Contents

Apart from this chapter, there are other five chapters and nine annexes. Chapter 2 presents the literature review of material management in construction projects. Chapter 3 presents the Methodology. Chapter 4 presents research results. It includes the questionnaire design, the method of analysis, and analysis of the surveyed results and discussion of these results. Construction materials management software is discussed in detail in Chapter 5. The discussions include the concepts, the description, the implementation, and evaluation of the material management system. Chapter 6 presents conclusions, recommendations for the main parties involved in the construction industry, and recommendations for further studies.

There are nine Annexes, which supplement these chapters. They are:

Annex 1: The questionnaire (In Arabic).

Annex 2: The questionnaire (English Version).

Annex 3: A review of construction materials management software.

Annex 4: A sample of Activities List.

Annex 5: The system evaluation questionnaire (In Arabic).

Annex 6: The system evaluation questionnaire (English Version).

Annex 7: Recording incoming of construction materials device.

Annex 8: The Average waste percentage of construction material in different countries.

Annex 9: Help and method of use



CHAPTER 2 CONSTRUCTION MATERIALS MANAGEMENT

2.1 Introduction

Construction materials constitute a major cost component in any construction project. The total cost of installed materials (or value of materials) may be 50% or more of the total cost (Stukhart 1995, Bernold and Treseler 1991), even though the factory cost may be a minor part of the total, probably less than 20-30%. This is because the manufactured item must be stored, transported, and restored before it is put in place or "consumed" at the site. The total cost of materials will include, in addition to the manufacturer selling cost, the cost of procurement (cost of placing processing and paying the material, physical distribution, the distributor's cost, and the transportation of materials), and the site-handling costs (cost of receiving, storage, issuing, and disposal). The efficient procurement and handling of material represent a key role in the successful completion of the work. It is important for the contractor to consider that there may be significant difference in the date that the material was requested or date when the purchase order was made, and the time at which the material will be delivered. These delays can occur if the contractor needs a large quantity of material that the supplier is not able to produce at that time or by any other factors beyond his control. The contractor should always consider that procurement of materials is a potential cause for delay (Willis, 1986).

Poor planning and control of materials, lack of materials when needed, poor identification of materials, re-handling and inadequate storage cause losses in labor productivity and overall delays that can indirectly increase total project costs. Effective management of materials can reduce these costs and contribute significantly to the success of the project.

2.2 Background

The Webster's dictionary defines materials as "the elements, constituents, or substances of which something is composed or can be made." Ballot (1971) defines materials as the physical materials that are purchased and used to produce the final product and does not suggest that materials are the final product. In other words, materials are the parts used to produce the final product.



Bailey and Farmer (1982) define materials as the goods purchased from sources out of the organization that are used to produce finished products. Stukhart (1995) defines materials as the items that are used to produce a product and which include raw materials, parts, supplies and equipment items.

Chandler (1978) states that construction materials can be classified into different categories depending on their fabrication and in the way that they can be handled on site. He classifies the materials into five categories. They are:

- Bulk materials- these are materials that are delivered in mass and are deposited in a container.
- Bagged materials- these are materials delivered in bags for ease of handling and controlled use.
- Palleted materials- these are bagged materials that are placed in pallets for delivery.
- Packaged materials- these are materials that are packaged together to prevent damage during transportation and deterioration when they are stored.
- Loose materials- these are materials that are partially fabricated and that should be handled individually. Table 2.1 presents some examples of commonly used materials in construction and their classification.

Material	Bulk	Bagged	Palleted	Packaged	Loose
Sand	Х				
Gravel	Х				
Topsoil	Х				
Paving Slabs					Х
Structural Timber					Х
Cement	Х	Х	Х		
Concrete	Х				
Pipes				Х	Х
Tiles				Х	
Doors			Х		
Electrical Fittings				X	

 Table 2.1: Classification of Materials (Adopted from Chandler, 1978)

Stukhart (1995) states that the main categories of materials encountered in a construction project are engineered materials, bulk materials, and fabricated materials.

• Bulk materials- these are materials manufactured to standards and are purchased in quantity. They are bought in standard length or lot quantities.



Examples of such materials includes pipes, wiring, and cables. They are more difficult to plan because of uncertainty in quantities needed.

- Engineered materials- these materials are specifically fabricated for a particular project or are manufactured to an industry specification in a shop away from the site. These materials are used for a particular purpose. This includes materials that require detailed engineering data.
- Fabricated materials- these are materials that are assembled together to form a finished part or a more complicated part. Examples of such materials include steel beams with holes and beam seats.

Stukhart (1995) defines material management as the activities involved to plan, control, purchase, expedite, transport, store, and issue in order to achieve an efficient flow of materials and that the required materials are bought in the required quantities, at the required time, with the required quality and at an acceptable price.

Plemmons and Bell (1995) define material management as the plan and control of all activities to ensure the correct quality and quantity of materials and equipment to be installed as specified in timely manner, obtained at reasonable cost and are available when needed.

Dobler and Burt (1996) state that material management is designed to improve the activities related to the flow of materials. They add that material management should coordinate purchasing, inventory control, receiving, warehousing, materials handling, planning, and transportation.

2.3 Importance of Materials for a Project

Problems related to managing the flow of materials can be found in every organization. The efficient management of materials plays a key role in the successful completion of a project. The control of materials is a very important and vital subject for every company and should be handled effectively for the successful completion of a project. Materials account for a big part of products and project costs. The cost represented by materials fluctuates and may comprise between 20-50% of the total project cost and sometimes more. Some studies concluded that materials account for around 50-60% of the project cost (Stukhart, 1995 and Bernold and Treseler, 1991). Materials are critical in the operations in every industry since unavailability of materials can stop production. In addition, unavailability of materials when needed



6

can affect productivity, cause delays and possible suspension of activities until the required material is available.

Unavailability of materials is not the only aspect that can cause problems. Excessive quantities of materials could also create serious problems to managers. Storage of materials can increase the costs of production and the total cost of any project. When there are limited areas available for storage, the managers have to find other alternatives to store the materials until they are needed. Some of these alternatives might require re-handling of materials, which will increase the costs associated with them. Provisions should be taken to handle and store the materials adequately when they are received. Special attention should be given to the flow of materials once they are procured from suppliers.

It is obvious that materials should be obtained at the lowest cost possible to provide savings to the company (Damodara, 1999). In the late 1970's, construction companies experienced an increase in costs and a decrease in productivity. Owners of these companies thought that these increases in cost were due to inflation and economic problems. Further research concluded that these companies were not using their resources efficiently and that the decrease in productivity was also attributable to poor management (Stukhart, 1995). Material management has been an issue of concern in the construction industry. 40% of the time lost on site can be attributed to bad management, lack of materials when needed, poor identification of materials and inadequate storage (Baldwin et. al, 1994).

The need for an effective materials planning system becomes mandatory. Some companies have increased the efficiency of their activities in order to remain competitive and secure future work. Many other firms have reduced overheads and undertaken productivity improvement strategies. Considerable improvement and cost savings would seem possible through enhanced materials management. Timely availability of materials and systems are vital to successful construction. Materials management functions are often performed on a fragmented basis with minimal communication and no clearly established responsibilities assigned to the owner, engineer or contractor.

Better material management practices could increase efficiency in operations and reduce overall cost. Top management is paying more attention to material



management because of material shortages, high interest rates, rising prices of materials, and competition. There is a growing awareness in the construction industry that material management needs to be addressed as a comprehensive integrated management activity.

2.4 What is Material Management?

Different researchers provide different definitions for material management. Therefore, different definitions can be found in different references. Basically, material management is concerned with the planning, identification, procuring, storage, receiving and distribution of materials. The purpose of material management is to assure that the right materials are in the right place, in the right quantities when needed. The responsibility of one department (i.e. material management department) for the flow of materials from the time the materials are ordered, received, and stored until they are used is the basis of material management.

Ballot (1971) defines material management as the process of planning, acquiring, storing, moving, and controlling materials to effectively use facilities, personnel, resources and capital. Tersine and Campbell (1977) define material management as the process to provide the right materials at the right place at the right time in order to maintain a desired level of production at minimum cost. The purpose of material management is to control the flow of materials effectively. Beekman-Love (1978) states that a material management structure should be organized in such a way that it allows for integral planning and coordination of the flow of materials, in order to use the resources in an optimal way and to minimize costs.

Chandler (1978) states that material management systems should be implemented to plan, order, check deliveries, warehousing, controlling the use of materials, and paying for materials. He adds that these activities should be interrelated. Ammer (1980) defines material management as the process in which a company acquires the materials that it needs to achieve their objectives. This process usually begins with the requisition of materials from the supplier until the material is used or incorporated into a product. Bailey and Farmer (1982) define material management as a concept concerned with the management of materials until the materials have been used and converted into the final product. Activities include cooperation with designers,



purchasing, receiving, storage, quality control, inventory control, and material control. Gossom (1983) indicates that a material management system should have standard procedures for planning, expediting, transportation, receipt, and storage to ensure an efficient system for materials control. Cavinato (1984) states that material management involves the control of the flow of goods in a firm. It is the combination of purchasing with production, distribution, marketing and finance. Arnold (1991) states that material management is a function responsible for planning and controlling of materials flow. He adds that a materials manager should maximize the use of resources of the company.

Materials management is an important element in project planning and control. Materials represent a major expense in construction, so minimizing procurement or purchase costs presents important opportunities for reducing costs. Poor materials management can also result in large and unavoidable costs during construction. First, if materials are purchased early, capital may be tied up and interest charges incurred on the excess inventory of materials. Even worse, materials may deteriorate during storage or be stolen unless special care is taken. For example, electrical equipment often must be stored in waterproof locations. Second, delays and extra expenses may be incurred if materials required for particular activities are not available. Accordingly, insuring a timely flow of material is an important concern of project managers. Materials management is not just a concern during the monitoring stage in which construction is taking place. Decisions about material procurement may also be required during the initial planning and scheduling stages. For example, activities can be inserted in the project schedule to represent purchasing of major items such as elevators for buildings (Dubler and Burt, 1996).

The availability of materials may greatly influence the schedule in projects with a fast track or very tight time schedule. Sufficient time for obtaining the necessary materials must be allowed. In some cases, more expensive suppliers or shippers may be employed to save time. Materials management is also a problem at the organization level if central purchasing and inventory control is used for standard items. In this case, the various projects undertaken by the organization would present requests to the central purchasing group. In turn, this group would maintain inventories of standard items to reduce the delay in providing material or to obtain lower costs due to bulk purchasing (Cavinato, 1994).



This organizational materials management problem is analogous to inventory control in any organization facing continuing demand for particular items. Materials ordering problems lend themselves particularly well to computer based systems to insure the consistency and completeness of the purchasing process. In the manufacturing realm, the use of automated materials requirements planning systems is common. In these systems, the master production schedule, inventory records and product component lists are merged to determine what items must be ordered, when they should be ordered, and how much of each item should be ordered in each time period. The heart of these calculations is simple arithmetic: the projected demand for each material item in each period is subtracted from the available inventory.

When the inventory becomes too low, a new order is recommended. For items that are non-standard or not kept in inventory, the calculation is even simpler since no inventory must be considered. With a materials requirement system, much of the detailed record keeping is automated and project managers are alerted to purchasing requirements (Stukhart, 1995).

The role that a materials manager plays in an organization is strictly economical since the materials manager should keep the total cost of materials as low as possible. The person in charge of handling materials should keep in mind the goals of the company and insure that the company is not paying extra money for materials. The goal of every company is to make a profit. This is the basis for company survival, costs should not exceed income, but keeping in mind customer's expectations.

The typical tasks associated with a material management system are (Tersine and Campbell (1977), Ammer (1980), Stukhart (1995)):

- Procurement and purchasing
- Expediting
- Materials planning
- Materials handling
- Distribution
- Cost control
- Inventory management / Receiving/ Warehousing
- Transportation

Purchasing and procurement deals with the acquisition of materials to be used in the



operations. The primary function of purchasing and procurement is to get the materials at the lowest cost possible, but keeping in mind quality requirements. Expediting is the continuous monitoring of suppliers to ensure on time deliveries of materials purchased. The purpose of materials planning is to procure the materials for the dates when they are needed, storage facilities, and handling requirements. The primary function of materials handling is to manage the flow of materials in the organization. The manager has to assure that the costs associated with handling materials are kept to a minimum. In cost control, the manager has to insure that the costs to buy materials are kept to a minimum. In other words, the manager has to insure that he is buying the products at the lowest possible price. The inventory management deals with the availability of materials. Transportation involves using the safest most economical means to transport the materials to the site where they are needed.

Figure 2.1 depicts the different phases of the material management process including the relationship and interdependency between the different activities in each phase. From this figure it can be seen that decisions taken at each phase in the system, directly affect the activities of the phases that follow.



Figure 2.1: Typical Materials Management in Construction (Source: Thabet , 2001)



2.5 Need for Material Management Systems

The costs associated with material management are hidden in other activities or included as overhead costs. Stukhart (1995) states that studies from the Construction Industry Cost Effectiveness Project (CICEP) concluded that senior management have not recognized the contribution of material management to cost issues in projects, that personnel involved in material management activities do not receive an adequate training, and that the computer systems used by companies are not good sources of information for materials control. Historically, managers had paid more attention to the costs associated with personnel, equipment and plant and little attention has been given to materials. For manufacturing organizations, the costs related to materials have increased and had become the largest expenditure of the organization; therefore more attention has been placed into activities related to materials (Tersine, 1978). The cost of materials has escalated to twice the cost of labor between 1975 and 1980 inducing companies to pay more attention to activities related to materials (Bernold and Treseler, 1991).

Navon 2002 developed and evaluated an automated model for management and control of materials ordering, purchasing, and supply and use. In order to evaluate the model under real conditions, the model was implemented in a prototype system and used in ongoing construction projects. The model provides a comprehensive approach, encompassing materials purchasing aspects, their delivery to the site and their dispatching for use in the building. The model can reduce the time needed for materials management, reduce wastage caused by manually ordering the materials and ensure that materials on site on time, in the right quantity and according to specifications.

Figure 2.2 illustrates a typical flow of materials, and material activities in an organization. From the figure, it can be seen that decisions taken at early stages in the material management flow might affect other activities and decisions to be made in later stages. For example, if the proposals from suppliers are not analyzed (i.e. step 6 in the purchasing activities), then the selection of suppliers might be affected (i.e. step 3 in the supply management activities).

Coordination is needed in order to reduce the impact that a decision at a certain stage might have in other activities. Communication is essential among members of the team to avoid conflicts and to take the better decisions regarding materials flow.



12



Figure 2.2: Relationship of Purchasing/Procurement/supply management with material management (Adopted from Dobler and Burt, 1996)

2.6 Goals of Material Management

As it was mentioned previously, the role of the materials manager is strictly economical within an organization. This section will describe some of the aspects that the materials manager should keep in mind to handle all activities related to materials appropriately.

Cavinato (1984) states that the objectives of a material management system should include lowest final cost, optimum quality, assurance of supply, and lowest administrative costs. The materials manager should obtain the materials needed at the lowest cost possible.

By buying products at the lowest possible costs, operating costs can be reduced and profits can be increased. Proper handling and storage of materials can reduce the total cost of materials; therefore the materials manager should ensure that materials are handled properly and stored in the most adequate places. Quality is a very important aspect that the materials manager has to keep in mind. When specifications require a



high quality product, quality could become the most important objective. Suppliers play an important role in any organization. Many companies rely greatly on outside suppliers for the materials needed for production. Good relations with suppliers might be decisive for a company to be in business.

Companies that have good relations with suppliers could be more successful in attracting customers than companies that have bad relations with suppliers. When a company has good relations with its suppliers it could benefit from cost reductions, cooperative environment from the employees of the supplier, and willingness to help with materials ordered and orders pending. When a company has bad relation with their suppliers it might be possible that it experiences late deliveries or wrong materials delivered. This will have an impact on the total cost of the product, possibly increasing the total costs, and delaying the completion of the final product. Materials acquisition from the procurement time until it is received in the field can have a significant impact on the schedule of a construction project. Based on the studies presented, it is clear that effective management of materials could have on the overall schedule and cost of the project. The materials manager should assure that effective and economical transportation are used to transport materials to the site.

2.7 Benefits of Material Management

An effective material management system can bring many benefits for a company. Previous studies by the Construction Industry Institute (CII) concluded that labor productivity could be improved by six percent and can produce 4-6% additional savings (Bernold and Treseler, 1991). Among these benefits are:

- Reducing the overall costs of materials
- Better handling of materials
- Reduction in duplicated orders
- Materials will be on site when needed and in the quantities required
- Improvements in labor productivity
- Improvements in project schedule
- Quality control
- Better field material control



- Better relations with suppliers
- Reduce of materials surplus
- Reduce storage of materials on site
- Labor savings
- Stock reduction
- Purchase savings
- Better cash flow management

From a study of twenty heavy construction sites, the following benefits from the introduction of materials management systems were noted (Stukhart and Bell, 1987):

- In one project, a 6% reduction in craft labor costs occurred due to the improved availability of materials as needed on site. On other projects, an 8% savings due to reduced delay for materials estimated.
- A comparison of two projects with and without a materials management system revealed a change in productivity from 1.92 man-hours per unit without a system to 1.14 man-hours per unit with a new system. Again, much of this difference can be attributed to the timely availability of materials.
- Warehouse costs were found to decrease 50% on one project with the introduction of improved inventory management, representing a savings of \$ 92,000. Interest charges for inventory also declined, with one project reporting a cash flow savings of \$ 85,000 from improved materials management.

Against these various benefits, the costs of acquiring and maintaining a materials management system has to be compared. However, management studies suggest that investment in such systems can be quite beneficial.

2.8 Scope of Materials Management

This section deals mainly with the attributes of material management and the responsibilities of those involved in carrying out the material management functions.



A detailed understanding of each contributing function is required in order to comprehend the interfaces between material management functions. A materials management system includes the major functions of identifying, acquiring, distributing, and disposing of materials on a construction site (Stukhart, 1995).

By definition, material management is the management system for planning and controlling all of the efforts necessary to ensure that the correct quality and quantity of materials are properly specified in a timely manner, are obtained at a reasonable cost, and are available at the point of use when required. Each firm has its particular materials management system where the responsibility for the various activities is spread between engineering, purchasing, and construction. Some assign full responsibility and accountability to a material manager, but for most firms the responsibility is divided and therefore prone to problems (Stukhart, and Bell, 1987).

Figure 2.3 represents the logical steps in the process from identifying material needs to delivering the materials when required at the point of use. Each step is a link in the chain of events and the strength of the chain is only as good the weakest link. From review of these steps it is apparent that the system can break down in numerous places as the result of misdirected effort or lack of effort by the many individuals involved. These steps resulting delays in delivery. It always costs more if things are not done right the first time. Delays and additional effort are costly, reduce efficiency of a construction operation because of reallocation of resources, and therefore negatively impact productivity. The steps shown are only the key element and are part of the whole material management process which by definition includes the planning and controlling of all supporting efforts (Ahuja et al, 1994).

2.9 Material Management Steps

There are several steps within the scope of material management and each of these steps can give rise to potential problems. The more the responsibility is divided, the more potential problems that exist.

Figure 2.3 shows the steps in material management and the pertinent action related to these steps. Some actions are described in terms of the documentation produced, such as receiving report and vendor data. (Ahuja et al, 1994)



Sequence	Contribution action
1-RFQ (Requisition)	Drawings, specifications.
	Material bills
	Terms and conditions
2-Bids	Approved bidders list
	Pre qualification of bidders
	Bid evaluations
3-P.O. (Purchase Order)	Bid clarification
	Notice of award
4-Expediting	Vender data
	Manufacturer inspection
	Delivery
	Routings
5-Transport	Carrier and route
	Ownership en route
	Custom
6-Receiving	Inspection and acceptance
	Receiving report
	Storage
7-Inventory	Dispersal (i.e. material handling)
	Inventory level

Figure 2.3: Material Management steps (Ahuja et al, 1994)

2.9.1 Request for Quotation (RFQ)

Specifications and drawings are needed to implement the request for quotation process successfully. The specifications and drawings are utilized by a rather diverse group of participants. The specifications and drawings help the contractor to estimate, control, manage and direct the works. Also they help the purchasing department to purchase materials and equipments that described in the drawings and specifications, finally they help the owner to know what to buy and what he is entitled to receive.

There are relations between the specifications and drawings will clear and generally show the following information and items. From drawings we can obtain information about the location of materials, equipment, fixtures, details and overall dimensions, interrelation of materials, equipment and space, sizes of equipment, identification of materials at is locations, and another alternatives. And from specifications we can obtain type and quality of materials, equipment and fixtures, quality of workmanship, methods of fabrication, installation, erecting, test and code requirements, unit, options and alternatives (Ahuja, and Dozzi 1994)



2.9.2 Purchasing

The purchasing function is central to material management. Purchasing has the responsibility and the authority to commit project funds for materials, equipment, and services. This activity may be accomplished by the home office, the field, or a combination of both depending on the size and the scope of the project. The home office must maintain planning, procedural, and policy direction over the field operations in order to ensure consistent purchasing practices. (Stukhart, and Bell, 1987)

Vendor selection follows policy and procedures as a key step in accomplishing the work. In selecting vendor for the project, purchasing is forming the foundation for the success or failure of the project. Vendors must be selected on the basis of their capabilities, geographical location, prior experience, and owner preference. Measurement of capabilities includes such considerations as past performance, financial condition, bargaining agreements, capacity, competitiveness, responsiveness, and schedule adherence. (Stukhart, and Bell, L.C, 1987)

Several methods of contracting are available to the purchasing organization, depending on the commodity or service required. Purchasing orders are the most common form of contract utilized on projects. Although blanked orders and other forms of agreement are used in varying degrees. Under any form the contract it must encourage the on time delivery and completion of the work.

"Standard" or "general" terms and conditions of the order or contract generally address various commercial aspects of transaction; they define the respective rights, duties, and obligations of the contracting parties.

Special terms and condition also must be incorporated into the body of the purchase order or contract. Items such as schedule test information, data submittals, drawing approvals, expediting, and terms of payment are typical of the information, which must be clearly specified.

Purchase orders often require technical service agreements to complete the scope of work when the vendor's technical representative is required at the site to supervise installation and or erection.

2.9.3 Expediting

Several types of expediting exist, each with a different level of intensity and cost. The least intense type of expediting is simple status reporting. Periodic telephone contact



is made with the vendor to determine the status or progress of an order, and the information is reported to the project in some systematic format. This type of expediting provides basic information to the project, but does little to prevent or overcome delays or problems with an order.

Reactive or correction expediting is more intense than the simple status reporting. But it is initiated only in response to some event or action. Vendor contact may be made in response to a problem of delayed or late delivery (Ahuja and Dozzi 1994).

Finally, proactive or preventative expediting is the most intense aggressive type of expediting. Here, vendor and sub vendor contact is initiated as soon as the order is issued and continues through the live of the order. The expeditor will review all elements of the order to ensure that the vendor understand the various submittal, testing, and delivery requirements.

The expeditor will seek to gain a thorough understanding of the vendor's engineering, purchasing, and manufacturing operations as they relate to the particular order. This enables the expeditor to monitor all elements of the vendor's performance with the intent of anticipating and resolving problems before they seriously impact the projects.

Experienced professional expeditors serve as a key bridge between the engineering and purchasing activities that specify and order materials and the field operations that are dependent on those materials for their progress. Accurate and dependable expediting information is essential for informed management of the projects, and facilitates the mobilization of buyer and vendors resources in response to problems or delays (Ahuja and Dozzi 1994)

2.9.4 Transportation

The movement of equipment, materials, and personnel to the job site represents a unique and specialization element of materials management. Experienced traffic personnel can have a positive impact on the execution of the project while minimizing transportation cost (Ahuja and Dozzi 1994)

Significant saving is possible with national agreements or negotiated project transportation, and through various commercial arrangements for the transportation of goods, materials, documentation, or personnel. Special consideration is required in setting terms, thereby determination the proper point for transfer of materials ownership and liability. The prime contract, especially insurance clauses, may have a



direct impact on the purchasing terms and conditions concerning transportation.

Early specialized activities in the project planning phases, such as properly performed route survey and consideration of local traffic conditions, can significantly affect later execution of the work. These front end efforts affect engineering by defining shipping envelopes, weight limits, and schedule limitations, the traffic function or group significant input to purchase documents including packing specifications, shipping instructions, invoicing instructions, and document requirements.

This group's expertise is necessary in developing routing guides, shipments progress reports, and troubleshooting as transportation problems develop.

Transportation or traffic expertise aids the materials management team in handling numerous types of special loads from delicate electronics to massive modules, each requiring transport equipment that is specially designed or of limited availability. Knowledge of requirements, source and availability of this equipment may be critical to successful execution of the work.

Transport permitting requirements also must be considered early in the project. Assigning the above responsibilities to suppliers may present an easy upfront decision, but can later lead to painful lessons if the expertise is not available to the materials management team to ensure that traffic functions are handled properly.

Traffic or logistics for foreign sites present an added dimension to the transportation requirements for a project. Each phase of the transportation effort is more complex, with often-stringent requirements due to ocean shipment and transportation to remote areas of the world. Each country's customs requirements are unique with potentially significant duties, taxes, and delays that must be considered in the planning efforts.

2.9.5 Surplus Materials

All projects can expect a certain amount of surplus, however, the key to successful surplus materials management is a well-conceived and well-executed materials management plan. Various shortcomings in the engineering, materials control, procurement, and field materials management phases of the work may results in surplus materials. Understanding and anticipating these potential problems areas are the first in minimizing surplus (Stukhart and June 1987)

Many causes of surplus can be identified. Surplus can be caused by a poorly



performed materials take off (MTO). Engineering revision and changes are yet another cause of surplus, particularly if the MTO occurs early and systems are not adequately responsive to changes. Inadequate construction materials management practices also may lead to surplus, particularly on fast track projects. Primary causes are:

- Duplicate buying and poor control systems/procedures leading to procurement of unnecessary materials.
- Minimizing surplus on a project requires a proactive and timely system of communication among all functions involved in the materials acquisition and installation cycle.
- Option for disposal include using the surplus in the alternative services, using the surplus materials on other projects, returning them to the vendor, or selling them to a third party. All options require complete records and timely reporting to achieve optimum results. The best option is to do the necessary planning and to implement the necessary materials management system to reduce surplus at the source(Ahuja and Dozzi 1994)

2.10 Responsibilities

The purpose of clearly establishing the responsibilities and authority of the participants is not for attaching blame should something go wrong in the process, but to communicate clearly what is expected and avoid misunderstandings as to who does what and when. The scope of each participant's involvement must be clearly defined. If not, increased effort will be expended to rectify missed expectations in quantity, quality, or cost. Unexpected effort reduces productivity of the operation. A quality effort is required in all parts of the project, otherwise poor quality in the material management process becomes apparent immediately at the point of use. By comparison, poor quality of engineering, for example, may not become apparent at all. (Ahuja and Dozzi, 1994)

Several participants contribute to the material management process and the scope of their involvement should be clearly stipulated in the contractual document. Figure 2.4 shows the contractual relationships (shown with double-ended arrows) and the key document that are used to establish the scope of material management of each participant. An efficient material management system leads to improve productivity



and must necessarily include all participants. The alternative is an inefficient, incomplete plan, which will prove counterproductive.

If an owner purchases a long-lead item and later assigns the purchase order to the contractor, a clear understanding of the purchase order is required, as well as full knowledge of any relevant correspondence, to ensure that nothing is overlooked.



Figure 2.4: Relationships and key documents (Ahuja and Dozzi, 1994)

2.11 Legal Aspects of Document

A purchase order (P.O.) is a contract, which specifies technical, delivery, warranty, and cost details for the goods and services to be provided. Besides providing technical specifications, it is important to clearly state the required delivery date and whether the goods are FOB (free on board) at the supply source or the delivery destination. Vendor information such as engineering data sheets and drawings are often required in order to finalize the engineering drawings, specifications, or both. The P.O. must clearly defined what drawings are required and when. Two examples are the final anchor bolt layout for a piece of machinery or electrical grounding locations. This information is required in order to complete foundation design and construction, and late delivery of vendor information could result in delays and additional costs to overcome the effects of delay.

Leases should be read carefully to understand the limitations respecting insurance coverage. The wording on leases leaves little doubt that the lessee assumes


responsibility for the piece of equipment or vehicle that is being leased from the lesser. The legality of contracts is readily ascertained, but it is not apparent to many involved in procurement of materials that some of the otherwise bothersome paperwork has legal implications. A bill of lading transfers responsibility for the goods from the supplier to the transporter of the goods.

Upon delivery to the receiver of the goods, at both ends of the shipping phase, the bill of lading is signed as being correct by the receiver of the goods. Signed releases by inspectors are another group of documents that allow fabrication or processing to proceed during the manufacture of items. Pressure vessels and piping are examples of fabricated is not allowed until signed approval is received from an authorized inspector. (Ahuja and Dozzi, 1994) It is commonly required to produce mill test reports for materials used in the shop fabrications or field construction. Also, reports for specific testing of materials are required to fulfill the contractual requirements for supply of some goods. In fact, materials should not be unloaded without the proper and complete documentation.

The absence or delay of the required documentation can delay fabrication progress and potentially delay the project. Besides, there are serious longer term legal implications for use of material that has not been properly certified as acceptable for use in a particular application. A dramatic example of such an occurrence is the major fire damage exceeded 100 millions of dollars.

2.12 Just In Time (JIT)

The acronym JIT has been highly visible since the late 1980's, as manufacturing attempted to meet competitive challenges by adopting newly emerging management theories and techniques, referred to by some as Lean Production (Womack and et all, 1991). Manufacturing JIT is a method of pulling work forward from one process to the next "just-in-time"; i.e. when the successor process needs it, ultimately producing throughput. One benefit of manufacturing JIT is reducing work-in-process inventory, and thus working capital. An even greater benefit is reducing production cycle times, since materials spend less time sitting in queues waiting to be processed. However, the greatest benefit of manufacturing JIT is forcing reduction in flow variation, thus contributing to continuous, ongoing improvement.



2.12.1 Construction JIT vs. Manufacturing JIT

JIT is a technique developed by Taichi Ohno and his fellow workers at Toyota (Ohno, Taichi., 1987). Ohno's fundamental purpose was to change production's directives from estimates of demand to actual demand. A purpose originally rooted in the absence of a mass market and the need to produce small lots of many product varieties. In assembly line production systems managed by lean production concepts, the directives for production are provided by means of kanban from downstream processes. This system insures that whatever is produced is throughput, i.e. is needed for the production of an order. Kanban works as a near-term adjusting mechanism within a system of production scheduling that strives for firm and stable aggregate output quantities, and provides all suppliers in the extended process progressively more specific production targets as the plan period approaches, resulting ultimately in a firm 2-6 week production schedule. This system provides sufficient flexibility to adjust to actual demand, while assuring that all resources are applied to the production of throughput. In manufacturing, the need for flexibility comes from a potential difference between forecast and actual demand. Many products are being produced, so it is important to minimize the time required to produce any specific type of product demanded. In construction, there is only one product produced once. And in the case of industrial construction, that product is the facility for producing manufacturing's products. It is consequently important to reduce the time needed to produce the facility, not necessarily the time to produce any component.

The application of JIT to construction differs substantially from its application to manufacturing because construction and manufacturing are different types of production, and because of the greater complexity and uncertainty of construction. The extent and significance of uncertainty in construction has been adequately addressed, but a moment's reflection supports the view that construction is complex. The number of parts, relative lack of standardization, and the multiple participants and constraining factors easily make the construction of an automobile factory more difficult than the production of an automobile in that factory. When this complexity is joined with economic pressures to minimize time and cost, that uncertainty results is not surprising.



24

2.12.2 Using JIT to reduce variation and waste: Manufacturing vs.

construction

By minimizing inventories between processes, Ohno removed the safety stock that allowed a downstream process to continue working when a feeder process failed. He also required that operators stop the production line when they were unable to fix problems. Consequently, it became necessary to solve problems rather than simply passing bad product down the line. Problems also became highly visible since they could result in line stoppages. Forced confrontation with problems together with analysis to root causes produced a progressively more steam lined and smoother running production process, with fewer end-of-the line defects and higher throughput. How might this work in construction? Construction is schedule driven. Given a wellstructured schedule, if everyone stays on their part of the schedule, the work flows smoothly and maximum performance is achieved. However, as we all know, it is rare that projects perform precisely to their original schedule. Business conditions change, deliveries slip, a design requires correction, etc. If a schedule has sufficient slack in the impacted activities, changes may not impact end dates. When there is little or no slack, players are pressured to make it up in accelerated production.

2.12.3 Types of Construction Buffers

There are two types of inventories that can serve the function of buffering downstream construction processes from flow variation. The most familiar type is piles of stuff; materials, tools, equipment, manpower, etc. These piles of stuff may originate in decisions to insert certain time intervals between scheduled activities, e.g. between fabrication and installation of pipe spools. Consequently, while they take the form of stuff, they often also represent time added to project duration, so these are called "schedule buffers" (Howell et. all, 1994). Less familiar are inventories of workable assignments, produced by planning processes that make work ready for downstream production (Ballard et. all, 1994). These buffer by enabling a reliable, predictable flow of output from each process. They need not imply the existence of piles of stuff, depending upon the predictability of flow between supplier and customer processes. Ballard et. all called these inventories of workable assignments "plan buffers."





BUFFERED FLOW PROCESS

Figure 2.5: Types of construction buffers (Howell et. all, 1994)

2.12.4 Functions of Schedule Buffers

In the construction of process plants (petroleum, chemical, food processing, pulp and paper, etc), projects are frequently fast track; i.e. construction begins before design is completed (Huovila et. all, 1994). Late delivery of drawings and materials has led construction contractors to demand earlier delivery, reducing the time available for engineering to complete design, resulting in more delivery problems and demands for even earlier deliveries. This is clearly a vicious circle. Large schedule buffers between suppliers and construction may shield the contractor from the impact of late deliveries (Howell et. all, 1994), but does nothing to address the root causes of variation. Further, the shielding is expensive, both in time and money. There is a better way. A suggested rule: Place schedule buffers just after processes with variable output. For example, that suggests placing schedule buffers between engineering and fabrication, rather than between fabrication and installation. The fabrication and delivery processes are highly predictable, unless drawings are incorrect or incomplete, or drawings are pulled out of fabrication to be revised. A schedule buffer in front of fabrication would provide more time for engineering to complete its work and do it correctly. It would also provide the fabricator an opportunity to select and bundle work to meet his needs for production efficiency and the contractor's needs for quantities and sequence.

Another suggested rule: Size schedule buffers to the degree of uncertainty and variation to be managed. Research (Howell et. all, 1994) has shown that schedule buffers are sized without regard to the toughness of projects; i.e. their level of uncertainty. This amounts to wasting time and money accumulating piles of stuff not all of which is needed



CHAPTER (3) METHODOLOGY

3.1 Introduction

This chapter includes the methodology used in this research. It provides the information about the research strategy, research design, population, sample size, various approaches to data collection and data analysis. It also identifies the questionnaire design, pilot study, validity content, and reliability.

3.2 Research strategy

Research strategy can be defined as the way in which the research objectives can be questioned (Naoum, 1998). The explanation of mass behavior often requires mass attitude data that can only be obtained by a survey (Weisberg and Bowen, 1977). The people who provide information to the researchers are referred to as subjects, study participants, or respondents in quantitative research or as study participants or informants in qualitative research (Polit and Hungler, 1999). There are two types of research strategies, namely, 'quantitative research and qualitative research' (Naoum, 1998). Data may take the form of narrative information (qualitative data) or numerical values (quantitative data), (Polit and Hungler, 1999). Quantitative research is 'objective' in nature. It is defined as an inquiry into a social human problem, based on testing a hypothesis or a theory composed of variables, measured with members, and analysis with statistical procedures (Naoum, 1998). Quantitative researchers focus on the relationship between the independent variables and dependant variables (Polit and Hungler, 1999).

In this study, both quantitative and qualitative approaches are used. The questionnaire of this study is designed to get the factual information about local practices of contractors in managing construction materials in building projects as well as the opinions of contractors about these practices.



3.3 Research design

The purpose of this research is to explore the current practices of construction materials management of building projects and develop a computerized materials management system to be used in materials management works.

A structured questionnaire with personal interviews is used together in this research. The structured questionnaire is probably the most widely used data collection technique for conducting surveys. Questionnaires have been widely used for descriptive and analytical surveys in order to find out facts, opinions and views (Naoum, 1998). It enhances confidentially, supports internal and external validity, facilitates analysis, and saves resources. The advantages of interviews as summarized by Naoum (1998) are:

- the response rate is relatively high.
- providing more accurate answers and
- eliminating the tedium and idleness of the responds.

Data collected in a standardized form from samples of population. The standardized form allowed the researcher to carry out statistical inferences on the data, often with the help of computers. Using a questionnaire has some limitations such as: it must contain simple questions, no control over responds and respondents may answer generally (Naoum 1998).

3.4 Limitation of the research

The study is limited to Gaza strip contracting companies that are classified as first, second and third degrees, which have a valid registration in PCU. The subcontractors and contracting companies of fourth and fifth categories were excluded because these companies are too small to have specific arrangements for materials management.

3.5 Data collection and questionnaire design

In this research, few methods of data collection were used including observation, documentations, interviews and questionnaire and documentary analysis.

The good design of the questionnaire is a key to obtain good results and warranting a high rate of return. The questions of the research questionnaire are constructed based on:

• Literature review.



28

- Several interviews (39 interviews) with contractors to obtain different thoughts, which can be useful for creating questions.
- The experience of the researcher and some engineers in construction management in Gaza strip.

The questionnaire was built mainly using closed questions, and it was divided into five sections as follows:

Section one: company profile, which includes 10 items.

<u>Section tow:</u> Application of construction materials management tools and techniques in construction projects, which includes 17 items

<u>Section three:</u> The effect of closure in Gaza strip on materials availability and cost, which includes 5 items

Section four: Computer applications in materials management systems in construction projects, which includes 11 items

<u>Section five:</u> Implementation of construction materials management systems, which includes 12 items.

The questionnaire was developed in Arabic (Annex 1) to be more understandable by respondents. An English version is prepared (Annex 2) to help in documenting this research.

3.6 Field work research

The problem solving approach accompanied with a field survey was adopted for conducting this research. In the problem-solving approach (also named action research), the researcher reviews the current situation, identifies the problem, gets involved in introducing some changes to improve the situation and, possibly, evaluates the effect of his/her changes. This type of research is more attractive to practitioners, industrialists, and students from the professional background that has identified a problem during the course of work and wish to investigate and propose a change to improve the situation (Naoum 1998).

A questionnaire was designed and constructed to survey the situation and reality of construction materials management practices of Gaza strip contracting companies. The data which is collected by the questionnaire is analyzed and discussed. Figure 3.1 illustrates the methodology flow chart.





Figure 3.1: Methodology flow chart 3.7 Pilot study

A pilot study provides a trial run for the questionnaire, which involves testing the wording of questions, identifying ambiguous questions, testing the technique that used to collect the data, etc. (Naoum 1998). After the preliminary testing, a pilot study was conducted to evaluate the questionnaire; the researcher distributed the questionnaire to a sample of five different contracting companies to fill them. The purpose of this step is to discover if the questions are well understandable or not, also to find out any problem that may raise in filling the questionnaire. Generally speaking, it appeared that respondents had no difficulty in understanding the items or the instructions to complete the questionnaire.

The validity content of the questionnaire was tested by five contractors. Each of them has full information about the research objectives. Each of them was requested to evaluate validity content for each item based on rating the index of content validity. The contractors were then requested to rate each item based on relevance on the four-point ratings scale. The point scale developed by Yaghmaie (2003) as "1 = not relevant; 2 = item need some revision; 3 = relevant but need minor revision; 4 = very relevant". Based on comments of the contractors some minor changes, modifications, and addition were introduced to the questions.



The relatively of the questionnaire is a major criterion for assessing its quality and adequacy. The value of the reliability coefficient theoretically can range between - 1.00 and +1.00. For most purposes, reliability coefficients above 0.7 are considered satisfactory (Polit and Hungler, 1999). Five questionnaires were redistributed. The reliability coefficient was calculated which indicated a high level of reliability. For more accuracy, reliability coefficient was calculated for important parts.

section No.	contents	No. of questions	reliability coefficient
2	Application of construction materials management tools and techniques in construction projects.	43	0.9356
3	The effect of closer in Gaza strip on materials availability and cost.	5	0.7542
4	Computer applications in material management systems in construction projects.	13	0.7327
5	Implementation of construction materials management.	109	0.9626

Table 3.1 Reliability coefficient for the questionnaire.

3.8 Interview

The researcher interviewed 39 contractor company representatives and asked them to fill the questionnaire during the interview. The interview gave to a far extent, accurate and clear answers due to the clarifications which made by the researcher.

Before carrying out the interview, the questionnaire was sent to a company and specific time and date were determined for interview. This provided a chance for the contractors to study the questions before answering them. In the beginning of the interview the researcher introduced himself to the respondent to create a friendly atmosphere, then he thanked the respondent and affirmed that all the data to be collected would be used only for the research and would not be transferred to any other party.

3.9 Population and sample

The studied population includes the contracting companies in Gaza strip who have a contractor's union valid registration in January 2006. As materials management is a somehow sophisticated activity, the researcher addressed his study towards the top contracting companies of the first, second, and third class according to the contracting Union classification. The total number of these companies is 84 companies and they are stratified as follows: the first class has 34 companies; the second class has 21 companies; the third class has 29 companies.



3.10 Data coding and data analysis

The use of computers requires that answers given by the respondents be coded into numbers before the actual data analysis (Weisberg and bowen 1977). The coding process for closed ended questions consists of recording the number of response. The response categories must be mutually exclusive, so no answer can fit more than one of the categories (Weisberg and bowen 1977).

The returned questionnaire was numerically coded to enter the data systematically and efficiently. Data was entered using the statistical package for social sciences (SPSS) software. Double check, manually and by computer, was made to ensure data cleaning.

Statistical analysis for the data was conducted using the SPSS as follows:

- Defining and coding of variables.
- Summarizing the data on raw data sheet.
- Entering data.
- Cleaning data (Double check).

After applying the above-mentioned steps, the descriptive statistic method has been used. The descriptive method is one of analysis method which provides a general overview of results. It gives an idea of what is happening (Naoum 1998). In this study, frequency distribution and percentage were used to describe aspects of data. The researcher uses this method because large amounts of data were gathered. It is often useful to distribute the data into categories and to determine the number of individuals or cases belonging to each category. This is called "category frequency" (Naoum 1998). In this research, the data were presented in forms of tabulation, bar charts, and pie charts.

3.11 Developing and evaluating the computerized system

The researcher has developed a tailored construction materials management software that suits Gaza strip contracting companies. This software was developed depending on survey results, literature review, the experience of the researcher and other experts in construction materials management in Gaza strip, and the relevant computerized packages that the researcher reviewed such as Navon study (2002). He developed and evaluated an automated model for management and control of materials ordering, purchasing, and supply and use. In order to evaluate the model under real conditions,



the model was implemented in a prototype system and used in ongoing construction projects. The software was developed using *Microsoft Excel* platform. This is because contracting companies are familiar with Excel and find it easy to use. To increase the capability and skills of the researcher in Microsoft *Excel* package, he reviewed some books explaining the use of Excel, internet Excel publications, and Excel manual. These skills increased the capability of the researcher to develop the software. The developed software was tested by four contractors asking them to try it on real projects. They were also asked to fill a questionnaire for evaluating the software (See annex 5)



CHAPTER (4) SYRVEY RESULTS

4.1 Introduction

In this chapter, the results of the field survey are presented and discussed. The chapter illustrates and discusses the characteristics of the study population, application of construction materials management tools and techniques in construction projects, computer applications in material management systems in construction projects, and Implementation of construction materials management systems.

4.2 Study population characteristics

The general characteristics of the study population were investigated. They include the field of work, classification of contractors, number of employees and their qualifications, number of executed projects and their values during the last five years, and the person in-charge-of managing construction materials.

4.2.1 Year of establishment

Table 4.1 shows that only (14.2%) of the contracting companies were established before 1994. (36.9%) of companies were established from 1994 to 1997, while (48.8%) of them were established after 1997. This indicates that most of companies are relatively newly established having less than 13 years of experience.

Ver	ahla	Contractors				
var	ladie	Frequency	Percentage%			
XZ C	Before 1994	12	14.2			
Y ear of establishment	1994 - 1997	31	36.9			
establishment	After 1997	41	48.8			

 Table 4.1: Year of establishment of contracting companies



4.2.2 Field of work

Table 4.2 demonstrates that all respondents are involved, in a way or another, in building works, (94.9%) are involved in water and sewerage works, and (95%) are involved in roads works. This show how important is the building construction to the construction industry in Gaza strip.

Company work	Main	Secondary	Unspecialized	
Building work field	Frequency	75	6	0
Building work held	Percentage%	92.6	7.4	0
Water and sewerage	Frequency	44	30	4
works	Percentage%	56.4	38.5	5.1
Doodo works	Frequency	44	32	4
Roaus WOIKS	Percentage%	55.0	40.0	5.0

 Table 4.2: Field of company specialization

4.2.3 Classification of contractors

Table 4.3 illustrates that about half of contractors (43%) are classified as first class in building works. (22.6%) of them are classified in first class in sewage and water works and (19.0%) of the respondents are classified in first class in roads works.

 Table 4.3: Degree of classification

Company classificat contracting union for	First class	Second class	Third class	Others	
Duilding works	Frequency	34	21	29	0
Dunuing works	Percentage%	43.0	26.6	30.4	0.0
Water and sewerage	Frequency	19	20	36	9
works	Percentage%	22.6	23.8	42.8	10.7
Poode works	Frequency	16	26	35	7
RUAUS WULKS	Percentage%	19.0	30.9	41.7	8.4

4.2.4 Number of employees and their qualifications

From Table 4.4, it has been found that (97.6%) of respondents employ civil engineers, while (76.2%) of them employ architects, (55.9%) of them employ electrical engineers, and (59.5%) of contracting companies employ mechanical engineers.



Variable	No.	Percentage%
Civil engineer	82	97.6
Architect	64	76.2
Electrical engineer	47	55.9
Mechanical engineer	50	59.5
other specialist engineer	14	16.5
Technician	75	89.3

 Table 4.4: Qualification of employees in the contracting companies

Table 4.5 demonstrates that (45.2%) of respondents have 10 employees or less, whilst (3.6%) of contractors have more than 40 employees. The other (51.2%) of contracting companies has from 11 - 40 employees.

Number of employees	Frequency	Percentage%
10 and below	38	45.2
11-20	29	34.5
21-30	11	13.1
31-40	3	3.6
More than 40	3	3.6
total	84	100.0

 Table 4.5: Distribution of respondents' number of employee.

These results reveal how small the organization size of contracting companies in Gaza strip is generally. This means that most contractors execute their projects mainly using subcontractors. In fact, subcontracting is an essential component of almost any project.

4.2.5 The number and value of executed projects during the last five years

Table 4.6 shows that (42.9%) of contractors executed less than 10 projects during the last five years and only 3 contractors executed more than 40 projects. The majority of contracting companies (53.6%) executed from 20 - 40 projects in the same period.



Number of executed projects during the last five years	Frequency	Percentage%
10 and below	36	42.9
11-20	31	36.9
21-30	10	11.9
31-40	4	4.8
More than 40	3	3.6
total	84	100.0

 Table 4.6: Distribution of number of executed projects

Table 4.7 shows that (53.5%) of respondents executed projects with a value of less than 3 million dollars, during the last five years. (27.3%) of contracting companies executed projects with a value between 3 and 6 million dollars, and (19%) of contractors executed projects with a value of more than 6 million dollars. This indicates that most of executed projects are of small size.

Total amount of executed projects during the last five years (in million dollars)	Frequency	Percentage%
1.5 and lesser	27	32.1
1.6-3	18	21.4
3.1-4.5	17	20.2
4.6-6	6	7.1
6.1-7.5	6	7.1
More than 7.5	10	11.9
Total	84	100.0

 Table 4.7: Distribution of value of executed projects.

4.2.6 Distribution of respondent's person in-charge-of managing materials in construction projects

Table 4.8 shows that none of contracting companies has a specific section for managing construction materials. For other companies, the person in-charge-of managing construction materials is the director in (26.2%) of companies and the project manager in (50%) of companies.



The person in charge of managing construction materials in construction projects	Frequency	Percentage%
Company Director	22	26.2
Project Manager	42	50
Site Engineer	20	23.8
specific section	0	0.0
Others	0	0.0
Total	84	100.0

 Table 4.8: Distribution of respondents' person in charge of managing construction

 materials

4.3 Application of construction materials management techniques in construction projects

Considering the necessity and degree of using of some techniques in managing materials process in general, Table 4.9 illustrates that (69.5%) of contracting companies believe that updating the database for materials price when change occurs is the most important technique.

The results show general consistency between the necessity and usage degree of establishing database techniques and the necessity and usage degree of corresponding updating database techniques. The results show also consistency between the necessity and usage degree of each of these techniques.

Also Table 4.9 shows that the most important technique in establishing database group is "*creating materials price database*" (72.8%), and the most important technique in updating database group is "*updating the database for materials price when change occur*" (69.5%). It is noticed that the most important tool in establishing database group and updating database group is the tool that related to the prices of materials. This results may referred to the fact that in Gaza strip materials cost has the lion's share of the project cost.



			Nec	essity de	gree	Usage degree			
	No.	Techniques	Necessary %	Somehow Necessary %	Unnecessary %	Usually%	Occasionally%	Rarely	
base	1	Establishing categorized materials database	61.3	35.0	3.7	50.0	44.4	5.6	
g Data	2	Creating local suppliers database.	65.4	32.1	2.5	56.9	40.3	2.8	
lishing	3	Creating international suppliers database.	38.5	47.4	14.1	33.8	48.6	17.6	
Estab	4	Creating materials price database.	72.8	23.5	3.7	60.3	36.8	2.9	
e	5	Updating the database of local suppliers.	60.8	32.9	6.3	57.5	32.9	9.6	
ıtabas	6	Updating the database of international suppliers.	39.2	41.8	19.0	36.0	38.7	25.3	
lating da	7	Updating the database for materials price when change occurs.	69.5	22.0	8.5	64.7	29.4	5.9	
Upc	8	Using internet for knowing the new materials and their prices.	50.9	35.3	13.8	49.3	29.6	21.1	

Table 4.9: Technique necessity and usage in managing construction materials

4.3.1 The necessity, degree of usage, and method of use of some techniques in construction materials management

From the information in Table 4.10, it is clear that most contracting companies (85%) consider that using daily recording for used materials in the project is necessary, (77.8%) of contractors usually use this technique, and (55.6%) of them recording materials by using a form.

Other observation, the relatively least important technique in managing materials is reporting the situation of materials in project's store where (63.3%) of respondents consider it necessary and (56.0%) of them usually use it. This somehow come in line with Al-Ostaz's study (2004) who concluded that (75.6%) of the respondents report the situation of materials in the project's store in managing construction materials in Gaza strip.



The results demonstrate that (21.4%) of companies recording the received materials on site by using a computerized form, but in Al-Ostaz's study (2004), (2.4%) of the companies recording the received materials on site by using a computerized form. In general about half of contracting companies using the techniques shown in table 4.10 by using a form, but Al-Ostaz's study (2004) concluded that less than (10%) of contracting companies using a form to apply some techniques shown in Table 4.10.

		Necessity degree			Usage degree			Method of use			
No.	Techniques	Necessary %	Necessary somehow %	Unnecessary %	Usually %	Occasionally %	Rarely %	By using a computerized form %	Recording by using a form %	Recording without form %	Without recording %
1	Providing a list of materials in project that includes for example (material name, material number unit price).	74.1	22.2	3.7	68.0	24.0	8.0	20.0	50.0	25.7	4.3
2	Daily recording of using materials in the project.	85.0	13.8	1.3	77.8	22.2	0.0	20.8	55.6	22.2	1.4
3	Providing material cards at site store that contain for example (input-output- balance).	68.3	23.2	8.5	52.8	33.3	13.9	17.9	50.7	14.7	16.4
4	Providing materials purchase order including for example (order number-material description-required quantity –price).	76.9	17.9	5.2	60.8	28.4	10.8	21.2	62.1	10.6	6.1
5	Recording the received materials on site, the record shows for example (delivery number-supplier name-material description- quantity).	82.7	16.0	1.3	75.3	23.3	1.4	21.4	58.6	17.1	2.9
6	Reporting the situation of materials in the projects' store, the report shows (supplier name-order number-quantity input- quantity output-balance).	63.3	25.3	11.4	56.0	32.0	12.0	16.9	52.1	18.3	12.7
7	Reporting the problems for examples (wastage and breakage-thief and loss-shortage in delivery).	73.2	24.4	2.4	58.7	36.0	5.3	8.3	51.4	27.8	12.5
8	Following up the prices in the market and recording the variation of prices.	83.1	13.3	3.6	74.7	18.7	6.7	14.3	55.7	20.0	10.0
9	Following up the closure variations in the `market	80.0	8.2	11.8	68.1	23.6	8.3	10.0	50.0	22.9	17.1

 Table 4.10:
 The necessary, degree of usage, and method of use some techniques in construction materials management



4.4 The effect of closure of Gaza strip on materials availability and price

Table 4.11 represents the effect of closure on materials availability and price as expressed by respondents. These results show that all respondents (100%) agree that the closure causes a delay of a project completion. (98.9%) of them agree that the closure has big effect on the prices of basic materials such as cement, aggregate and reinforcement steel. On the other hand, the results show that (54.2%) of contracting companies believe that the closure increase subcontractor rates. The researcher expected all of these results. He was expected, in general, that the closure causes a noticeable increase in the main material prices, an increase of other material prices, a delay of a project completion, and causes an increase of total project cost.

No.	Factor	Very high degree effect %	High degree effect %	∑ Very high+ High degree%	Mid degree effect %	Little degree effect %	Very little degree effect %
1	The closure causes a noticeable increase in the main material prices specially (cement, aggregate, and steel).	81.0	17.9	98.9	1.2	0.0	0.0
2	The closure causes an increase of other material prices (painting, electrical, sewage material).	51.2	35.7	86.9	8.3	4.8	0.0
3	The closure increases the subcontractors' rates.	28.9	25.3	54.2	12.0	10.8	22.9
4	The closure causes a delay of a project completion.	69.0	31.0	100	0.0	0.0	0.0
5	The closure causes an increase of total project cost.	59.5	35.7	95.2	4.8	0.0	0.0

Table 4.11: The effect of closure of Gaza strip on materials availability and price



4.5 Using software for supporting the construction materials management

Table 4.12 shows that (28.6%) of the respondents do not use computer applications in material management systems in construction projects and (61.9%) use spreadsheetbased software like (Ms Excel) because it is a familiar application for all construction companies. On the other hand (9.5%) of them Use specialized software and this is somehow strange as the researcher did not get any evidence from interviewees of these software that they claimed they had used.

Table 4.12:	Using software f	or supporting t	he construction	materials management	

Using software for supporting the construction materials management	Frequency	Percentage%
Not use	24	28.6
Use spreadsheet-based software (Ms Excel)	52	61.9
Use specialized software.	8	9.5
total	84	100.0

4.5.1 Respondent's efficiency of using popular computer software

Table 4.13 shows the respondents efficiency of using popular computer software. These results illustrate that most of respondents (93.9%) have very good capabilities in using MS-Excel spreadsheets, where (60.3%) of contractors have very good in using Ms-Project. Only (24.1%) of them have the same capabilities in using Ms-Access. The distinguished place of Excel may refer to its advantages, which are stated by christofferson (1999) as the following:

• It is easy to use.

- It can be customized to the company style of doing business.
- It is powerful.

Table 4.13: Efficiency of respondents in using popular computer software

No.	software	V. Good %	Good %	Acceptable %	Low %	V. Low %
1	Excel	93.9	4.9	1.2	0.0	0.0
2	Word	91.6	7.2	1.2	0.0	0.0
4	Ms-Project	60.3	30.8	0.000	0.0	9.0
3	Access	24.1	17.7	12.7	11.4	34.2



4.5.2 Obstacles that are facing the local companies in using a construction materials management software

Table 4.14 represents the obstacles, which are facing the contractors in using construction materials management software. These results show that the majority of respondents (80%) consider absence of understanding of construction materials management software is the most important obstacle that affects the using of computerized materials management packages. (79.2%) of the respondents think that non-realization of importance of construction materials management system by the contractor has big effect on using computerized materials management packages. Another observation noted from Table 4.14, (49.5%) of the contractors believe that implementing the system wastes the time of project supervisors. The researcher notices that the most important obstacles are the cost related obstacles (obstacle Nos. 7 and 8).

The importance of technical obstacles has been supported by Navon study (2002); he concluded that the main reason for non-using computer in materials management is the lack of suitable software for managing construction materials.

No.	Obstacle	Very high degree effect %	High degree effect %	\sum Very high+ High degree%	Mid degree effect %	Little degree effect %	Very little degree effect %
1	Non- realization of importance of construction materials management system by the contractor.	32.9	55.3	79.2	4.7	7.1	0.0
2	Absence of understanding of construction materials management system.	25.9	54.1	80.0	16.5	3.5	0.0
3	Inability implementation of the system.	23.8	53.8	77.6	8.1	14.3	8.3
4	Shortage of user friendly of construction materials management system.	27.1	36.5	63.6	20.0	15.3	1.2
5	Shortage of qualified persons in using a construction materials management system.	44.7	32.9	77.6	15.3	7.1	0.0
6	Simplicity of manual managing a construction material.	24.1	34.9	62.0	28.9	7.2	4.8
7	The high cost of a construction materials management system.	25.0	32.1	57.1	17.9	16.7	8.3
8	The thinking that implementing the system wastes the time of project supervisors.	22.4	27.1	49.5	23.5	17.6	9.4

 Table 4.14: The obstacles which are facing contractors in using construction materials management systems .



4.5.3 Contractors' opinions regarding willingness to get materials management packages

In general, most of contractors surveyed (87%) are willing to get and use user friendly and low cost construction materials management software in managing materials. Table 4.15 shows that (70.2%) and (16.8%) of contractors surveyed are strongly agree and agree, respectively, to get and use a user friendly and inexpensive construction materials management software in managing construction materials.

I am ready to get and use user friendly and inexpensive cost construction material management system	Frequency	Percentage%
strongly agree	59	70.2
Agree	14	16.8
Neutral	5	5.90
Disagree	1	1.2
strongly disagree	5	5.9
total	84	100.0

Table 4.15: Contractors' willingness to get materials management packages.

4.6 Implementation of construction materials management systems

Implementation of construction materials management systems on construction projects provides the contracting company with many benefits and it can solve many problems that may face the contracting company.

4.6.1 Benefits of implementation of materials management on construction projects

Table 4.16 outlines the benefits of construction materials management systems according to contractors' opinions. The most important benefits are listed below:

- Materials are timely available on site with the right quantity (95.3%).
- Reducing duplication of materials orders (90.5%).
- Improving cash flow (86.6%).
- Complying to time schedule (85.1%).

On the other hand the benefits, which have lesser effect, are:

- Better relationships with suppliers (66.5%).
- Complying to enhancement of quality control. (74.7%).
- Reducing the space for materials on site (74.7%)



No.	Benefit	Very high degree effect %	High degree effect %	\sum Very high+ High degree%	Mid degree effect %	Little degree effect %	Very little degree effect %
1	Reducing the costs of project materials.	40.0	38.8	78.5	17.6	3.5	0.0
2	Better handling of materials.	40.5	40.5	81	17.9	0.0	1.2
3	Reducing duplication of materials orders.	41.7	48.8	90.5	8.3	1.2	0.0
4	Materials are timely available on site with the right quantity.	54.8	40.5	95.3	4.7	0.0	0.0
5	Improving labour productivity.	37.3	47.0	84.3	10.8	3.6	1.2
6	Complying to time schedule.	44.4	40.7	85.1	14.8	0.0	0.0
7	Complying to enhancement of quality control.	28.9	45.8	74.7	21.7	3.6	0.0
8	Improving follow up and monitoring of construction materials.	31.0	44.0	75	23.8	1.2	0.0
9	Better relationships with suppliers.	32.5	33.7	66.5	31.3	2.4	0.0
10	Waste reduction.	42.7	35.4	78.1	19.5	2.4	0.0
11	Reducing the space for materials on site.	32.5	42.2	74.7	19.3	6.0	0.0
12	Obtaining better price for the construction materials.	44.0	33.3	773	19.0	2.4	1.2
13	Improving cash flow.	50.0	36.6	86.6	8.5	3.7	1.2

Table 4.16: Benefits of implementation of materials management software on construction projects

4.6.2 Importance of materials management systems to solve some problems

Table 4.17 lists the contractors' opinion about the extent of effect of implementing materials management system to reduce material problems on construction site. The results show that the majority of contracting companies believe that its important to implement a construction materials management system to overcome most of material management problems. The results also show that the problems which have more conscious on its importance are "*Materials are not available with required quantity*" (92.9%), " *Slow response from the consultant engineer about submittals*" (91.5%), and "*Materials are not available*" (90.6%)



No.	Problem	Very important %	Important %	Σ Very important+ Important %	No difference%	Absolutely not important %	Absolutely never Important %
1	Materials are not available.	43.5	47.1	90.6	3.5	3.5	2.4
2	Materials are not available with required quantity.	35.3	57.6	92.9	5.9	0.0	1.2
3	Late delivery to the site.	38.8	49.4	88.2	11.8	0.0	0.0
4	Slow response from the consultant engineer about submittals.	32.5	59.0	91.5	8.4	0.0	0.0
5	Deliver wrong materials.	42.2	47.0	89.2	9.6	1.2	0.0
6	Deliver materials with wrong dimensions.	31.0	52.4	83.4	14.3	2.4	0.0
7	Deliver materials with wrong quantities.	38.1	47.6	85.7	13.1	1.2	0.0
8	Increase materials quantity in storages.	36.5	51.8	88.3	7.1	4.7	0.0
9	Burglary, theft and vandalism	39.3	44.0	83.3	14.3	2.4	0.0
10	Destroyed materials when deliver.	33.3	50.0	83.3	14.3	1.2	1.2

 Table 4.17: The importance of implementing a materials management system

4.6.3 The responsibility of ordering construction materials

Table 4.18 shows that (30.9%) of the respondents have the company manager responsible for ordering materials, and (48.8%) of them have the site engineer responsible for this duty. Another observation noted from Table 4.18, none of contracting companies has a specific section for ordering materials process.

The person who responsible about Ordering materials	Frequency	Percentage%
Company manager	26	30.9
site engineer	41	48.8
procurement department	17	20.2
Persons worked in a specific section	0	0

 Table 4.18:
 The responsible about ordering construction materials

4.6.4 The techniques used for ordering materials from suppliers

From Table 4.19, it has been found that most of contractors (97.7%) use the telephone for ordering materials from suppliers, (92.9%) of them prefer to use the fax for ordering materials, while (47%) of respondents use the E-Mail, and (44.7%) of them use internet for ordering materials from suppliers. It is noticed that the telephone is the most important tool for ordering materials. This result may refer to the fact that in



Gaza strip construction materials are ordered from local suppliers. It is noticed that the modern techniques for ordering materials such as websites are not use in Gaza strip.

No.	Technique	Very important %	Important %	Σ Very important+ important %	I don't know	Not important %	Not Important at all %
1	Internet.	11.8	32.9	44.7	10.6	35.3	9.4
2	E-mail.	12.9	34.1	47	8.2	38.8	5.9
3	Fax.	55.3	37.6	92.9	4.7	2.4	0.0
4	Telephone.	71.8	25.9	97.7	2.3	0.0	0.0
5	Personal meeting.	54.1	31.8	85.9	4.7	4.7	4.7

 Table 4.19:
 Techniques for ordering construction materials from suppliers.

4.6.5 Available covered and open storages

As illustrated in Table 4.20, most of respondents (52.4%) have an area of 200m2 and below of covered storages, and (53.6%) of the contractors have an area of 1,000 m2 and below of opened storages. It is noted that none of the contractors have an area more than 2,000 m2 of covered storages, and only one of them have an area more than 10,000 m2 of opened storages.

Available area of covered storages	Frequency	Percentage%	Available area of opened storages	Frequency	Percentage%
200m2and below	44	52.4	1,000m2and below	45	53.6
201-500m2	26	30.9	1,001-3,000 m2	22	26.2
501-1,000m2	9	10.7	3,001-6,000 m2	13	15.5
1,001-2,000m2	5	6	6,001-10,000 m2	3	3.6
more than 2,000m2	0	0	more than 10,000m2	1	1.1
total	84	100	total	84	100

 Table 4.20:
 Available covered and open storages.

4.6.6 Materials that are stored in the company's storages

Table 4.21 shows how often the construction materials are stored in the contractor's storages. The most important materials that often stored in the contractor's storages are (shuttering timber, cement, sanitary materials, electrical materials, and pipes). On the other hand, materials that are rarely stored in the contractor's storages are (curbstone, stone, blocks, terrazzo tiles, and sand)

Table 4.21: How often contracting companies store construction materials



No.	Material	Always %	Often %	Σ Always+ Often%	Sometimes %	Seldom %	Never %	∑ Seldom+ Never %
1	Reinforcement steel.	6.0	14.3	20.3	39.3	27.4	13.1	40.5
2	Shuttering timber.	28.6	13.1	41.7	20.2	33.3	4.8	38.1
3	Cement.	11.9	22.6	34.5	23.8	35.7	6.0	41.7
4	Sand.	7.1	4.8	11.9	29.8	35.7	22.6	58.3
5	Aggregate.	4.9	4.9	9.8	42.7	35.4	12.2	47.6
6	Blocks.	4.8	3.6	8.4	27.7	44.6	19.3	63.9
7	Tiles.	4.8	7.1	11.9	29.8	48.8	9.5	58.3
8	Painting.	8.4	3.6	12	31.3	44.6	12.0	56.6
9	Stone.	6.1	4.9	11	22.0	35.4	31.7	67.1
10	Pipes.	13.1	11.9	25	31.0	32.1	11.9	44.0
11	Terrazzo tiles.	1.2	6.0	7.2	32.1	39.3	21.4	60.7
12	Curb stone.	0.0	6.0	6.0	26.2	46.4	21.4	67.8
13	Base course.	3.6	19.0	22.6	23.8	36.9	16.7	53.6
14	Electrical materials.	11.9	19.0	30.9	28.6	31.0	9.5	40.5
15	Sanitary materials.	14.3	16.7	31	33.3	28.6	7.1	35.7

4.6.7 The importance of some factors in qualifying suppliers

Table 4.22 shows that (98.8%) of the respondents believe that "*price competitiveness*" is the main important factor in qualifying suppliers, and the least important factor is "*having many branches in different geographical areas*".

No.	Factor	Very high degree effect %	High degree effect %	∑ Very high+ High degree%	Mid degree effect %	Little degree effect %	Very little degree effect %
1	Price competitiveness.	57.6	41.2	98.8	0.0	1.2	0.0
2	Short period to deliver materials to the contractor	56.5	41.2	97.7	2.4	0.0	0.0
3	Quality of materials	45.9	50.6	96.5	2.4	1.2	0.0
4	Increase payment period	60.0	31.8	91.8	3.5	4.7	0.0
5	The capability of supplier in the market	61.2	29.4	90.6	5.9	3.5	0.0
6	Having many branches in different geographical areas.	46.4	35.7	82.1	6.0	8.3	3.6

Table 4.22: The factors which are affecting qualifying suppliers.

4.6.8 Frequency of contractor problems at qualifying suppliers



Table 4.23 shows that the most frequent problems that are facing contractors in the process of qualifying suppliers are the prevailing political condition, the sole supplier and lack of penalty measures against defaulted suppliers, respectively.

From the information in Table 4.23, it has been found that only (1.1%) of the contracting companies never face the problems shown in Table 4.23.

No.	problem	Always %	Often %	Sometimes %	Seldom %	Never %
1	Sole supplier.	29.8	50.0	15.5	4.8	0.0
2	Lack of required information about suppliers.	13.4	62.2	17.1	7.3	0.0
3	Wrong information about suppliers.	16.7	39.3	36.9	6.0	1.1
4	Lack of penalty measures against defaulted suppliers.	31.0	41.7	22.6	4.8	0.0
5	Inability to enforce contract condition on suppliers.	28.6	41.7	17.9	11.9	0.0
6	Prevailing political conditions.	54.8	36.9	8.3	0.0	0.0

 Table 4.23: Problems which are facing the companies when qualifying suppliers.

4.6.9 The importance of knowing waste percentage for different building materials

As Table 4.24 indicates, most contractors think that knowing waste for different building materials helps them to determine the exact quantities required for a project, to price tenders more accurately and to prepare accurate bill of quantities

No.	Importance	Strongly agree %	Agree %	Σ Strongly agree+ Agree %	Neutral %	Disagree %	Strongly disagree %
1	Help to determine the exact required quantities.	50.0	47.6	97.6	2.4	0.0	0.0
2	Increase the chance for obtaining the project finance.	22.6	52.4	75	19.0	6.0	0.0
3	Help for preparing accurate bill of quantities.	38.1	59.5	97.6	2.4	0.0	0.0
4	Help contractors to price tenders more accurately.	50.0	44.0	94	6.0	0.0	0.0
5	Help the contractor to have a better chance to win the tender.	39.3	47.6	86.9	8.63	3.6	1.2
6	Knowing the real requirements for the project (materials-time-	34.5	51.2	85.7	13.1	1.2	0.0

Table 4.24: The importance of knowing materials waste percentage.



No.	Importance		Agree %	Σ Strongly agree+ Agree %	Neutral %	Disagree %	Strongly disagree %
	cost)						
7	Help to finish the project successfully and have profits.	40.5	51.2	91.7	7.1	1.2	0.0
8	Help in preparing a good schedule program including project resources.	33.7	44.6	78.3	15.7	6.0	0.0
9	Help contractor to prepare a good materials schedule program.	34.5	45.2	79.7	14.3	6.0	0.0

4.6.10 Material waste on site

In this section, the researcher divides the factors that are causing increase in material waste on construction site into five groups. They are: on site practice; material handling; transportation; site management and supervision on site. The total numbers of factors are 35 factors.

a. On site practice

Table 4.25 represents the factors related to on site practice which cause increase waste on construction site. The results show that the majority of contracting companies believe that most of the factors, which are shown in Table 4.25, cause waste increase on construction site. They believe that the factors which have bigger effect on causing waste increase on construction site are poor quality of materials (89.1%), materials damage on site (88.2), and improper cutting of materials (84.7%).

On the other hand, the factor which they believe has the lowest effect on causing waste increase on construction site, is the existence of unnecessary materials on site (64.3%). This comes in line with Al-Mogany study (2006), who concluded that more than 80% of contractors consider that which factors causes waste on construction sites.

 Table 4.25: Factors that increase waste on the construction site (Group a)



	No.	Factors causes increase waste on site	Very high degree effect %	High degree effect %	∑ Very high degree effect+ High degree effect %	Mid degree effect %	Little degree effect %	Very little degree effect %
	1	Materials damage on site.	50.6	37.6	88.2	10.6	0.0	1.2
	2	Improper cutting of materials.	40.0	44.7	84.7	15.3	0.0	0.0
e	3	Existence of unnecessary materials on site.	23.8	40.5	64.3	28.6	6.0	1.1
actic	4	Overproduction/Production of a quantity greater than required or earlier than necessary.	23.5	45.9	69.4	18.8	9.4	2.4
Id	5	Manufacturing defects.	29.8	48.8	78.6	16.7	4.8	0.0
ite	6	Burglary, theft and vandalism.	36.9	46.4	83.3	14.3	0.0	2.4
n s	7	Poor quality of materials.	38.1	51.2	89.3	9.5	0.0	1.2
0	8	Lack of materials (due to closure).	49.4	32.5	81.9	3.6	7.2	7.2
(F	9	Lack of on site materials control.	41.7	34.5	76.2	19.0	4.8	0.0
Ľ	10	Poor storage of materials.	37.3	42.2	79.5	20.3	0.0	0.0
	11	Over sizing structural elements during execution.	29.8	44.0	73.8	20.2	3.6	2.4
	12	Using excessive quantities of materials.	30.6	40.0	70.6	21.2	8.2	0.0

b. Materials handling

Table 4.26 represents the factors related to materials handling group which cause material waste increase on construction site. (75.8%) of the contracting companies think that *"duplication of transporting material on site"* causes waste increase, (69.1%) of them believe that *"insufficient instructions about handling materials on site"* causes waste increase, and (67.2%) of the respondents believe that *"improper handling of materials on site"* causes waste increase.

	No.	Factors causes waste increase on site	Very high degree effect %	High degree effect %	∑ Very high degree effect+ High degree effect %	Mid degree effect %	Little degree effect %	Very little degree effect %
ials g:	13	Improper handling of materials on site.	32.1	35.1	67.2	29.8	2.4	0.0
(B):Mater handlin	14	Duplication of transporting material on site.	30.5	45.1	75.8	18.3	6.1	0.0
	15	Insufficient instructions about handling materials on site.	26.2	42.9	69.1	23.8	7.1	0.0

 Table 4.26: Factors that increase waste on the construction site (Group b)

c. Material transportation



Table 4.27 represents the factors related to transportation group which cause material waste increase on construction site. (81.2%) of the contracting companies believe that *"improper material transportation"* causes waste increase, and (69.5%) of them think that *"storing materials in far away stores"* causes waste increase.

	No.	Factors causes increase waste on site	Very high degree effect %	High degree effect %	∑ Very high degree effect+ High degree effect %	Mid degree effect %	Little degree effect %	Very little degree effect %
(C): Transportation	16	Improper materials.	36.5	44.7	81.2	16.5	2.4	0.0
	17	storing materials in far away stores	22.4	47.1	69.5	24.7	5.9	0.0

 Table 4.27: Factors that increase waste on the construction site(Group c)

d. Site management

Table 4.28 represents the factors related to Site management which cause waste increase on construction site. The results show that the majority of contracting companies believe that the factors which have bigger effect on causing material waste increase on construction site are "Lack of material and time waste management plan" (83.8%), "Ineffective control of the project progress by the contractor" (80%), and "delay in project commencement" (79.7%). On the other hand, the factors, which have lower effect on causing waste increase on construction site, are "providing project team with insufficient information" (64.7%), "poor site layout" (70.6%), and "lack of a quality management system" (71.5%).

 Table 4.28: Factors that increase waste on the construction site (Group d)



	No.	Factors causes increase waste on site	Very high degree effect %	High degree effect %	\sum Very high degree effect + High degree effect %	Mid degree effect %	Little degree effect %	Very little degree effect %
	18	Lack of material and time waste management plan.	31.8	52.9	83.8	14.1	2.1	0.0
ţ	19	Lack of a quality management system aimed at waste minimization.	17.9	53.6	71.5	26.2	2.4	0.0
nen	20	Poor site layout.	31.8	38.8	70.6	24.7	4.7	0.0
nagen	21	Poor qualification of the contractor's technical staff assigned to the project.	34.1	42.4	76.5	22.4	1.2	0.0
te ma	22	Providing project team with insufficient information.	22.4	42.3	64.7	32.9	2.4	0.0
(D): Sit	23	Ineffective control of the project progress by the contractor.	36.5	43.5	80	18.8	0.0	1.2
	24	Shortage of technical professionals in the contractor's organization.	31.0	44	75	23.8	0.0	1.2
	25	contractors slowness in taking decisions.	27.4	48.8	76.2	20.2	2.4	1.2
	26	Delay in project commencement.	39.3	40.4	79.7	15.5	2.4	2.4

e. Site supervision

Table 4.29 represents the factors related to site supervision group, which causes material waste increase on construction site. The results show that the majority of contracting companies believe all factors shown in Table 4.29, have a high degree effect on waste increase on construction site. This result shows the adversely relationship between the contractor and the consultant.



	No.	Factors causes increase waste on site	Very high degree effect %	High degree effect %	∑ Very high degree effect+ High degree effect %	Mid degree effect %	Little degree effect %	Very little degree effect %
or:	27	Poor control of supervision and delay in giving instructions.	50.6	32.9	83.5	15.3	1.2	0.0
	28	Poor qualification of consultant engineer's staff assigned to the project.	56.5	24.7	81.2	17.6	1.2	0.0
	29	Slow response from the consultant team to contractor inquiries.	51.2	33.3	84.5	13.1	1.2	1.2
pervis	30	Delay in performing inspection and testing by the consultant team.	54.5	29.9	84.4	13.0	2.6	0.0
(E): Site sul	31	Poor coordination and communication among the consultant, the owner and the contractor.	47.1	37.6	84.7	9.4	5.9	0.0
	32	Owner's delay in handing over the site to the contractor.	34.1	48.2	82.3	10.6	4.7	2.4
	33	Suspension of work by the owner.	43.5	36.5	80	14.1	3.5	2.4
	34	Change orders.	41.2	45.8	87	11.8	0.0	1.2
	35	Poor cooperation of the owner towards settling contractors payments and claims	48.2	36.5	84.7	14.1	0.0	1.2

Table 4.29: Factors that increase waste on the construction site (Group e)



4.7 Conclusions

From the results obtained, analyzed, and discussed, the researcher concludes that:

- The contracting companies in Gaza strip are:
 - Relatively newly established.
 - Involved mainly in building works.
 - Small size organizations.
 - Depending heavily on subcontractors.
- Most contractors are interested in using some techniques in construction materials management such as:
 - Providing a list of materials in project that includes for example (material name, material number unit price).
 - Daily recording of using materials in the project.
 - Providing material cards at site store that contain for example (inputoutput-balance).
 - Reporting the situation of materials in the projects' store, the report shows (supplier name-order number-quantity input-quantity output-balance).
 - Following up the prices in the market and recording the variation of prices.
- The Israeli closure has big effect on materials availability and cost.
- Some obstacles that face the contractors in using computerized materials management systems are:
 - Non- realization of importance of construction materials management system by the contractor.
 - Absence of understanding of construction materials management system.
 - Inability of implementing of the system.
 - Shortage of user friendly of construction materials management system.
 - The high cost of a construction materials management system.
 - The belief that implementing the system wastes the time of project supervisors.
- Many benefits can be obtained from using a user friendly and inexpensive computerized materials management systems such as:
 - Improving cash flow.
 - Reduce duplication of materials order .
 - Materials are on site with required time and quantity.
 - Complying with time schedule.



- Complying with enhancement of quality control.
- Reduce the space for materials on site.
- Many problems can be reduced when contractors implement construction materials management systems such as:
 - Materials not available.
 - Materials not available with required quantity.
 - Late delivery to the site.
 - Deliver wrong materials
 - Slow response from the consultant engineer about submittals.
 - Destroyed materials when deliver.
- Many benefits can be obtained from knowing waste percentage for different building materials such as:
 - Help to determine the exact required quantities.
 - Help for preparing accurate bill of quantities.
 - Help contractors to price tenders more accurately.
 - Help to finish the project successfully and have profits.
- There are many factors that cause increase to material waste on site they can be categorized as the following:
 - a. On site practice
 - Poor quality of materials.
 - Materials damage on site.
 - Improper cutting of materials.
 - Burglary, theft and vandalism.
 - Lack of materials (due to closure).
 - b. Materials handling
 - Duplication of transporting material on site.
 - Insufficient instructions about handling materials on site.
 - Improper handling of materials on site.
 - c. Material transportation
 - Improper materials.
 - storing materials in far away stores
 - d. Site management
 - Lack of material and time waste management plan.
 - Ineffective control of the project progress by the contractor.



- Delay in project commencement.
- Poor qualification of the contractor's technical staff assigned to the project
- e. Site supervision
 - Change orders.
 - Poor cooperation of the owner in settling contractors payments and claims.
 - Slow response from the consultant team to contractor inquiries.
 - Delay in performing inspection and testing by the consultant team.



CHAPTER 5

Construction Materials Management Software (CMMS)

5.1 Introduction

This Chapter presents the computerized system which the researcher developed to help the Gaze strip contractors to improve their practice in construction materials management. The author named this software Construction Materials Management Software (CMMS). It also discusses the general concepts on which the development of software was based. The chapter describes the software components, and the method of use. The software evaluation and limitations are also discussed.

5.2 Concepts

The researcher reviewed the current situation of construction materials management in Gaza strip by interviewing eighty-four contractors. He found out that the construction materials management practices are generally inadequate and in the first stages. In addition, he concluded that construction materials management practices are not done in a systematic way. The researcher aims, by introducing CMMS, at improving the common practices of local contractors.

Ahuja et al. (1994) summarize the criteria for selection a software system as follows:

- 1. The software must be relatively easy to install and operate. The input data must be easy to prepare, and the output reports must be understandable.
- 2. Data sorting is one of the base uses of computers.
- 3. It must be a fully tested system and should have a proven record.
- 4. The program should be flexible and have capacity for handling many types of application.
- 5. The database must contain all the necessary elements so it can be managed to generate the desired information reports.
- 6. The program should be compatible with other programs and systems in use in the company.
- 7. The system must be economical in terms of installation, operation, and maintenance.

The author tries his best to accommodate, as much as possible, the above mentioned criteria in development of CMMS.



58
5.2.1 Computerized system

Figure 5.1 illustrates a schematic diagram for CMMS. The software consists of four parts. Part one (input data and basic calculations), part two (materials purchase decisions), part three (materials cards), and part four (waste control).

They are detailed as follows:

<u>Part 1 (Input data and basic calculations)</u>: The input data and basic calculations can be classified into company and project information; materials pool; activity pool; project activities; activities materials quantities; materials needed; "2 dates materials to order"; materials spaces and cash requirements; earned value calculations and earned value S-curve.

<u>Part 2 (Materials purchase decisions)</u>: The materials purchase decisions can be classified into lead time for delivery and ordering; "materials must be purchased at"; materials spaces at (purchase order) P.O. date; materials prices at P.O. date; cumulative prices and spaces; "purchase order and materials must be on the site at".

Part 3 (Materials card): It has one sheet called materials card.

Part 4 (waste control): It has one sheet called waste control.



Figure 5.1: Construction Materials Management System (CMMS) 5.2.2 Excel environment

The software is built within Excel environment. It includes spreadsheets processed by number of functions which automate repetitive steps. Spreadsheets have many advantages such as:

• They are versatile.



- They are inexpensive.
- They are easy to use.
- They can customize to your style of doing business.
- They are powerful.
- They provide a clear and complete picture of inputs and outputs (Christofferson 1999).

The field survey illustrated that most companies use Excel more than other programs such as MS word, MS project, and Access. For this reason and for the previous mentioned advantages, the researcher has chosen Excel environment for developing CMMS. This will make the user more comfortable and confident in using CMMS.

5.3 Software description and components

CMMS basically consists of number of spreadsheets processed by functions. Template sheets with embedded formulas are also utilized to make CMMS more user friendly. The following is a general description of CMMS components:

5.3.1 Start sheet

Figure 5.2 illustrates the start sheet. It provides a summarized identification of the software. The user can open the main menu sheet by clicking on "*Enter*" *icon* of the bottom at this sheet.



Figure 5.2: Start sheet



5.3.2 Main menu

This sheet contains many icons which represent the sheets and components of the software. By clicking on any icon, the sheet related to this icon opens automatically. The basic topics of the main menu are:

- Part 1 (Input data and basic calculations).
- Part 2 (Materials purchase decision).
- Part 3 (Material card).
- Part 4 (Waste control).
- Help and method of use.

The followings are more detailed description of these icons of the menu:

5.3.2.1 Part one (Input data and basic calculations)

It includes basic input data about the company and the project, activities pool, materials pool, project activates, activities materials quantities, materials needed, "2 dates materials to order", materials spaces and cash requirements, earned value S-curve calculations, and earned value S-curve diagram. Basic input data can be either directly entered or electronically transferred, especially the project activities sheet, which can be transferred from the *Microsoft project*. The followings are details of the basic input data and calculations sheets:



Figure 5.3: "Main Menu" sheet.



A. "Company and project information" sheet

"Company and project information" sheet involves data about the company like the company name, address, city, phone number, location, and fax number. It also includes data about the project like the project name, number, location, total cost, start date, and finish date. This sheet can be opened by clicking on the company and project information icon in the main menu. The data of the company and project can be entered by typing or electronic transmission. Once this is entered it automatically appears on the other sheets. The user can turn back to the main menu by clicking on "*Back to Main Menu*" icon at the bottom of this sheet. Figure 5.4 illustrates the company and project information sheet.



Figure 5.4: "Company and project information" sheet

B. "Activity pool" sheet

"Activity pool" sheet can be opened by clicking on the activities pool icon in the main menu. This sheet includes data about activity description, activity unit; activity pool code, materials name and its quantity, which is required for achieving one unit of an activity. These materials and quantities are entered for one time for all building projects. An activity code has four digits, the first two digits are reserved to main



work type classification, the researcher assigned for example 01 for general works, 02 for earth works, 03 for concrete works, 04 for block works, 05 for plaster works, 06 for tiling works, 07 for painting works, 08 for carpentry works, 09 for metal and aluminum works, 10 for electrical works, 11 for mechanical works, 12 for isolation works, 13 for special works and 14 for finishing works. The other two digits is reserved to different work types such as building blocks works 10, 15 and 20 cm thick. The materials and its quantities which are used for one unit of activity are transferred into project activates sheet automatically and these quantities are multiplied by total quantities of each activity. (See annex 4)

28									Mi	crosoft Excel - F	Partil 📕	FX
:	بؤالأ للتغليمات	اکتب س				تعليمات	لطار	ق أدوات بيانات	اج تنصي	تحرير عرض إدر	(🖻 ملف	- 8 ×
: .	~		Arial	(S)	Dian							
: 🛄			Ana	- 11 - 11	- B				0 7 - 0		- <u></u> -	· 🚢 🍾 🗧
	E13 👻	f× 0								4		
A	В	C	D	E	F	G	н		J	к	L	~
2	Al Haddad cr	nmany										
3	Gaza Al Tofa	h	BA	CK TO M. MENU								
4	Gaza		-									
5	Phone :	08 2814926		ENTER	100		BAC	K TO MAT.POOL				
6	Fax:	08 2814926			-	Sec. 1997						
7												
8		ACTIVITY POOL										
40												
10						11-1	a rial	Day One I	Init O	6 Antio iter		_
11	k					Iviat	erial	s Per One c		ACUVILY		
	Act.				Mat1		Mat2		Mat3		Mat4	
	Pool	Activity Description	Unit	Mat. 1 Name	Quan./	Mat. 2 Name	Quan./	Mat. 3 Name	Quan./	Mat. 4 Name	Quan./	Mat
12	Code		a la constante		Unit		Unit		Unit		Unit	
13	0101	أتمال النجهيز	LS		-		Î			6	i i	-
14	0201	أعمال الحفر	m3	رمل	<u> </u>				1			-
15	0202	أعمال الردم	m3	حصمه اسمت								
16	0301	خرسانة B200 لزوم باطون النظافة أسفل فواحد المبنى ر بسماكة ١٠ سم	m3	8150 8200	1			÷		¢.		
17	0302	باطون B 250 لزوم الفواعد المنفصلة	m3	8250	1	Fy4200	0.06					1
18	0303	باطون B250 لزوم الأحزمة الأرضية	m3	8350	🐱 1	Fy4200	0.08					
19	0304	خرسانة مسلحة B300 لزوم رقاب الأحمدة	m3	B300	1	Fy4200	0.05		1			1
20	0305	باطون B200 لزوم الأرضيات ويسمك ١٠ سم	m3	B200	1	Fy4200	0.04					1
21	0306	خرسانة مسلحة B300 لزوم الأحمدة	m3	B300	1	Fy4200	0.5					
22	0307	خرسانه مسلحه B250 گزوم السقف بسماکه ۲۰ سم	m3	B250	1	Fy4200	0.07					
23	0308	خرسانة مسلحة B250 تزود جلسات واعتاب الايواب والشبابيك	m3	B250	1	Fy2800	0.09		0.010			
24	0401	بناء بلوك اسمنني ٢٠٠١ - ٦ مد مفرع تزور الفواطع الحارجية	m2	بتوات مفرع ۲ سم	13	اسمنت	0.008	رمن	0.016	شيد	0.004	
25	0402	بتوت اسمندي ١٠٦٠٠٠٠ اسم مفرع تروم الفواطع الداخلية. منذأ منتشأة مناقد الدائلية من أن تتمتل معين ممالياتا من	m2	بدوت مفرع ۲۰ سد	13	Citeral .	0.008	<u>رمن</u>	0.016	يسترد	0.004	
20	0501	موله المسيد تروم المستارة الداعمية حمى ان مع حمى وجهين العام المد	m2	cate of	0.004	(m) (h)	0.000		0.002			
27	0502	مونة من الأسمنت الأست والكدارة عارجية على عارت وجوة الروم العبني مدنة من الأسمنت الأست، والكدارة لا مد أتتمال الا شقة الخارجية	m2	اسمنت	0.004	 	0.008	200	0.002	استنت انتض وفقة	0.008	24
20	0601	تدريد وتركيب بالأكسر وخاد للأرضيات	m2	لاطرندراذه كسر ريخاد ٢٥ س	1	داندا، ندر از و ۲۵ سد	1	اسمنت	0.002	141	0.000	
30	0602	توريد وتركيب بالإط بورسالان لأرضية الصالة	m2	بالإط بور سالان	1	بانش ندر از و ۲۵ سد	1	اسمنت	0.006	د من	0.025	
31	0603	وريد وتركيب بلاط سير اميكا لأرضيات الدورات	m2	بلاط كراميكا ارضيات	1	اسمنت	0.006	رمان	0.025	شيد	0.003	
32	0604	توريد وتركيب بلاط سير اميك لحوائط الدورات	m2	هد کر امیکا حوائط ۲۰*۳۰	1	اسمئت	0.006	رمن	0.025	شيد	0.003	
33	1401	توريد وتركيب رخام ايطالى نوع كرارا لزوم دعسات الأدراج	ml	رخام كرارة	1		2.11			¢		1
34	1402	توريد وتركيب رهام خليلي بسماكةً ٣ سم وحرض ٢٦ سم لطسات الشبابيك	m2	رخام خليلي	1				1			
35	0701	توريد و دهان الاسقف نلانة اوجه على الأثل بوليسيد او مايعادته	m2	بوليسيد	0.05				5 (S			1
36	0702	نوريد و دهان بليور مطفى للحوائط الداخلية وجهين	m2	زيت	0.04				9			
37	0801	تورید وترکیب باب کیس مقاس ۹۰×۲۲۰ سم	no	باب خشب ۲۰ مم	1							
38	0802	كما سيق مقاس ٨٠ × ٢٠٠ سم والزرفيل WALLY نوع اشارة	no	باب خشب ۸۰سم	1							
39	0803	كما سبق مقاس ١٠٠× ٢٠٠ سم والزرفيل WALLY نوع اشارة	no	باب خشب ۲۰۰۰	1							
40	0804	كما سبق مقاس ١١٠×١٠٠ سم والزرفيل WALLY نوع اشاره	no	باب خشب ۱۱سم	1							~
H 4	► ► \ Inter	Face / main menu / com@pro.inf / Mat.Pool \ Act.Pool / I	Pro.Act	/ Qua.of Mat Of Ac	: / Ma	t To Order 🖌 2Dat	e Mat.T	o Order / Mat.Ass	Ear \	/. S-Curve Cal / Ea	ar < 🚺	2
	NUM											جاهز
. 15	CO 00 AT	Part4 RD		Part3 M1		Part2 Mit		Paul 🖬	MAS			1 44
B. 10		i altri 🛃						T GILL 2				

Figure 5.5: "Activity Pool" sheet

C. "Materials Pool" sheet

Materials pool sheet contains materials code, material name, material unit, the space needed to store one unit of the material and unit price. This data can be typed manually or transferred electronically from another sheet. Most of materials used in building works entered in the sheet for one time. Material unit can be typed or chosen from a list as shown in Figure 5.6.



I					
	تغليمات	🚽 اكتب سؤالاً لل			
		👌 🖂 Ι 🚉 Ι 🕰 • 🕩 Ι 🤊 • Ι Σ - 🖇	1 %10	• 🕜 谋	Arial
-	A1				
1	В	C	D	E	F
2	Al Haddad	company			
3	Gaza Al To	fah 😽	BACP	TO M. MENU	1
4	Gaza Phone	08 2814926		ENTER	
6	Fax :	08 2814926			
7		MUTTERLUT DOOL	-	P.	
9		MATERIAL POOL		<u> </u>	
11					
12	M. Code	Material Name	Unit	Unit Price	Unit Space
13	1	رسۇ	m3	15.00	0.50
14	2	حصمة	m3	88.00	1.00
15	3	اسمنت D450	ton	550.00	2.00
10	- 4	B 130 B 200	m3	210.00	3.00
18	6	B250	m3	230.00	2.00
19	7	B300	m3	300.00	4.00
20	8	B350	m3	320.00	2.00
21	9	Fy4200	ton	3000.00	1.00
22	10	Fy2800	ton	3000.00	2.00
23	11	سنته تربيند	ton	50.00	6.00
24	12	1	Kg ka	50.00	2.00
26	14	مسامع ٢	kg	45.00	3.00
27	15	خشب طوبار	m3	1200.00	1.00
28	16	بلوڭ مصمت ٢ سم	no	3.00	3.00
29	17	بلوڭ مفرغ ٢٠ سم	no	2.00	2.00
30	18	بلوك مفرغ ١٥ سم	no	1.70	3.00
31	19	بتوى مفرع ١١ سم	no	1.60	3.00
32	20	بتوات معرج ١٠ سم داد 2 سفف ١٧ سد	no	2.00	3.00
34	22	بلوڭ سفف ۲۰ سې	10	2.50	2.00
35	23	اسمنت ابيض ٥٠ كخ	no	30.00	1.00
36	24	کوارنز ۲۵ کغ	no	25.00	3.00
37	25	خيش فصارة	no	5.00	2.00
38	26	شيد	ton	450.00	3.00
39	27	شبكه بلاستيكه للفصارة	no	15.00	3.00
40	28	روی مجمعه تعمیاره الحد تبداده کسر دخاد ۲۵ سد	m2	5.00	2.00
47	30	بانش نیر از و ۲۰ سر	ml	40.00	4.00
43	31	بالط نیر ازی کسر رخام ۳۰ سم	m2	55.00	3.00
44	32	بانیل نیر ازی ۳۰ سم	mi	30.00	2.00
45	33	بالطّ کر امیکا حوائط ۲۰۴۰ سم	m2	60.00	4.00
14 4	► ► ► Ir	nter Face / main menu / com@pro.inf)	Mat.Pool	Act.Pool	(Pro.Act / C
	1	MUM			
14	.157 📸 0	AB Microsoft Word What	يدار هدار ال	. 19	Part4

Figure 5.6: "Materials Pool" sheet

D. "Project Activities" sheet

"Project activities" sheet contains story number, activity pool code, activity description, unit, duration, start and finish dates, quantity, unit price and total price. This data can be obtained from Ms. Project by copying and pasting or typing manually. Activity code in this sheet must match the activity code in activity pool sheet. For example, B300 for column works in activity pool sheet have the code number 0360. This activity in the project sheet must have the same code even if this activity used in any storey (The activity code of B300 for the ground floor column works is 0360, and the activity code of B300 for the tenth floor column works is the same code 0360). Figure 5.7 shows a sample of project activities. Uniqueness of activity code is achieved by combining the storey No. and activity code.



×												Microsoft Exce	el - Parti 🔳 🖬 🔀
1	ات	بؤالأ للتعليم	🚽 اکتب س					بات	ر تعليه	بيانات إطا	تنمبيق أدوات	ير عرض إدراج	🛪 🗗 _ 🗐 ملف تحر
: -	~			PP : A	dal	201 11					1.000 04	*.0 .00 1	
: 🗳				5 i A	lia	• 11	• B 1	Ŭ ≡=		E L	• 1 • 1/0 •	100 ÷10 1 1P- 1P	- 1 🕮 🕶 📯 🏧 🎽 🛓
	E13	-	∱ LS	_									
1	A	C	D	E	F	G	н	1	J	К	LM	N O	P Q A
2		Al Haddad	d company										
3		Gaza Al T	ofah 🔶 🖊 🚛 🖬	BACK	TO M. MENU								
4		Gaza									_		
5		Phone :	08 2814926		ENTER				BACK	O ACT.POOL			
6		Fax :	08 2814926								-		
7							PRO	IECT A	СТП	TTIES			
8							I MO.	/LCI /I	C117	THE			
9			ENTER PROJECT SATRT DATE		/1/2006								
10			ENTER PROJECT EINISH DATE	2	51412006								
11			ENTER PROSECT TIMISTI DATE		5/4/2000								
	Story	Act		1	N	AND AND	- meroper		100 100				
	Story	Pool	Activity Description	Unit	Duration	Satart	Finish	Quantity	Unit	Total			
	No	Code	Activity Description	Omit	Duration	Date	Date	Quantity	Price	Total			
12	4	0404	أعمال التجهيز	16		4/4/2006	2000	40	70	2800			
10	1	0101	أعمال الجنب	m3	2	3/4/2006	4/4/2006	600	20	10000			
15	1	0201	أعمال الدد	m2 m)	4	18/1/2006	21/1/2006	300	45	13500			
16	1	0301	خرسانة B200 لزوم باطون النظافة أسفل أفراعد المنبي واسماقة ١٠ سو	ton	2	5/1/2006	6/1/2006	50	95	4750			
17	1	0302	باطون B 250 لزوم الفراعد المنفصلة	no	6	7/1/2006	12/1/2006	60	120	7200	-		
18	1	0303	باطين B250 لزوم الأحزمة الأرضية.	0	5	22/1/2006	26/1/2006	70	145	10150			
19	1	0304	خرسانة مسلحة B300 لزوي رقاب الأحمدة	m3	5	13/1/2006	17/1/2006	80	170	13600	T		
20	1	0305	باطون B200 لزوم الأرضيات ويسمك ١٠ سو	m3	3	23/1/2006	25/1/2006	90	195	17550	-		
21	1	0306	خرسانة مسلحة B300 لزوم الأعمدة (1)	m3	5	27/1/2006	31/1/2006	100	220	22000			
22	1	0307	خرسانة مسلحة B250 لزوم السقف بسماكة ٢٥ سم (١)	m 3	25	3/2/2006	27/2/2006	110	245	26950			
23	1	0308	خرسانة مسلحة B250 لزوم جلسات وأعتاب الأبواب والشبابية (١)	m3	5	1/3/2006	5/3/2006	120	270	32400			
24	1	0401	بناء بلولة أسمنني ٢٠ *٢٠ * ٥ من مغرغ لزوم القواطع الخارجية. (١)	m2	4	28/3/2006	31/3/2006	130	295	38350			
25	1	0402	بلوظ أسمنني ٢٠ * ٢٠ * ١٠ سم مفرغ لزوم القواطع الداخلية. (١)	m 2	7	29/3/2006	4/4/2006	140	320	44800			
26	1	0501	مونة أسمنتية لزوم القصارة الداخلية على أن تتم على وجهين مع الخلط جيد (١)	m2	5	29/3/2006	2/4/2006	150	345	51750			
27	1	0502	مونة أسمنتية لزوم التصارة الخارجية على ثلاث وجوه لزوم المبنى (١)	m2	7	25/3/2006	31/3/2006	160	10	1600			
28	1	0503	مونة من الأسمنت الأبيض والكوارتز لزوم أعمال الرشقة الخارجية (١)	m2	2	1/4/2006	2/4/2006	170	15	2550			
29	1	0601	توريد وتركيب بالط كسر رخام للارغيات (١)	m2	5	19/4/2006	23/4/2006	180	100	18000			
30	1	0602	نوريد وتركيب باثط بورسان لارغياء الصاله (١)	m2	3	21/4/2006	23/4/2006	190	72	13680			
31	1	0603	وريد وتركيب بالط سيراميك لارضيات الدورات (1)	m2	2	22/4/2006	23/4/2006	200	76	15200			
32	1	0604	توريد وبرغيب بالد سيراميد تحواط الدورات (1)	m2	3	22/4/2006	24/4/2006	210	80	16800			
33	1	1401	الوريد وبرعب رهم المعالي موج فرارا مرود دخسات الدراج (١)	mi	3	22/4/2006	24/4/2006	220	84	18480			
34	1	0704	توريد ويرقيب وهم عليتي بمناطب العم وترامل ١٠ عم تبصلك العجبيك (١)	m2	2	0/4/2000	0/4/2000	230	00	20240			
35	4	0701	الوريد والمعلى المست مريد الهيد على الم الما المالية والمتحد (1)	m2	3	40/4/2000	13/4/2000	240	32	22000			
27	1	0102	لوريد و محل جيور مسي سور معالي و ويون ()) دريد وزركيت بابر كيس مقاس و ۲۱ سد ()	00	2	9/4/2000	10/4/2000	250	100	24000			
38	1	0802	کما سیش مقادی ۸۰×۲۰۰۰ سر والت (عل ۲۷ ۱۹ ۱۹۷۷ لید ۶ اشار قی (۱)	00	2	9/4/2006	10/4/2006	270	104	28080			
39	1	0803	كما سيق مقاس ٢٠٠×٢٠٠ سم والزرائيل WALLY نوع اشارة (1)	no	2	9/4/2006	10/4/2006	280	108	30240			
40	1	0804	كما سبق مفاس ٢٠٠×٢٠٠ سم والزرقيل WALLY نوع اشارة (١)	no	2	9/4/2006	10/4/2006	290	112	32480			1
14		Inter F	Face / main menu / com@nro inf / Mat Pool / Act Poo	Dro	Act / Oue o	f Mat Of Ac	/ Mat To	Order / 2	Date M	at To Order	/ Mat Ass / F	Far V. S.Curvo Cal	7 Earles St
		An Incert	Acception A comeptionin A macrool A Acception	VLIC	And A Godie	A Hac Of Mc	- A macine	STUDI A 2	Sate M	att to crubi	V Harress V F	u vi orcuivo cal	
		NUM											جاهز
•		69	MASTER FINA 🚞 🛛 Microsoft Word 🖳		🦉 بدون عنوا		Part4 📳				Part2 💐	Part	1 🖄 🛛 🛵/ 🐉

Figure 5.7: "Project Activities" sheet

E. "Activities Materials Quantities" sheet

Figure 5.8 shows a sample of "Activities Materials Quantities" sheet. This sheet consists of story number, activity pool code, activity description, unit, total quantity, materials name, and its quantity, which is required for constructing the whole quantity for each activity. Materials names and materials quantity can be obtained automatically from activity pool sheet, but after the quantities be multiplied with total quantity for each activity. The user can turn back to the main menu sheet by clicking on *Back to Main Menu* icon at the top of this sheet. Also the user can open the "materials to order" sheet by clicking on *Enter* icon at the top of the sheet.



25												Micr	osof	t Excel - Par	t1 💶	FX
:	فليمات	الأ للت	🚽 اکتب سؤ						تعليمات	بيانات إطار	دوات	دراج تنصيق أ	ض إ	لف تحرير عرا		- 8 ×
in	2				Arial	- 11		B / II		ন কাৰি দ	ΨΠ		-00		8.	A -
	11.4		S =VLOOKUP(C14:Act Pool(\$E\$13:\$G\$412)	6.EA	ISE)	900 J	2021				16.10	-B 10 100	2.0 1			-
1	114 3 C	2	B	E	EGE) F	G	н	1	J	к	L		N	0	P	Q 00-
1				<u> </u>			2 - V				67					^
3	Gaza	addad a Al T	ofah 🔶	BACK	TO M. MENU	1										
1	Gaza	9		_												
5	Fax	ne:	08 2814926		ENTER											12
7			00 10 14010													
8			ACTIVITIES MATERIAL QUANTITIES	-												
10																
11	Ac	4						0	othulth	an Motorial Dr		tiono				
12 Sto	PO	ol	Activity Description	Unit	total		1000	А	CUVIU	es material FI	uhui	uons		and the second se	1.000	
13	" Co	de			Quantity	Mat1	Qua.	Mat2	Qua.	Mat3	Qua.	Mat4	Qua.	Mat5	Qua.	e.
16 1	01	01	أعمال التجهيز	LS	40		500		\vdash							
15 1	02	01	اعمل الطر أعمل الردد	m3	300	U40	500								+	
17 1	03	01	خرسفة B200 لزوم باطون النظافة أسفل ألواعد المبنى رابسماعة ١٠ سم	m3	50	B200	50									
18 1	03	02	بغلون B 250 الزوم القواعد المنفصلة	m3	60	B260	60	Fy4200	3.6							
19 1	03	03	باطون 2011 لروم الاحرمة الارمنية خرسقة مسلحة 1900 لارمد رقاب الأعمدة	m3 m3	70	B250 B300	70	Fy4200 Fy4200	5.6		1. 17 - 13					
21 1	03	05	باطون 8200 لزوم الأرغبيات ويسمك ١٠ سم	m3	90	B200	90	Fy4200	3.6							í.
22 1	03	06	خرسانة مسلحة B300 اؤوم الأعمادة (١)	m3	100	B300	100	Fy4200	50							,
23 1	03	07	خرسانة سلحة 1250 لزوم العلق بسمانة ۲۰ سر (۱) دسلية سلحة 1260 لزمد داسات واغاني الأماني والاسانيال (۱)	m3	110	B250 B250	110	Fy4200	7.7		2			<u></u>		
25 1	04	01	بناء بلولة أسمننى ٢٠ *٢٠ *٢٠ عام مغرغ لزوم القواطع الخارجية (١)	m2	130	بلوات مغزغ ۲۰ سم	####	استنت	1.04	راس	2.1	شيد	0.5		-	
26 1	04	02	بلولة أسمنني ٢٠ ٣٠ ٣ ١ سم مغرع لزوم القواطع الداغلية (١)	m2	140	بلوڭ مغرغ ١٠ سم	####	اسطنت	1.12	رمان	2.2	شيد	0.6			
27 1	05	01	مونة اسمنتية لزوم القصارة الااخلية على ان تتم على وجهين مع الخلط جيد (٢) مسترف اسمنتية لتمد التصادة الخاصية على ان تتم على وجهين مع الخلط جيد (٢)	m2	150	اسم نت اسمانت	0.5	ريل	1.2	<u>شيد</u>	0.3					
29 1	05	03	مونة من الأسمنت الأبيض والكوارتز لزوم أعمل الرشقة الشارجية (١)	m2	170	اسمنت	0.6	ريس ريل	1.36	<u>مب</u> نبد	0.3	اسمنت ابيض ٥٠ كخ	1.4	کوارټز ۲۵ کغ	34	
30 1	06	01	تورية وتركيب بالط كسر رخام للأرضيات (١)	m2	180	ط تیرازی کسر. رخام ۳۵ ب	180	بقیل تیرازی ۲۵ سم	180	اسملت	1.1	رياس	4.5	شيد	0.5	
31 1	06	02	توريد وتركيب باط بورسلان لارغنيه الصله (١) مند ماذكير براط سراسال لأنشيك الصبات (١)	m2	190	بالطبورسلان	190	بقیل تیرازی ۲۵ سم	190	اسم نت دمار	1.1	ريعل	4.8	شيد	0.6	
33 1	06	04	تورید وترکیب باط سیرامیل لدوانط الاورات (۱)	m2	210	لا کرامیکا حوانط ۲۰*۲۰	210	استغت	1.26	40	5.3	يىيە ئىمىر	0.6			
34 1	14	01	توريد وتركيب رغام ايطلي نوع كرارا الزوم دعسات الأدراج (١)	mi	220	رخام كرارة	220	1			86 - J					
35 1	14	02	توريد وتركيب رغام خليلي بسماعه ۳ سم وعرض ۲۱ سم ليلسات الشبابيك (۱) تمييز مريد محمد الالاستغير الالام امحاد على الاقار بمارسيد ام ما علام (۱)	m2	230	رخەر خايلى بەلىسىن	230							1	-	
30 1	07	02	تورید و دمان استعلا کات اوجه سی ادین بویسید او میعدید (۱) تورید و دمان بلیور مطفی للدوانط الداخلیة وجهین (۱)	m2 m2	240	<u>بوبيمي</u> زيت	12									
36 1	08	01	تورید وترکیب باب کیس مقانن ۲۰×۲۲۰ سم (۱)	no	260	باب خشب - ٩سم	260									
39 1	08	02	كما سبق مقلس ۸۰ × ۲۰۰ سم والزرقيل WALLY نوع اشارة. (۱)	no	270	باب خشب - ٨سم	270				2 2					
41 1	08	04	کما سبق مقلس ۲۰۰۰٬۰۰۰ سم والزدار کا WALLY نوع اشارد. [۱]	no	280	ياب خشب ١١٠ سر	290				-				-	
12 1	08	05	كما سبق مقاس ٢٠٠×١٢٠ سم والزرقيل WALLY نوع اشارة. (١)	no	30	باب خشب ١٢٠ سم	30	1			8 - Z					
43 1	09	01	توريد وتركيب شبابيك الومنيوم ابيض نوع كليل وثالثه مجاري مقلن ٢٠ × ١٠٠ سم (١)	m2	50	المنبوم لزوم الشبغيك	50									
45 1	03	02	تورید وترکیب شباق الومنبود مقان ۲۰×۲۰۰ سم ۲۱] تورید وترکیب شباق الومنبود مقان ۲۰×۰۰ سم لزود الحمانات (۱)	m2 m2	90	المنيوم لزوم الشبغيك	90									
45 1	09	04	حديد حماية لزوم الشبابيك مقاص ٢٠×١٠ ١ سم (١)	m2	110	حديد حماية	110									
47 1	10	01	أعمل كهرباء أسود (۱)	Is	130	مواد کهرباء اسود	130				<u>)</u> ;					
49 1	10	02	احمال عبون () أعمال عسمن أسوة (()	1S Is	170	مواد ههرب ابیس مواد سباکهٔ اسود	170									
50 1	11	02	اعمل منتي أبيض (١)	Is	190	مواد سبكة ابيض	190				1 A					
51 2	2 03	06	غرسانة مسلمة B300 لؤوم الأعدة (٢)	m3	210	B300	210	Fy4200	105						-	
	03	07		113	230	6200	230	Fy4200	10.1		1				+	×
14 4	 MU 	Inte	r Face / main menu / com@pro.inf / Mat.Pool / Act.Po		Pro.Act λQ	ua.of Mat Of A	KC /	Mat To Order 🖌	2Date	e Mat.To Order	<u>, Ma</u>	at.Ass / Ear V.	5-Cur	ve Cal / Ear	<	. .
		NUM		1	, عنوان - الرسا	ېدون	_		_						-	جاهز
e - 4	YE 😨	Ø,	MASTER FINA 🚞 Microsoft Word 🕮	او	ېدون عنوان ۱	y	Pa	14 🗐		art3 🛍		Part2		Part1 🐏	Ĺ	y/ 🐮

Figure 5.8: "Activities Materials Quantities' sheet.

F. "Materials Needed" sheet

Figure 5.9 shows a sample of "Materials Needed" sheet. This sheet contains story number, activity pool code, activity description, unit, total duration and total quantity, start and finish dates, the period of executing the activity and the distribution of materials according to the duration of the activity. This data can be obtained automatically from project activities sheet. Materials are equally distributed along the duration days of an activity. The user can turn back to the main menu sheet by clicking on *Back to Main Menu* icon at the top of this sheet, and he or she can open the "2 days materials to order" sheet by clicking on *Enter* icon at the top of this sheet.



×											Micro	soft E	kcel - F	Part1 📕	
8	بات	كتب سؤالأ للتعلي	•				ات	لطار تعليم	ت بيايات	نصيق أدواء	إدراج ت	عرض	تحرير	🖻 ولف	- 8 ×
: 6	1 12 1		Arial		- 14 -	B Z	π ≡	= = 5	নী বি ৮ৰা-	0%	,	.00		a - 8	- A -
	A12		fe Story		and the second					1 - 2	100	-10 1 -1			
	A	С	No												
1		A111-44-4													1000
3		Gaza Al Tofal	npany 🔶 🚛 🛙	BACK TO	NI. MENU										
4		Gaza													
5		Phone :	08 2814926	EN	TER	10	•								
6		Fax:	08 2814926												
8			MATERIALS NEEDED												
9		1	MATERIALS NEEDED												
10															
	Stone	Act Dool			Total	Total	Satart	Finish	Daum	Act.	Mot 1	Mat1.	Quan.	Mat 2	Mat2.
12	No	Code	Activity Description	Unit	Dur	Quan	Date	Date	• Days در ساعد	Dates	Name	total	to	Name	total
13	110.	Couc	22.23		Dui.	Guun.	Date	Date	و, تارار		Hame	Quan	order	name	Quan
14	1	0101	أعمال التجهين	LS	2	40	1/1/2006	2/1/200	الاسی - ۱	1/1/2006					1
10	1	0201	أعمال الحفر	m3	2	500	3/1/2006	4/1/200		3/1/2006	رمل	500	250.00		<u> </u>
45		0201			-		0.112000	2		4/1/2006			250.00		
74	1	0202	أعمال الردم	m3	4	300	18/1/2006	21/1/200		18/1/2006					
75								6		19/1/2006					1
77								3		21/1/2006			<u> </u>		
104	1	0301	خرسانة، B200 لزوم باطون النظافة أسفل قواعد المبنى ر بسماكة ١٠ سم	m3	2	50	5/1/2006	6/1/200	0	5/1/2006	B200	50	25.00	-	
105			N					3	2	6/1/2006			25.00	-	
134	1	0302	باطون B 250 لزوم الفواعد المنعصلة.	m3	6	60	7/1/2006	12/1/2006	1	7/1/2006	B250	60	10.00	Fy4200	3.6
136									3	9/1/2006			10.00		1
137									4	10/1/2006			10.00		1
138									5	11/1/2006			10.00		1
139	1	0303	باطون B250 لزوم الأحزمة الأرضية	m3	5	70	22/1/2006	26/1/2006		22/1/2006	B250	70	14.00	Fv4200	5.6
165							LLINEUUU	LOINEDOD	2	23/1/2006	DLOU		14.00	.,	0.0
166									3	24/1/2006			14.00		1
167									4	25/1/2006			14.00		1
194	1	0304	خرسانة مسلحة B300 لزوم رقاب الأعمدة	m3	5	80	13/1/2006	17/1/2006	1	13/1/2006	B300	80	16.00	Fy4200	4
195									2	14/1/2006			16.00		
196									3	15/1/2006			16.00		1
197									5	17/1/2006			16.00		
224	1	0305	باطون B200 لزوم الأرغيات ويسمك ١٠ سم	m3	3	90	23/1/2006	25/1/2006	1	23/1/2006	B200	90	30.00	Fy4200	3.6
225									2	24/1/2006			30.00		
226	1	0306	خرسانة مسلحة B300 لزور الأحمدة (١)	m3	5	100	27/1/2006	31/1/2006	3	26/1/2006	B300	100	30.00	Fv4200	50
255		500	(/ - 100 0000				2111/2000	01.1/2000	2	28/1/2006	5500		20.00	. 9 .200	
256							a. /		3	29/1/2006			20.00	Land.	~
IN	4 + 1	\ Inter Face	(main menu / com@pro.inr / Mat.Pool / Act.Pool / Pro.Act /	Qua.of M	lat Of Ac (Mat To	o urder / 2	2Date Mat.	To Urder /	Mat.Ass /	Ear V. S	-curve I	a / Ea	ar I <	2
		NUM		Ave. 1			Ŷ.		e T						جاهز
¢	• 9.87	EN EN	ون عنوان - الر Microsoft Word 🛄 Microsoft Word	H 🖉		Part4		Part3 🔮		Part2 P	1		Part1 📲		<u>a</u> / 🛃

Figure 5.9: "Materials needed" sheet

G. "Materials to Order between two Dates" sheet

"Materials to order between two dates" sheet contains material code, material description, and quantity of material. Figure 5.10 shows a sample of "2 dates materials to order" sheet. The data can be obtained by entering two dates at the top of this sheet. The user can return to the main menu sheet by clicking on *Back to Main Menu* icon, and he or she can open the "Materials spaces and cash requirements" sheet by clicking on *Enter* icon at the top of this sheet.





Figure 5.10: "Materials to Order between two Dates" sheet

H. "Materials Spaces and Cash Requirements" sheet

Figure 5.11 illustrates "Materials Spaces and Cash Requirements" sheet. This sheet consists of story number, activity pool code, activity description, all materials and their quantities required for each activity, unit space, unit price, total space, and total price. From this sheet the user knows total material quantities, prices, and spaces required for each project activity. The user can return to the main menu by clicking on the *Back to Main Menu* icon at the top of this sheet, and he or she can open the "earned value S-curve calculations" sheet by clicking on *Enter* icon at the top of this sheet.



×														Micr	osoft l	Excel - Part1		×
1	6	للتغليمان	🚽 اكتب سؤالاً							ي تعليمات	ت إطار	ات بيايا	ق أدِو	دراج تنصي	عرض ا	أيلف تحرير	98	×
1	1 🖻			· · · · ·	Arial		¥	12 🗸 🚺	BII) kn -	199 9	/o , *.0	.00		- A -	
	C14	4	 fx =Pro.Act!C13 								Sector 1							
	8	С	D	E	F	G	н	1	J	к	L	10	N	0	P	Q	R	-
1 2		Al Hado	dad company															^
3		Gaza A	i Tofah 🚽 🛶 🗰 🗰	BACK TO M. I	WENU	8												
4		Gaza Phone	08 2814926	ENTER		10												-
6		Fax :	08 2814926			0												
7			ΜΑΤ ΥΡΑΓΕΥ & ΓΑΥΗ Ρ	FORFDME	MTY			-										
9 10			BLAT. SI ACES & CASH R	EQUIENNE	1410			1										
11		A.4			1			1					(1	_		<u> </u>	
12	Story	Pool	Act. Description	Mat1	Total	Unit	Unit	Total	Total	Mat2	Total	Unit	Unit	Total	Total	Mat3	Total	
13	No.	Code			Quan.	Price	e	Price	Space		Quan.	Price	Space	Price	Space		Quan.	
14	1	0101	أعمل التجهيز		1 3													
15	1	0201	أعمل المفر	(A)	500.00	15.00	0.50	7500.00	250.00									
16	1	0202	اعمال الرام مُحسقة Ponn المحيفة الثقافة أستان قداعه المنتحيسمافة علاسة	P200	50.00	220.00	2.00	11000.00	150.00				-					-
18	1	0302	بالون 200 الزور التواحد النفصلة باللون 200 الزور التواحد المنفصلة	B250	60.00	230.00	2.00	13800.00	120.00	Fv4200	3.60	3000.00	1.00	10800.00	3.60			
19	1	0303	باطون B250 لزوم الأعزمة الأرضية	B250	70.00	230.00	2.00	16100.00	140.00	Fy4200	5.60	3000.00	1.00	16800.00	5.60			i.
20	1	0304	خرسفة مسلمة B300 لزوم رقاب الأعمدة	B300	80.00	300.00	4.00	24000.00	320.00	Fy4200	4.00	3000.00	1.00	12000.00	4.00		$ \rightarrow $	
21	1	0305	باطوق B2UU لویم الاردمینات ویسطه ۲۰ سم خدسانهٔ مسلحهٔ B2OU ادم الاعمدة (۱)	B200	100.00	220.00	3.00	30000.00	270.00	Fy4200	3.60	3000.00	1.00	10800.00	3.60		\vdash	-
23	1	0307	خرسانة سلحة B300 لزوم السلف بسطة (١)	B250	110.00	230.00	2.00	25300.00	220.00	Fy4200	7.70	3000.00	1.00	23100.00	7.70			-
24	1	0308	خرسفة مسلحة B250 لزوم جلسات وأعقاب الأبواب والشبابية (١)	B250	120.00	230.00	2.00	27600.00	240.00	Fy2800	10.80	3000.00	2.00	32400.00	21.60			ł.
25	1	0401	بناء بلوك أسمنتي ٢٠٠٢، ٢٠ مم مغرغ لزوم القواطع الغارجية (١)	بلوڭ مغرغ ٢٠ سم	1690.00	2.00	2.00	3380.00	3380.00	استنت	1.04	550.00	2.00	572.00	2.08	uks	2.08	-
25	1	0402	بنونه اسمنسي ٢٠٠٣، ١ سم مترع تروم التواقع الاخلية (١) مونة أسمنتية لذود التصلية الالخلية على أن تتد على محمين مع الخلط حد (١)	بنوات معرع ١٠ سم اسمنت	0.42	1.50	2.00	231.00	0.84	(Las	0.64	15.00	2.00	352.00	1.28	ريس شيد	1.28	Ā
28	1	0502	مونة أسمنتية لزوم القصارة الخارجية على ثلاث وجوه لزوم المبنى (١)	اسمغت	0.56	550.00	2.00	308.00	1.12	iky	1.28	15.00	0.50	19.20	0.64	شيد	0.32 /	4
29	1	0503	مونة من الأسمنت الأبيض والكوارتز لزوم أعمل الرشقة الخارجية (١)	اسمئت	0.30	550.00	2.00	163.63	0.60	رامل	0.68	15.00	0.50	10.20	0.34	شيد	0.17 4	4
30	1	0601	نورية وتركيب بالطكسر رشام للارضيات [1] تسمير منابع الماريين الارتخاب المارين		4 5										-		$ \rightarrow $	-
30	1	0602	وريد وتركيب باط سراميك لارغيب المنتدر ()			1												-
33	1	0604	توريد وتركيب بالط سيراميك لحوائط الاورات (١)															
34	1	1401	توريد وتركيب رخام ايطلي نوع كرارا لزوم دعسات الأدراج (١)														$ \rightarrow $	
35	1	0701	الوزية ويركيب ركم خليلي بسماطه ۲ سم وعرض ۲۱ سم لجسات السبعين (۱) تمريد و دهلة الاستف ثلاثة اوجة على الاقار بولسيد او ماسعات (۱)				-											-
37	1	0702	توريد و دمان بليور مطنى للدوانط الااخلية وجهين (١)			-												
36	1	0801	تورید وترکیب باب کیس مقانن ۹۰×۲۲۰۰ سم (۱)		1).		i d										
39	1	0802	کما سبق مقلس ۸۰ ۲۰۰۰ سر والزرفیل WALLY نوع اشاره. [۱] کداست مقلب و ۲۰۰۰ سر والزرفیل ۲ (۱۵/۱۱ نوع اشاره (۱				-											-
41	1	0804	کما سبق مقادی ۲۰۱۰×۲۰۰ سم والزیفیل WALLY نوع اشارة. (۱)															1
42	1	0805	كما سنق مقاس ٢٠٠×٢٠٠ سم والزرقيل WALLY نوع اشارة. (١)															
43	1	0901	د وتركيب شبابيك الومنيوم ابيض نوع كليل وثلاثة مجاري مقان ١٢٠×١١٠ سم														$ \rightarrow $	
15	1	0902	هما سبق ونعن معنی ۲۰۰۰ سم (۱) تورید وترکیب شمال المغنود مقابی ۲۰×۰۰ سم (۱)				-						-		-			
45	1	0904	هديد هماية أزوم الشبابيك مقانن ١٢٠×١٢٠ سم (١)		4 3	2)			ġ.
67	1	1001	أعمل كهرباء أسود (١)															
48	1	1002	اعمال کهریاء اییش (۱) أعمال صحر آسم (۱)		<u> </u>								-					-
50	1	1102	أكمل محي أييض (١)															1
51	2	0306	خرسفة سلمة B300 لزوم الأعدة (٢)	B300	210.00	300.00	4.00	63000.00	840.00	Fy4200	105.00	3000.00	1.00	315000.00	105.00			1
52	2	0307	خرسانة سلحة 1826 لزوم السلف بسمانة ٢٠ سم [1] ثير فقر الحق 1850 لزمر طبيات ماتزار الأمار بمالاردان (1)	B250	230.00	230.00	2.00	62900.00	460.00	Fy4200	16.10	3000.00	1.00	48300.00	16.10			v
H	1+	H \ Int	er Face / main menu / com@pro.inf / Mat.Pool	/ Act.Pool / P	ro.Act	Qua.of	Mat C	fAc / M	lat To O	der / 2Date M	at.To C	rder \	1at.As	s / Ear V.	S-Curve	Cal / Ear <		
		N	JM					~									<u>ن</u> ه	5
	33.4	00	ENMASTER FINA 🦳 Microsoft Wo	. الى 🕅 🕅 ba	ون عنوان	ا 🗑 بد		Part	121	Part3			Part	2 201		Part1	1511 1	
																	-	

Figure 5.11: "Materials Spaces and Cash Requirements" sheet

I. Earned value S-curve calculations sheet

Figure 5.12 shows a sample of "Earned value S-curve calculations" sheet. It contains activity code, activity description, unit, start and finish dates, quantity, duration, unit price and total price. In this sheet, the user must enter the project start date and project finish date at the top of the sheet to obtain the total earned value for each activity. From this output information earned value S-curve is drawn and it can be used as a guideline for making a decision.

J. Earned value S-curve Sheet

It is a graph, which shows the cumulative earned value of the project against the time, (see Figure 5.13)



28											Microsoft E	ixcel - Part1 🔳 🖻 🚺
1	عليمات	◄ اكتب سؤالأ للت							إطار تعليمات	, أدوات بيانات	عرض إدراج تنسيز	: 🗗 _ 📳 مِلْف تحرير
En	PR 10			Arial		1.	B 7	п 🗆		a	0/	
				2 · · · · ·		100		9 1 =			10 / 100 -0.0 H	
	EH9	* /x		-								
1	y c	0	E		6	н		J	ĸ	LEU	EE EF EI	
2	Al Hadda	id company										
3	Gaza Al	Tofah 🗧 🗧	- 3	BACK TO	M. MENU							
4	Gaza			_								
6	Phone :	08 2814926		ENI	ER							
0	rax :	00 2014920										
7						EARN	ED VAL	. S-C	CURVE CA	ILC.		
8												
9		PROJECT SATRT DATE	1.	1/2006								
10		PROJECT FINISH DATE	25	14/2006	1							
11			_		2							
	Act.	Act Description	Unit	Duration	Satart	Finish	Quantity	Unit	Total			
12	Code	Act Description	Unit	Duration	Date	Date	Quantity	Price	Total			
13	0101	أعمال النجهين	LS	2	1/1/2006	2/1/2006	40	70	2800			
14	0201	أعمال الحض	m3	2	3/1/2006	4/1/2006	500	20	10000			
15	0202	اعمال الردم	m3	4	18/1/2006	21/1/2006	300	45	13500			
16	0301	هرسانه 2000 لزوم باطون النطالة استل لواغد المبنى ريسماهه ۲۰ سم باط ، معد جان بالا المطالة السطاليا ال	m3	2	5/1/2006	6/1/2006	60	95	4750			
17	0302	باللون B25 للروية الفواط المستعلية.	m3	6	7/1/2006	12/1/2006	70	120	1200			
10	0303	خد سالة مسلحة B200 لادد دقات الأعددة	m3	5	13/1/2006	17/1/2006	80	140	13600			
20	0304	au) - B200 - B200 - B200	m3	3	23/1/2006	25/1/2006	90	195	17550			
21	0306	خرسانة مسلحة B300 لدرد الأعددة (()	m3	5	27/1/2006	31/1/2006	100	220	22000			
22	0307	خرسانة مسلحة B250 لزود السلف بسماكة ٢٠ سر (١)	m3	25	3/2/2006	27/2/2006	110	245	26950			
23	0308	خرسانة مسلحة B250 لزوم جلسات وأعتاب الأبواب والشبابيك (1)	m3	5	1/3/2006	5/3/2006	120	270	32400			
24	0401	بناء بلوك أسمنتي ٢٠*٣٠ × ٤ مم مفرع لزيم القواطع الخارجية (١)	m2	4	28/3/2006	31/3/2006	130	295	38350			
25	0402	بلوڭ أسمنني ٢٠٣٢٠٢٠ سم مفرغ لزوم القواطع الداخلية. (1)	m2	7	29/3/2006	4/4/2006	140	320	44800			
26	0501	مونة أسمنتية لزوم القصارة الداخلية على أن تتم على وجهين مع الخلط جيد (١)	m2	5	29/3/2006	2/4/2006	150	345	51750			
27	0502	مونة أسمنتية لزوم القصارة الخارجية على ثلاث وجوه لزوم المبنى (١)	m2	7	25/3/2006	31/3/2006	160	10	1600			
28	0503	مونة من الأسمنت الأبيض والكوارتز لزوم أعمال الرشفة الخارجية. (١)	m2	2	1/4/2006	2/4/2006	170	15	2550			
29	0601	تورید وترکیب باط کسر رخام للارضیات (۱)	m2	5	19/4/2006	23/4/2006	180	100	18000			
30	0602	نوريد وتركيب با9د بورسائل لارشيه الصاله (١)	m2	3	21/4/2006	23/4/2006	190	72	13680			
31	0603	وريد ولرغيب بلاط سيراميت ورغيات الدورات (1)	m2	2	22/4/2006	23/4/2006	200	76	15200			
32	1404	الدرية وتركيب وخلا الطالب لد 9 كرارا الدرد دوسات الأبراح (١)	mz	3	22/4/2000	24/4/2006	210	84	10000			
24	1401	تروید و تروید و تروید بیشانی می تلوید این این این این این (۱) این این (۱) (۱) (۱) (۱) (۱) (۱) (۱) (۱) (۱) (۱)	m2	2	5/4/2006	6/4/2006	220	88	20240			
35	0701	توريد و دون الاستقباق الألية اوجه على الأقل بوليسيد او مايعادله (()	m2	5	9/4/2006	13/4/2006	240	92	22080			
36	0702	توريد و دمان بليور مطفى للحوائط الداخلية وجهين (۱)	m2	4	10/4/2006	13/4/2006	250	96	24000			
37	0801	تورید وترکیب باب کیس مفاس ۹۰×۲۲۰ سو (۱)	no	2	9/4/2006	10/4/2006	260	100	26000			
38	0802	كما سيق مقاس ٨٠×٢٠٠ سم والزرفيل WALLY نوع اشارة (١)	no	2	9/4/2006	10/4/2006	270	104	28080			
39	0803	كما سبق مقاس ٢٠٠×.٢٠ سم والزرقيل WALLY نوع اشارة (١)	no	2	9/4/2006	10/4/2006	280	108	30240			
40	0804	كما سبق مقاس ۲۰۰×۲۰۰ سم والزرقيل WALLY لوع النارة (۱)	no	2	9/4/2006	10/4/2006	290	112	32480			
41	0805	كما سبق مقاس ١٢٠×٢٠٠ سم والزرقيل WALLY نوع اشارة (١)	no	2	9/4/2006	10/4/2006	30	116	3480			
42	1 0901	، بد رؤ کس شیاطة ألویندو (سف، نوع کلیا، مثلاثة محادو مقاور ۲۰٬۲۱۰ سو ۱۱۱) Inter Face / main menu / comfiner inf / Mat Dool / Act	Dool	/ Pro Act	112/4/2006	lat Of ac	/ Mat To (120 Order /	2Date Mat To	Order / Mat &s	Ear V. S-Curve	Cal / Far & S
	1.00	area race Y mainmenia Y comitiporan Y watchool Y wet	001	A HOMEL)	Qualoris	INCO ME	V macini	A dei X	Lucie Macino	Cition V InduAs		
		NUM	_			- T			-			جاهز
- ص	• 0. YO 🧐	83		- الرسام	بدون عنوان	194	Microsol - واج		Microso	ift Excel - Part1 🔀	MASTER FINAL	FULL 🗁 🔰 🛓 🏄

Figure 5.12: "Earned value S-curve calculation" sheet



Figure 5.13: "Earned value S-curve" sheet



5.3.2.2 Part two (Materials purchase decision)

It includes the following sheets: "Materials to order and deliver on site" sheet; "materials must be purchased at" sheet; "materials spaces assignment" sheet; "materials prices assignment" sheet; "materials cumulative spaces sheet & prices at purchase order" sheet; "purchase order" sheet; and "materials must be purchased at" sheet. The followings are details of part two sheets:

A. "Materials to order and deliver on site" sheet

Figure 5.14 shows a sample of "materials to order and deliver on site" sheet. This sheet consists of material name, lead-time to order, and lead-time for delivery. In this sheet, in the column of "lead time for orders", the user should enter the number of days before which construction materials are to be ordered to have it available for delivery on time. The same thing he has to do for "lead time for delivery". The user can open "materials must be purchase at" sheet by clicking on *Enter* icon at the top of this sheet.



Figure 5.14: "Materials to order and deliver on site" sheet



B. "Materials must be purchased at "sheet

Figure 5.15 shows a sample of "Materials must be purchased at" sheet. Data in this sheet is obtained automatically; this sheet shows materials assignment quantities against the time for two months only. From this sheet the user will know the total quantities of construction materials needed for project activities which required to be purchased. The user can open "materials spaces assignment" sheet by clicking on *Enter* icon at the top of this sheet.

28														Mic	rosoft E	ixcel - Pa	art2 💶 🗗	X
:	اكتب سؤالأ للتعليمات	-								ات	طار تعليم	ہیانات اِ	ل أدوات	راج تنسيق	<u>عر</u> ض إد	ب ت <u>حرير</u>	غل <u>م</u> الله ملغ	9 ×
: 0		1 1 19 3	-Σ-	01 XV (0 2:	Arial		- 1.	BI	UIE		রারা⊳	r - 1 000	% , *:	-08 E	e eiz i pos	- 3 - A	
	A1 •	fx	The State Disease	A Column 1				acca -> - a					- wit	1.5 (1. 100)	1 210 1 27			
	A B	C	D	E	F	G	н	E	J	к	L	M	N	0	Р	Q	R	-
1	Al Haddad compa	DV.																^
3	Gaza Al Tofah	iy i																
4	Gaza																	
5	Phone :	08 2814926		Enter														
7	Fax:	08 2814926																
8	MATERIA	IS MUST	RE PURCH	ASEDAT														
9		20 11007	DE l'UNOIN	-020 AT														
11	POD For M	laterials	From Date	25/12/2005	ŧ.													
12	Assignment	Period Is	To Date	25/02/2006														
13	Tissigniterie	CIICAIS	TO Date	LUIULILUUU														
14	Mat Owner At		1	14 C						1								_
1.5	P O Date	رسان	حصبة	أسمنت	B150	B200	B250	B300	B350	Fy4200	Fy2800	سلكا تربيط	سلك مجدول	مسامیر ۱۰	مسامیر ۲	خشب طوبار	ست ۲۰ سم	<u>م</u>
16	(250.00																-
17	25/12/2005	250.00																
18	27) 12/ 2005 28/ 12/ 2005									0.60								_
19	30/12/2005									0.60								_
21	01/01/2006									0.60						0		-
22	08/01/2006					25.00				0.60								
23	05/01/2006 06/01/2006		-			25.00	-			0.60								_
24	07/01/2006 08/01/2006						10.00			0.80								-
26	09(0)/2006						10.00			0.80				1		6		
27	11/01/2006						10.00			0.80								_
28	05/01/2006	<u>}</u> }		1			10.00			0.80	2							
30	08/01/2006						10.00											_
31	09/01/2006																	
32	10/01/2006							16.00		1.12								-
34	12/01/2006							16.00		2.32			-					
35	13/01/2006							16.00		2.32	S					8 8		
36	14/01/2006)						16.00		2.32	1							
37	16/01/2006			22			-			10.00	-					2		-
39	17/01/2006									10.00								
40	18/01/2006					30.00	14.00			10.00								
41	19/01/2006			-	_	30.00	14.00			10.00	-					-		
43	21/01/2006					30.00	14.00			10.00								-
44	22/01/2006						14.00											
45	23/01/2006	fora OSA		ust Do Durch	+ / Mat C	Docos 8+1	100	Mat Drie	or At D.O.F	0.31		10.0	beat be-	t Po On C	to At /			11
1.	Mat.Days be	1018 - 0.5 &	FIO AMALIM	iust de Purtri. P	rt A Mat. 2	paces At I	0.0 /	mac. Pric	es al P.U.L	 X come 	n ei o op	XF.07	mail: Mus	st be Off 3	ILE AL /			
	NUM		_		an I			-						T			التصغية	وصع
ص				بدون عنوان - الرسام	<u>س</u>	• Microsi		2		Part2			Parti 🛃	MAST	FINAL -	FOLL	<u> </u>	U

Figure 5.15: "Materials must be purchased at" sheet

C. "Materials Spaces Assignment" sheet

Figure 5.16 shows a sample of "Materials spaces assignment" sheet. This sheet consists of material name and material spaces in m2 to be purchased. Data in this sheet is obtained automatically. From this sheet, the user knows the spaces required for storing construction materials daily and for two months only. Data in this sheet is used for calculations used in "materials cumulative spaces & prices at purchase order" sheet in order to calculate materials cumulative spaces. He or she can open "materials prices assignment" sheet by clicking on *Enter* icon at the top of this sheet.



×															Micro	soft Exc	el - Part2 📕 🖬	
1	كتب سؤالأ للتعليمات										ليمات	لِطار ت	ت بيايات	ىسىق أدوا	إدراج تن	بر <u>ع</u> رض	_ 📧 مِلف تحري	ð ×
in		1 I I I I I I I I I I I I I I I I I I I	∢ ⊨ Ω +	Σ × 011 :	×v. • @	Mi : Aria	al l		1.1.1	B 7 U	1 📰 🗖		ि जि आ प	0/0	, *.0	-00 E E		-
	Δ1 -	f.	× 1 6 1					80						1 25	.00	9.0 - , ,		
A		C	D	E	F	G	н	E	J	К	L	M	N	0	P	Q	R	-
1	1												10					^
2	Al Haddad compar	iy .																
3	Gaza Al Toran Gaza																	
5	Phone :	08 2814926		Enter														
6	Fax:	08 2814926																
7																		
9	MATERIA	LS SPACI	ES ASSIC	GNMENT														-
10																		-
11																		
13																		
14	M. C. MI	_						_					1				(<u> </u>	
28	Mat . Space At	رمان	حصمة	اسمنت	B150	B200	B250	B300	B350	Fy4200	Fy2800	سللة تربيط	سلق مجدول	مسامير ١٠	مسامیر ۲	خشب طوبار	بلوڭ مصمت ٢٠ سم	e
15	P.O. Date	435.00								_	_				-			÷.
10	25/12/2005	125.00										-		2 8	-			+-
18	27/12/2006	TEORO				-				0.60					-			
19	29/12/2005									0.60								
20	31/12/2005									0.60								
21	02/03/2006									0.60								
22	01/03/2006					75.00				0.60							l	-
23	06/01/2006		2			15.00				0.80	-			2 3	-			
25	07/01/2006 08/01/2006						20.00			0.80					-			
26	09/03/2006 30/03/2006						20.00			0.80								
27	11/01/2006						20.00			0.80							l	
28	06/01/2006						20.00			0.80								
29	07/01/2006						20.00		1			1			-		l	+-
31	09/01/2006		-			-	20.00	2						s - 8	-			
32	10/01/2006		8		8			64.00				8		8 8				
33	11/01/2006							64.00		1.12					2			
34	12/01/2006					-		64.00	1	2.32					-			
35	13/01/2006							64.00		2.32							l	+-
37	15/01/2006							01.00		1.12		-		-				-
38	16/01/2006					1				10.00					1			
39	17/01/2006							-		10.00				S				
40	18/01/2006					90.00	28.00			10.00		2		<u> </u>	-			
41	19/01/2006		<u> </u>			90.00	28.00			10.00		-		2	-		l	-
43	21/01/2006					30.00	28.00			10.00		-		-				+-
44	22/01/2006						28.00											
45	23/01/2006									0.31								
46	24/01/2006							80.00		0.31								~
14 4	H Mat.Days E	Before - O.S	& P.O /	Mat. Must Be Pu	urch. At	Mat. Spa	ces At F	P.O.D / M.	at. Prices A	t P.O.D /	(comul. F	Pr&Sp /	P.O/ M	at. Must B	e On Site	e At /		
	NUM																تصفية	وضع ال
، ص	ori 9, 9 %		1	<u>وران -</u> الرسام	🙀 بدون عد	واجها	Microsof	it Word 🗐		6	Part2			art1 🗐	MASTER	FINAL - FUI	L 😂 🛛 İsil	-
						-					-							-

Figure 5.16: "Materials Spaces Assignment" sheet

D. "Materials Prices Assignment" sheet

Figure 5.17 shows a sample of "Materials Prices Assignment" sheet. Data in this sheet can be obtained electronically, such as dates between start, and finish dates for the two months, which entered in "materials to order and deliver on site" sheet, materials and prices required. Data in this sheet is used for calculations used in "materials cumulative spaces & prices at purchase order" sheet to calculate materials cumulative prices. The user can open" Materials Cumulative Spaces & Prices at Purchase Order" sheet by clicking on *Enter* icon at the top of this sheet.

E. "Materials Cumulative Spaces & Prices at Purchase Order" sheet

Figure 5.18 shows a sample of "materials Cumulative Spaces & Prices at Purchase Order" sheet. It contains purchase order dates, materials total space, materials total price, and materials cumulative space and price. The program provides data in this sheet automatically. The purpose of this sheet is to help the user for determining a period of time to buy construction materials for this period according to the available space and the available cash money. So the user can release a purchase order for this period by typing the start and finish dates of this period in purchase order sheet. He or she can open "Purchase Order "sheet by clicking on *Enter* icon at the top of this sheet.



24																	Micros	oft Excel - Pa	rt2 💶 🖻 🔀
8	بات	🚽 اكتب سؤالاً للتعليم											ليمات	إطار تع	ات بیانات	سىق أدو	إدراج تئه	ف تحرير عرض	
: -	2 T		124			1 M. C		1 Adal			1				Da Lee	1 500 07			
: 💷			••• •	1-) -	2 * 2	4 AT		Ana				B <u>1</u> <u>U</u>	= =		<u></u> 11 •		, .00 →.		· 🍊 · 🚘 · 👳
-	A1	≠ fx					-											-	
1	A	В	С	D	E	F	G	н		J	K	L	М	N	0	P	0	R	s A
2		Al Haddad compar	1 y			Microsoft Excel - Put2 													
3		Gaza Al Tofah																	
4		Gaza					0.000												
5		Phone :	08 281 492	26		Enter													
6		Fax:	08 281 492	26															
8		MATEDIALSO	DICES	15510	MARENT	1													
9		WATERIALS P	RICED I	433/6	Image: Second Excel Part Part Part Part Part Part Part Part														
10						Шислозани Excel - Part 2 2<													
12							B150 B200 B250 Fyd200 Fyd200												
13																			
		Mat Price At																	
		P O Date	رمل	حصمة	اسمنت	B150	B200	B250	B300	B350	Fy4200	Fy2800	سلك نربيط	سلك مجدول	مسامير ١٠	مسامیر ۲	خشب طوبار	لوڭ مصمت ۲۰ سم	مفرغ ۲۰ سم ،
10		(demail 6	3750.00																
17		25/12/2005	3750.00		-		-	-					-					-	
18		27/12/2006									1800.00								
19		29/12/2005									1800.00								
20		31/12/2005									1800.00								
21		02/01/2006									1800.00		-					-	
22		01/01/2006		<u> </u>		<u> </u>	5500.00			-	1800.00								-
23		05/01/2006 05/01/2006					00.00				2400.00								
25		07/01/2006 08/01/2006		-	-		-	2300.00			2400.00	-	-		-	-		-	
26		0%03/2006 10/03/2006						2300.00			2400.00								
27		11/01/2006						2300.00			2400.00								
28		06/01/2006						2300.00			2400.00								
29		07/01/2006					-	2300.00		-		-							
30		00/01/2006	<u> </u>		<u> </u>	<u> </u>		2300.00						<u> </u>		<u> </u>			
32		10/01/2006					-		4800.00										-
33		11/01/2006							4800.00		3360.00								
34		12/01/2006							4800.00		6960.00								
35		13/01/2006							4800.00		6960.00								
36		14/01/2006							4800.00		6960.00								
37		15/01/2006	<u> </u>	<u> </u>						-	3360.00								-
30		17/01/2006									30000.00								
40		18/01/2006					6600.00	3220.00			30000.00					-			
41		19/01/2006					6600.00	3220.00			30000.00		1		1				
42		20/01/2006					6600.00	3220.00			30000.00								
43		21/01/2006						3220.00											
44		22/01/2006					-	3220.00		-	024.00		-						
40		23/01/2006					-		6000.00		924.00								
14		Mat.Days Befor	e - 0.5 &	P.O /	Mat. Mu	st Be Purch.	At / Ma	t. Snaces	At P.O.I	Ma	t. Prices /	At P.O.D	(comul.)	Pr & Sn /	P.O./ M	at. Must I	Be On Site .	At /	n .
		NI IM													. = /				د داران مذهر العرب فراق
				_			and T				-			_		Cons. I			وصغ التصعيب
ص	- 0. TT	0,0 3			بام	ون عنوان - الره	• 🕲 .	M واجها.	icrosoft W	ord 🖭			Part2		Pa	n1 🗐	MASTER F	INAL - FULL 🚞	<u> </u>
-																-			

Figure 5.17: "Materials Prices Assignment" sheet



Figure 5.18: "Materials Cumulative Spaces & Prices at Purchase Order" sheet



F. "Purchase Order" sheet

Figure 5.19 shows a sample of "purchase order " sheet. This sheet consists of start and finish dates for purchase order, materials name, materials quantities, and unit. In this sheet, the user can obtain a purchase order for any period of the construction project by typing the start and finish dates in yellow colored cells. The user can open "Materials Must Be Purchased at" sheet by clicking on *Enter* icon at the top of this sheet.

28												Micros	soft Excel	- Part2	. 6 🗙	1
:	لأ للتغليمات	🚽 اکتب سۋا						عليمات	اِطْار ت	بيايات	ييق أدِوات	إدراج تنم	يرير عرض	🖻 ولف ت]_ 8 ×	c.
:	🗃 🖬 🖪		∛ ∽) • Σ • ≙↓	%¥0 🗸 🕢 🚰 i Aria	ŗ	- 1-	• B I	u I 📰 🚍	-] >n	•I 🛒 %	, *.0 .0	8 1 1 1	- 🗄	• A •	-
	F16	▼ fx														1
1	A B	<u> </u>	E	F	G	н	1	J	K	L	M	N	0 P	0	R	í
2	Al Haddad	company														
4	Gaza	лап														
5	Phone :	08 2814926		Enter												
7	Fax.	06 26 14920														
8	-	PURCHAS	ORDER													
10		20.0.5.	E	04/04/0000		-	054010	005								
11	Remark	P.O. D. For	From Date	01/01/2006	ls	-rom Dat	e 25/12/2	005								
12		r chou	To Date	01/05/2000		To Date	2310212	.000								
14	Pun	chase Order	From Date ??	25/12/2005												
15	run	chase Order	To Date ??	22/02/2006												
16	-			<u> </u>												
18		Material I	Name	Quan Required 🔻	Unit											
19 23		رمان B200		500.00	m3 m3											
24		B250		346.00	m3											l
25	-	E300 Fy420)	390.00	ton											
28		Fy2800)	2.16	ton											
94																
96 97																
98 99																
100																
102																
104																
106																
107																
109																
111																
113																Į.
115						1	_									ł
14 4	Mat Mat	t.Days Before - O.S	& P.O / Mat. Must Be	e Purch. At 🏑 Mat. Spac	es At P.O.D	🖌 Mat. Pri	ces At P.O.E) 🖌 comul. P	r&Sp)	P.0/	Mat. Must	Be On Site	At /		: الل بر الروب مر	1
	1 d4 🚳 Ø	AB Mierre	soft Word 🕅	Long Alay MM	Part4 PD		Parts M1		Paul2 Bill	1	Paul	R 1	75 , naster 13 Eul	ی 6 سجل م	نم العتور علم العالي (1
. 60	1.01 0 00	Auto - Autorio		34 20 4 W	1 31(4) 20		rana 🔡		ranz 🔡		Falt		naster i o rui			4

Figure 5.19: "Purchase Order" sheet



G. "Materials must be on the site at" Sheet

Figure 5.20 shows a sample of "Materials must be on the site at" sheet. CMMS provides data in this sheet automatically. This sheet consist of materials name, materials quantities. The purpose of this sheet is to help the user in controlling and managing construction materials. The user can return to main menu sheet by clicking on *Back to Main Menu* icon at the top of this sheet.



Figure 5.20: "Materials must be on the site at" sheet

5.3.2.3 Part Three (Materials Card)

Part 3 has one sheet called "materials card" sheet. Figure 5.21 shows a sample of this sheet. It shows the distribution of a construction material on project activities. The user can choose the name of construction material from a list, and then the software will show the distribution of this construction material and its quantities. The user can return to main menu sheet by clicking on *Back to Main Menu* icon at the top of this sheet.





Figure 5.21: "Materials Card" sheet

5.3.2.4 Part 4 (Waste Control)

Part four has one sheet called "waste control" sheet. It consists of story number, activity pool code, activity description, unit, total duration and total quantity, start and finish dates, the stipulated activity duration in days, distribution of materials according to the duration of activities, earned quantity, consumed quantity, and percentage of waste. In this sheet, the user should fill earned and consumed quantities in earned and consumed columns day by day, and then the program calculates waste percentage for all materials as follows:

Waste percentage % = (consumed quantity - earned quantity)*100 / earned quantity.



The user should calculate earned and consumed quantities before filling it in the program as follows:

Earned quantity = quantity achieved to date. – Previous achieved quantity.

Consumed quantity = quantity on site (Q) + IN quantity – Left quantity

Quantity on site (Q): Quantity available before.

IN quantity: New quantity brought to the construction site this day.

Left quantity: The quantity left at the end of the day.

×													Mic	rosoft E:	cel - Part		X
1	ات	نب سؤالاً للتغليم						÷	ي تيعليمان	يانات إطار	دوات ہ	مىيق	راج ت	عرض إد	مِلف تحِرير	🖲 - 6	4 >
1 E	1 12		🐧 🕞 🚜 + 🕩 🖉 - 🔍 - 🕺 🖓 Ai	ial		- 11	- B	U	= =		M + 1	%	, *.0	.00		A - C	-
-	N44		£ 100	197		and 1	and P. Stores										1080
-	A	8	C C	D	E	F	G	н	1	J	К	L	м	N	0	P	-
1			Ne station of the second se				1							20	2		1
2		Al Haddad co	mpany														
4		Gaza		Back	(To M. M	enu											
5		Phone :	08 2814926														
6		Fax :	08 2814926														
0			WASTE COMPOS	ľ.													
9			WASIE CONTROL													_	
10																	
	-			1	r	1			Dave	Act		Mat1	Quan			1	T
12	Story	Act.Pool	Activity Description	Unit	Total	Total	Satart	Finish	No	Dates	Mat.1	total	to	Earned	Consumed	Waste	
13	No.	Code			Dur.	Quan.	Date	Date		Datoo	Name	Quan	order	Quantity	Quantity	%	
14	1	0101	أعمال التجهيز	LS	2	40	1/1/2006	2/1/2006	1	1/1/2006	<u> </u>					i i i i i i i i i i i i i i i i i i i	Î
15									2	2/1/2006							
44	1	0201	أعمل النفر	m3	2	500	3/1/2006	4/1/2006	1	3/1/2006	(h)	500	250.00	100.00	120.00	20%	
40	1	0202	ault dari			200	19/1/2006	24/4/2006	2	4/1/2006	<u> </u>		260.00	280.00	240.00	-14%	ł
75		0202	proje dower	1115	1.1	500	10/ 1/2000	21/1/2000	2	19/1/2006							
76									3	20/1/2006	1						
77									4	21/1/2006	0000		05.00	20.00	25.00	4786	4
104	1	0301	خدسانة: R2∩∩ لاهد باطهن التظلاف استان فهاعد المشرد بسمانه ۲۰ سد	m3	2	50	6/1/2006	6/1/2006	2	6/1/2006	6200	50	25.00	25.00	26.00	4%	
134	1	0302	باطون B 250 الزوم القواعد المنفصلة	m3	6	60	7/1/2006	12/1/2006	1	7/1/2006	B250	60	10.00	i i		6	i
135									2	8/1/2006			10.00			<u> </u>	
136									3	9/1/2006			10.00			· · · · ·	
138									5	11/1/2006			10.00				
139									6	12/1/2006			10.00				1
164	1	0303	بطون B250 لزوم الاحزمة الارضية	m3	5	70	22/1/2006	26/1/2006	1	22/1/2006	B250	70	14.00	15.00	12.00	-20%	-
166									3	24/1/2006			14.00				
167									4	25/1/2006			14.00				
168				<u> </u>	<u> </u>		40,4 00000	1711 0000	5	26/1/2006	<u> </u>		14.00	Į		-	Ļ
194	1	0304	يونسييه وسييته (B300 يوقع ليهم يعيدوه	ma	0	80	13/1/2006	17/1/2006	2	13/1/2006	8300	80	16.00			-	
196									3	15/1/2006			16.00			-	
197									4	16/1/2006			16.00				
198	1	0205	the firms which we will poor a date			90	23/4/2008	25/1/2006	5	17/1/2006			16.00			-	ł
225		0.000	ha 1. annia arin()1,123, 8200, 9381	1113	l °	30	20/1/2006	20/ 1/2006	2	24/1/2006	6200	30	30.00				1
226									3	25/1/2006			30.00				1
254	1	0306	خرسقة مسلحة B300 لزوم الأعمدة (١)	m3	5	100	27/1/2006	31/1/2006	1	27/1/2006	B300	100	20.00	-		-	-
255									2	28/1/2006			20.00				-
257									4	30/1/2006			20.00				1
258									5	31/1/2006			20.00				
284	1	0307 Waste Co	خريمةة مسلحة B250 لزوم المقف سمعة ٢٥ سر (١) ما ا	m3	25	110	3/2/2006	27/2/2006	1	3/2/2006	B250	110	4.40				
		All MA	and a g					1.20									0.1
		MUM		_		-	Sector providences	and a strength				-	-14			التصغيه	وصع
	• 9:09	8 9	Part4 \min		Part1		MASTER FI	NAL - FUL 🕻	J	Aicrosoft Wo	rd • 1111	픤	مام	عنوان - الرس	🖉 بدون	<u> 4</u>]	Н

Figure 5.22: "Waste control" sheet



5.3.2.5 Help and Method of Use

Figure 5.23 shows a sample of "help and method of use" sheet. CMMS consists of four files, input data and basic calculations, materials purchase decisions, material cards, and waste control. The purpose of "help and method of use" sheet is to guide the user in using CMMS efficiently. It is written in Arabic to be more helpful and to suit Gaza strip building contractors. (See Annex 8)



Figure 5.23: "Help and method of use" sheet



5.4 CMMS Evaluation

5.4.1 Evaluation objectives

Face validity is used as a test for model evaluation. Face validity is asking wellexperienced people about the system whether the model and or its behavior are reasonable (Sargent 2000). CMMS was evaluated by this method, where four contractors were approached to apply CMMS in four different on going projects. Then they were asked to fill a questionnaire to evaluate the performance of CMMS. The objectives of CMMS evaluation are: (1) to assess the performance of construction materials management CMMS tools and techniques; (2) to check the suitability of CMMS design and structure; (3) to know the difficulties that faced the users during applying CMMS; (4) to recognize the contractors` criticisms or comments on the software and (5) to explore CMMS advantages.

5.4.2 Evaluation methodology

CMMS was tested in four under-construction projects belonging to four different contractors. All the four projects were in Gaza. The first project is a two-story building for the ministry of health (surgery department). The second project is a five floor building in Al- Remal area. The third and fourth buildings are for transportation department in security forces complex (Al- Saraya).

5.4.3 Evaluation result and discussions

Table 5.1 illustrates the contractors' responses to performance of CMMS tools and techniques. It is clear that all respondents have positive attitudes towards CMMS regarding the performance of many techniques especially "Actual cost report for project activities at any date", "Needed storage area report for project materials at any date", "Earned value S-Curve", and "Provide a card for any material and its assignment on projects' activities".

Table 5.2 illustrates the contractors` responses to the features of CMMS design and structure. The results illustrate that all contractors companies agreed that the reports and outputs are clear, and easy to read and understand.

Also it is noted that three of contractors companies strongly agreed that the software is flexible, the data can be updated easily, text and numbers shown are concise, and their sizes are suitable and readable, and the method of use is understandable. Not that some of the respondents did not answer some questions.



No	Techniques	Strongly agree	Agree	Intermediately agree	Weakly agree	Very weakly agree
1	Needed storage area report for project materials at any date.	3	1			
2	Required quantities of materials report project activities at any date.	2	1			
3	Actual cost report for project activities at any date.	4				
4	Required materials report to be purchased at any date.	2	1			
5	Required materials, to be available on site.	1	2			
6	Earned value S-Curve.	3	1			
7	Waste monitoring report for activity materials.	1	2			
8	Provide a card for any material and its assignment on projects` activities.	2	2			

 Table 5.1: The contractors` responses to performance of CMMS tools and techniques



•

No	Techniques	Strongly agree	Agree	Intermediately agree	Weakly agree	Very weakly agree
1	The software is flexible, and the data can be	3	1			
	updated easily.					
2	In general, it is easy to use.	4				
3	It saves time and effort.	1	2			
4	Method of entering data is easy and clear.		3			
5	The reports and outputs are clear, and easy to read and understand.		4			
6	Method of sorting data is easy.	1	3			
7	Text and numbers shown are concise, and their sizes are suitable and readable.	3	1			
8	The information can be inquired easily.	2	2			
9	The method of use is understandable.	3	1			
10	It is easy to handle as it is developed within Excel environment.	2	1			
11	Training to use the CMMS is easy and it does not need much time. In addition, it does not need a professional user to deal with it.		2			
12	It can be applied for most of Gaza strip projects. (It is suitable for Gaza strip contractors).	3	1			
13	It contributes in improving the construction materials management practice in Gaza strip.	1	2			

 Table 5.2:
 The contractors` responses to the features of CMMS design and structure.

Regarding the difficulties that the contractors were facing during the use of CMMS, almost all contractors said that they had not faced real difficulties.

As for the criticisms or comments of the contractors, there are two main comments. The first is that materials quantities per one unit in the "materials pool" sheet have to



be determined by experience. The second comment is the difficulty in determining the consumed and earned quantities for any materials used in the construction site day by day in the "waste control" sheet to determine waste percentage.

On the other hand, the main advantages expressed by contractors are:

CMMS is easy to use and flexible; it helps the management staff in managing the construction materials in future projects; also, it helps the management staff to decide when and how much of materials quantities to be bought; and CMMS input-output relationships are reasonable.

Overall, the results of evaluation indicate that CMMS tools and techniques are encouraging. Also, the contractors are generally satisfied with the design and structure of CMMS. Moreover, the contractors mention that a CMMS suite the Gaza strip contracting companies and has the potential to contribute in improving the construction materials management practice in Gaza strip.

The researcher suppose that this edition of CMMS must use by contractors in the construction site, and another edition of this software after testing and evaluating will be used on home office and attached them together with website.

5.5 Software Limitations

Some tools and functions are used in developing CMMS such as "macro" tool and "v-lookup" function. The security level in MS Excel should be low to make full use of CMMS.

The researcher tried to make use of the full capacity of Ms Excel 2003. However, the following are the limits, which directly, affect the function of CMMS:

- Maximum number of project activities in Project activity sheet is 65,536 (affected by the capacity of Excel 2003).
- Maximum number of materials in materials pool sheet is 65,536 (affected by the capacity of Excel 2003).
- The number of materials that the user can enter in activity pool for each activity is five materials.
- Maximum duration for one activity is 30 units (days, weeks, months).

To use CMMS for a new project all yellow cells must be deleted by determining them and press "0" then presses Ctrl and Enter



CHAPTER 6

CONCLOUSION AND RECOMMENDATIONS

6.1 Introduction

Construction Materials Management Software (CMMS) was developed to satisfy some needs of Gaza strip contracting companies in managing construction materials. This chapter introduces the research conclusions and recommendations for many parties involved in the construction process to improve the local practices in construction materials management. Recommendations for further studies are also included.

6.2 Conclusions

- The survey results show that contactors, in general, are interested in using many tools of managing construction materials. However, Most contractors did not actually apply some tools and techniques of construction materials management such as:
 - Creating database for materials categories, local suppliers, international suppliers, and materials cost.
 - Updating database for local suppliers, international suppliers, materials cost when change, and using internet for knowing the new materials and its prices.
 - Providing a list of materials in project, providing material cards at site store, and recording the received materials on site.

Even the few contractors who used the above-mentioned tools and techniques, they applied these tools either without recording at all, or with recording in an unsystematic way without using manual or computerized forms.

• All surveyed contractors believe that the Israeli closure on Gaza strip is the main element that affects construction materials availability and cost. In addition, the Israeli closure affects clearly the prices of main materials such as cement, aggregates, and reinforcement steel.



- Most contracting companies manage construction materials using non computerized forms. Shortage of suitable construction materials management software is considered the main obstacle to computerize materials management processes. Another important factor is lack of qualified persons in using computerized construction materials management packages.
- There is a consensus amongst contractors on the importance of using a computerized construction materials management system. The main advantages that can be obtained from using materials management software are:
 - Reducing the cost of materials needed in the project.
 - Better handling of materials.
 - Materials be on the construction site in time and with required quantities.
 - More effective waste control.
 - Improving follow up and monitoring construction materials.
- Gaza strip contractors did not use any software to support project materials management. This gave the researcher a thrust to develop a computerized construction materials management system that supports and improves this practice. The researcher names this software "construction materials management software" (CMMS). It was designed within Ms Excel environment where the results show that Ms Excel is the most used software by the contracting companies in Gaza strip.
- CMMS suits Gaza strip contracting companies and has the potential to contribute in improving the construction materials management practice in Gaza strip. It has a good performance and adequate accuracy.
- CMMS provides many reports such as materials to order between two dates, materials assignments, waste control, when to purchase construction materials, when materials must be on the site, and purchase order between two dates.



• CMMS provides the mechanism to decide when to buy construction materials and what quantities of construction materials the contractor need in the project.

6.3 Recommendations to the parties involved in construction

Top management of contracting companies is invited to encourage development and using construction materials management systems. They can make incentives for their staff members to attend training courses in construction materials management and its applications. They should be encouraged to actually use computerized construction materials management systems to save effort and time, and to achieve more accurate results.

Public employers can contribute in improving the current construction materials management practices of the contractors by requesting them to implement construction materials management systems during construction. This could be done by adding relevant clauses in the project conditions of contract.

Universities, contractors union, and engineering association have to do more efforts to improve the existing construction materials management practices, which may include:

- Encouraging the contractors to use construction materials management systems by addressing the importance of these systems.
- Helping the contractors to understand the system by initiating training courses, lectures, seminars, and workshops.
- Transferring of technology and experiences of other countries in the construction materials management field and adapting them to suit Gaza strip contractors

6.4 Recommendations for further studies

CMMS is a step along the way to establish a systematic construction materials management practice amongst Gaza strip contractors. Of course, it needs continuous modification and enhancement. The followings are some points, which need further research efforts:



- CMMS does not deal specifically with materials waste on the construction site. So, researchers are invited to put more effort to handle this aspect.
- The user can obtain the list of activities and all information involved in the project activities sheet by typing or copying and pasting these information from Microsoft Project software. This is a manual way to obtain data. So, researchers are invited to fully integrate Microsoft Project with CMMS to obtain data automatically.
- Different construction processes need more attention for study and research in order to determine more realistic waste percentage for construction materials. The resulting information will be helpful in determining required quantities for each activity.
- CMMS divides all construction materials required for each activity equally at the duration of the activity on "materials quantities of activities" sheet. This is not realistic in all cases, and the user may want to modify these quantities. Therefore, researchers are invited to handle this case.
- Researchers are invited to develop integrated packages that include CMMS. Integration can be approached at various levels such as:
 - Integrating materials management, materials control and monitoring functions.
 - Integration with a scheduling application packages such as MS-Project.
 - Integration with other software programs such as the software which is used in the construction storages.
- CMMS can be improved by adding to it some communications features that enable it to be used on-line between the site and the head office (See annex 7).
- There is a chance to alter the method of gathering site data such as materials delivered to the construction site. The proposed method is collection the data by using an electronic device instead of filling a paper form. Now there are global research studies to automate monitoring process to calculate the inprogress quantities. Researchers use cameras, electronic sensors, and other devices to capture the data of progress and transfer them to computer electronically (See annex 7).

Researchers are invited to develop a new version of construction materials management software (CMMS) with probabilistic capabilities



References

Ammer, Dean S. (1974) Materials Management, Richard D. Irwin, Inc. Homewood, Illinois

Ammer, Dean S. (1980) Materials Management and Purchasing, Richard D. Irwin, Inc. Homewood, Illinois

Arnold, J.R. (1991) Introduction to Materials Management, Prentice-Hall, Englewood Cliffs, New Jersey

Ahuja, Hira, N. and S, P, Dozzi (1994) Project management techniques in planning and controlling construction projects, Second Edition, John Wiley& Sons Inc., New York.

Al Mogany, S., (2006). Managing and Minimizing Construction Waste in Gaza Strip, pp 242. Unpublished Msc Thesis. The Islamic university of Gaza (IUG).

Al Ostaz, M., (2004). A cost Monitoring System for Gaza Strip Contracting, Unpublished Msc Thesis. The Islamic University of Gaza (IUG).

Bailey Peter and Farmer, David (1982) Materials Management Handbook, Gower Publishing Company Limited, Aldershot, Hants, England.

Ballot, Robert B. (1971) Materials Management: A Results Approach, American Management Association Inc., United States of America.

Ballard, Glenn and Gregory Howell, (1994) "Implementing Lean Construction: Stabilizing Work Flow." Conference on Lean Construction, Santiago, Chile.

Beekman-Love, G.K. (1978) Materials Management, Martinus Nijhoff Social Sciences Division, Boston.

Bernold, Leonhard E., and Treseler, John F., (1991) Vendor Analysis for Best Buy in Construction, Journal of Construction Engineering and Management, Vol.117, No. 4, December 1991, pp. 645-658.

Cavinato, Joseph, L. (1984) Purchasing and Materials Management, West Publishing Company, Minnesota.

Chandler, Ian E. (1978) Materials Management on Building Sites, The Construction Press Ltd, Lancaster, England.

christofferson , J. (1999) . Using Powerful Spreadsheet Application Tools to Increase the Efficiency and Effectiveness of Estimating , ASC Proceeding of the 35th Annual conference , California Polytechnic Stat University San Luis Obispo , California .

Chen, Z., Li, H., and Wong, C.T.C., (2002). An Application of Bar-Code System for Reducing Construction Wastes. Elsevier Science, Automation in Construction 11, pp 512-535.



Construction Industry Institute (1988) Project Materials Management Primer, Publication 7-2, Bureau of Engineering Research, The University of Texas at Austin.

Construction Industry Institute (1999) Procurement and Materials Management: A Guide to Effective Project Execution, Implementation Resource 7-3, Bureau of Engineering Research, The University of Texas at Austin.

Damodara, K., (1999) Materials Management: The Key to Successful Project Management, Journal of Management in Engineering, Vol.15, No.l, January/February 1999, pp. 30-34.

Dahman, M., Abo Modalala, M., Al Shabban; A. and Mohmad, O., (2004) Construction Waste Minimization. Bachelor researcher, Islamic University-Gaza.

Dobler, Donald W., and Burt, David N., (1996) Purchasing and Supply Management: Text and Cases, McGraw-Hill Series in Management, The McGraw-Hill Companies Inc, USA.

Dos Santos, A., Formoso, C.T. and Hinks, J. (1996) Method of intervention on the flow of materials in building processes, in: Proceedings of the Fourth Annual Conference of the International Group for Lean Construction, University of Birmingham, Birmingham, UK. Also at: http://web.bham.ac.uk/d.j.crook/lean.

Ericsson, Dag (1974) Materials Administration, McGraw Hill Book Company (UK) Limited, England.

Enshassi, A., (1996). Materials Control and Waste on building sites, Building research and Information, vol. 24, No. 1, pp 31-34.

Ekanaglee, l., l. and Ofer., G., (2004). Building waste assessment design-based tool. Elsevier science, Building and Environ meet 39, pp 351-861.

Gossom, W.J. (1983) Control of Projects, Purchasing, and Materials, PennWell Publishing Company, Tulsa, Oklahoma.

Howell, Gregory, Alexander Laufer and Glenn Ballard, (1993) "Uncertainty and Project Objectives" in Project Appraisal. Guildford, England: March..

Howell, Gregory and Glenn Ballard, (1994) "Lean Production Theory: Moving Beyond 'Can-Do'." Conference on Lean Construction, Santiago, Chile. September.

Huovila, Pekka, Lauri Koskela and Mika Lautanala, (1994) "Fast or Concurrent-The Art of Getting Construction Improved." Conference on Lean Construction. Santiago, Chile.

Naoum , S .(1998) . Dissertation Research and Writing for Cons traction students , Oxford : Butterworth – Heinemann .



Navon, R., and Berkovoich, O., (2005). Development and On Site Evaluation of Automated Materials Management and Control Model. Haifa, Israel.

Plemmons, James K., and Bell, Lansford C. (1995) Measuring Effectiveness of Materials Management Systems, Journal of Management in Engineering, Vol. 11, No.6, November/December 1995, pp. 26-32.

Palestinian Contractors Union (PCU), (2006). Contracting Companies and Classifications sheet. Received (15/5/2006).

Polit , D.F., Hunger , B.P., (1999) . Nursing Research Principles and method . 6th edition. Lippincott Williams and Wilkins . Philadelphia .

Stukhart, G., (1995) Construction Materials Management, Marcel Dekker Inc. New York.

Stukhart, G. and Bell, L.C. "Costs and Benefits of Materials Management Systems,", ASCE Journal of Construction Engineering and Management, Vol. 113, No. 2, June 1987, pp. 222-234.

Thabet, W. (2001), A Framework for an Integrated Materials Management System, Research Proposal Submitted to the Electrical Contracting Foundation, Inc.

Tersing, Richard J., and Campbell, John H. (1977) Modern Materials Management, North-Holland Publishing Company, Amsterdam, The Netherlands.

Thomas, H. Randolph, and Sanvido, Victor E. (2000) Role of the Fabricator in Labor Productivity, Journal of Construction Engineering and Management, Vol.126, No.5, September/October 2000, pp. 358-365.

Willis, E. (1986) Scheduling Construction Projects, John Wiley and Sons.

Womack, James P., Daniel T. Jones and Daniel Roos (1991). The Machine That Changed The World: The Story Of Lean Production. New York. 1st Harper Perennial Edition.

Weisberg ,H.F., and Bowen ,B.D.,(1997) . An introduction to survey research and data analysis . W.H. Freeman company . New York USA .

Koskela, Lauri. (1992), "Application of the New Production Theory to Construction." Technical Report #72, Center for Integrated Facilities Engineering, Stanford University. May.

Ohno, Taichi. (1987) Toyota Production System. Productivity Press.

Yaghmaie, F., (2003). Content Validity and Its Estimation. Journal of Medical Education.



LIST OF ANNEXES

- Annex 1: The questionnaire (In Arabic).
- Annex 2: The questionnaire (English Version).
- Annex 3: A review of Construction Materials Management Software (CMMS)
- Annex 4: A sample of activities list.
- Annex 5: System evaluation questionnaire (In Arabic).
- Annex 6: System evaluation questionnaire (English Version)
- Annex 7: Recording incoming of construction materials device
- Annex 8: The average waste percentage of construction material in different countries.



Annex 1

The Questionnaire (In Arabic)



The Islamic University - Gaza

Faculty of Engineering Deanery of Graduate Studies



الجامعة الإسلامية – غزة

كلية الهندسة عمادة الدر إسات العليا

استبيان عن

ممارسة إدارة مواد البناء في مشاريع التشييد بواسطة مقاولي قطاع غزة

السادة المقاولون المحترمون / نشكر تعاونكم ومساهماتكم في تعبئة هذا الاستبيان الذي يعتبر جزء من البحث التكميلي لنيل درجة الماجستير في إدارة المشاريع. من خلال الاستبيان نرجو التعرف على الواقع المحلي في إدارة مواد البناء في مشاريع التشييد لدى مقاولي قطاع غزة ومن ثم محاولة حل مشاكل هذا النظام من خلال تطوير برنامج حاسوب متخصص بعمل إدارة مواد البناء لمشاريع التشييد بقطاع غزة. **يتكون الاستبيان من الأجزاء التالية:** الجزء الأول: السيرة الذاتية للشركة. الجزء الثاني: استخدام أدوات وتقنيات نظام إدارة مواد البناء في مشاريع التشييد. الجزء الثاني: استخدام أدوات وتقنيات نظام إدارة مواد البناء في مشاريع التشييد.

> الجزء الرابع: استخدام الحاسوب في نظام إدارة مواد البناء في مشاريع التشييد. الجزء الخامس: تطبيق نظام إدارة مواد البناء على مشاريع التشييد.

جميع البيانات والمعلومات الواردة في الاستبيان ستستخدم فقط لتقيم ودراسة نظام إدارة مواد البناء في المتبع في المشاريع، والمعلومات الشخصية والخاصة الواردة لن تنقل إلى أي جهات أخرى. سوف نطلعكم بإذن الله على نتائج البحث للاستفادة منها.

الباحث م. إياد عبد القادر الحداد	اشراف د.م کمالین شعث	
الجزء الأول/ السيرة الذاتية للشركة:		
سنة تأسيس الشركة :		 اسم الشركة:
🗆 فرعي 🛛 لايوجد	🗆 رئيسي	 2- مجال عمل الشركة: أعمال أبنية:

ا 🎗 للاستشارات

٠	أعمال مياه وصرف صح <u>ي:</u>	🗆 رئيسي	🗆 فرعي	🗆 لا يوجد
•	أعمال طرق:	🗆 رئيسي	🗆 فرعي	🗆 لا يوجد

3- درجة تصنيف الشركة حسب تصنيف اتحاد المقاولين لمجالات العمل التالية:

•	أعمال أبنية:	🗆 درجة أولى	🗆 درجة ثانية	🗆 درجة ثالثة
•	أعمال مياه وصرف صحي:	🛯 درجة أولى	🗆 درجة ثانية	🗆 درجة ثالثة
•	أعمال طرق:	🛯 درجة أولى	🗆 درجة ثانية	🗆 درجة ثالثة

4- المركز الإداري لمن يقوم بتعبئة الاستبيان: □ مدير/نائب مدير □ مدير مشروع أو مشاريع □ مهندس موقع □ غير ذلك

5- متوسط عدد الموظفين خلال الخمس سنوات الماضية: _____

6- متوسط عدد الموظفين حسب الشهادات العلمية خلال الخمس سنوات الماضية:

ماجستير فأعلى:	🗆 لا يوجد	🗆 يوجد	عدد
بكالوريوس:	🗆 لا يوجد	🗆 يوجد	عدد
معهد متوسط:	🗆 لا يوجد	🗆 يوجد	عدر
ثانوية عامة:	🗆 لا يوجد	🗆 يوجد	عدد
دون الثانوية العامة:	🗆 لا يوجد	🗆 يوجد	عدد

7- متوسط عدد الموظفين الفنيين حسب مجالات التخصص الهندسي خلال الخمس سنوات الماضية:

 عدد	🗆 يوجد	🗆 لا يوجد	مهندس مدني	•
 عدد	🗆 يوجد	🗆 لا يوجد	مهندس معماري	•
 عدد	🗆 يوجد	🗆 لا يوجد	مهندس کهرباء	•
 عدد	🗆 يوجد	🗆 لا يوجد	مهندس مكانيك	٠
 عدد	🗆 يوجد	🗆 لا يوجد	مهندس متخصص بغير ما ذكر	٠
 عدد	🗆 يوجد	🗆 لا يوجد	فني	٠

8- عدد المشاريع المنفذة خلال الخمس سنوات الماضية:

ا فاقل 10 20-11 30-21 40-31 50-41 أكثر من 50

9- إجمالي قيمة المشاريع المنفذة خلال الخمس سنوات الماضية (بالمليون دولار): 9- 1.5 -

- أكبر من 7.5 🗆
 - 10. الشخص المخول بعملية إدارة المواد في مشاريع التشبيد لديكم: □ مدير/ نائب مدير □ مدير مشروع أو مشاريع □ مهندس موقع □ قسم خاص بذلك □ غير ذلك____


الجزء الثانى/استخدام أدوات وتقنيات نظام إدارة مواد البناء فى مشاريع التشيد: 11- يرجى إبداء رأيكم في درجة ضرورة ومدى استخدامكم للتقنيات التالية في إدارة مواد البناء لمشاريع التشييد:

غدام	، الاستخ	مدی	ورة	ة الضر	درجا		
تستخدم نادر ا	تستخدم أحيانا	تستخدم عادة	غير ضروري	ضروري نوعا	ضروري	الطريقة	مسلسل
						إنشاء قواعد بيانات:	
						إنشاء قاعدة بيانات لتصنيف المواد.	1
						إنشاء قاعدة بيانات للموردين المحليين.	2
						إنشاء قاعدة بيانات للموردين الخارجيين	3
						إنشاء قاعدة بيانات للمواد وأسعار ها	4
						تحديث قواعد البيانات:	
						تحديث قواعد البيانات للموردين المحليين.	5
						تحديث قواعد البيانات للموردين الخارجيين	6
						تحديث قواعد البيانات لأسعار المواد عند حدوث تغير في الأسعار .	7
						استخدام الإنترنت للتعرف علي المواد الجديدة والأسعار الجديدة	8

12- يرجي إبداء رأيكم في درجة ضرورة ومدي استخدام تقنيات إدارة مواد البناء وكذلك طرق الاستخدام: (ملاحظة/:يرجي إرفاق أي نماذج تستخدمونها في عملية إدارة مواد البناء لمشاريع التشييد)

ŕ	بستخدا	ريقة الا	Ь	دام	الاستخ	مدی	ورة	ة الضر	درج		
بدون تسجيل	تسجيل بدون استخدام نموذج	تسجيل باستخدام نموذج	باستخدام نموذج محوسب	تستخدم نادرا	تستخدم أحيانا	تستخدم عادة	غير ضروري	ضروري نوعا ما	ضروري	الطريقة	مسلسل
										توفر قائمة بالمواد الموجودة في موقع المشروع	1
										تحتوي مثلا علي (اسم المادة-رقم المادة-	
										الوحدة-السعر).	
										التسجيل اليومي للمواد المستخدمة في المشروع .	2
										وجود بطاقات لكل مادة لاستخدام مخزن	3
										المشروع يشمل (الرصيد-المصروف-المتبقي)	
										وجود نموذج طلبية مواد لموقع المشروع	4
										يوضح مثلاً (رقم الطلبية-وصف المادة-الكمية	
										المطلوبة-السُعر).	
										تسجيل المواد المستلمة في المشروع يوضح	5
										مثلا (رقم الإرسالية-اسم المورد-وصف المادة-	
										كمية ألمادة).	

المتسارات

					تسجيل حالة المواد في مخزن الموقع يوضح	6
					مثلا (اسم المورد-رقم الطلبية-الكمية الداخلة-	
					المصروفة).	
					تسجيل الإشكاليات مثلا (الفاقد-السرقة أو	7
					الضياع منقص في التوريد عند الاستلام).	
					متابعة الأسعار في السوق وتسجيل تغيرات	8
					الأسعار ا	
					متابعة تذبذب حالات الإغلاقات الأمنية في	9
					السوق	

الجزء الثالث / تأثير الإغلاقات الأمنية على وفرة المواد وأسعارها في قطاع غزة:

13- يرجى إبداء رأيكم في مدي تأثير الإغلاقات على العوامل التالية:

يؤثر بدرجة قليلة جدا	يؤثر بدرجة قليلة	يؤثر بدرجة متوسطة	يؤثر بدرجة كبيرة	يؤثر بدرجة كبيرة جدا	العامل	مسلسل
					يؤثر الإغلاق الأمني للقطاع علي زيادة أسعار بعض المواد الأساسية مثل (أسمنت+حصمة+حديد)	1
					يؤثر الإغلاق الأمني للقطاع على زيادة في أسعار بعض المواد	2
					الأخرى مثل (مواد الدهان+مواد الكهرباء+مواد صحية).	
					يؤثر الإغلاق الأمني للقطاع علي زيادة أسعار مقاولي الباطن.	3
					يؤثر الإغلاق الأمني للقطاع علي تأخير تنفيذ المشروع عن الوقت	4
					المحدد له	
					يؤثر الإغلاق الأمني للقطاع علي زيادة في تكلفة المشروع بشكل	5
					عام.	

الجزء الرابع / استخدام الحاسوب في نظام إدارة مواد البناء في مشاريع التشييد

14 ـ بين مدي استخدام شركتكم لبر امج حاسوب في دعم إدارة مواد البناء في مشاريع التشييد:

تستخدم برامج متخصصة 🛛 Excel تستخدم برامج جداول إلكترونية مثل الأكسل 🗆 لا تستخدم 🗆

15- بين مدي كفاءة استخدام شركتكم لبر امج الحاسوب التالية:

🗆 ضعيف جدا	🗆 ضعيف	🗆 مقبول	<u>جن</u> د	🗆 جيد جدا	Word
🗆 ضعيف جدا	🗆 ضعيف	🗆 مقبول	🗆 جيد	حید جدا	Excel
🗆 ضعيف جدا	🗆 ضعيف	🗆 مقبول	🗆 جيد	حید جدا	Access
🗆 ضعيف جدا	🗆 ضعيف	🗆 مقبول	🗆 جيد	🗆 جيد جدا	Ms-Project



16- بين مدى تأثير المعوقات التالية علي استخدام برامج الحاسوب المتخصصة في إدارة مواد البناء في مشاريع التشييد:

يؤثر بدرجة قليلة جدا	يؤثر بدرجة قليلة	يوثر بدرجة متوسطة	يؤثر بدرجة كبيرة	يؤش بدرجة كبيرة جدا	المعوقات	مىيل سىل
					عدم إدراك أهمية نظام إدارة مواد البناء للمقاول.	1
					عدم فهم نظام إدارة مواد البناء.	2
					عدم القدرة على تطبيق نظام إدارة مواد البناء.	3
					عدم توفر برامج حاسوب سهلة و متخصصة في إدارة مواد البناء.	4
					عدم وجود أشخاص مؤهلين لاستخدام برامج الحاسوب المتخصصة.	5
					إمكانية و سهولة الإدارة يدويا عن استخدام الحاسوب.	6
					التكلفة المالية للبرنامج المرتفعة نسبيا.	7
					الاعتقاد بأن تطبيق النظام يضيع وقت المشرفين على تنفيذ المشروع.	8

17- سأكون مستعدا لاقتناء واستخدام برنامج متخصص في إدارة مواد البناء في حال كونه سهل الاستخدام وقليل التكاليف

🗆 موافق جدا 🗆 موافق 👘 لا أدري 📄 غير موافق 👘 عير موافق جدا



الجزء الخامس / تطبيق نظام إدارة مواد البناء على مشاريع التشييد 5-1 أهمية تطبيق نظام إدارة المواد

18- بين مدى تأثير الفوائد التالية عند تطبيق نظام إدارة مواد البناء:

يؤثر بدرجة قليلة جدا	يۇثر بدرجة قليلة	يؤثر بدرجة متوسطة	يؤثر بدرجة كبيرة	يؤثر بدرجة كبيرة جدا	الفائدة	مسلسل
					تقليل تكاليف مواد البناء اللازمة للمشروع.	1
					تحسين طرق مناولة المواد Better handling of (material).	2
					التقليل من تكرار طلبات المواد.	3
					توفر المواد في موقع العمل في الوقت المناسب والكمية المناسبة.	4
					تحسين في إنتاجية العمال.	5
					الالتزام بالجدول الزمني للمشروع.	6
					التحكم في الجودة.	7
					تحسين متابعة المواد والسيطرة عليها.	8
					العلاقات الجيدة مع الموردين.	9
					التقليل من الفاقد.	10
					التقليل من ساحات تخزين المواد في موقع العمل.	11
					الحصول علي أسعار أفضل للمواد.	12
					التحسين في توفر السيولة النقدية(cash flow).	13

19- بين مدى أهمية تطبيق نظام إدارة المواد على التقليل من المشاكل التالية:

غير مهمة بتاتا	غير مھم	لا يوجد فرق	مهمة	مهمة جدا	المشكلة	مسلسل
					عدم توفر المواد كليا.	1
					عدم توفر المواد بالكميات المطلوبة.	2
					وصول المواد بشكل متأخر.	3
					تأخر الموافقة علي العينات المقدمة.	4
					استلام مواد غير صحيحة.	5
					استلام المواد و بمقاسات مختلفة عن المطلوب.	6
					استلام المواد بالكميات الغير مطلوبة.	7
					تكدس المواد في المخازن.	8
					سرقة المواد.	9
					تلف المواد أثناء الاستلام.	10

<u>5-2 طلب المواد Ordering materials</u>
2-5 علم العوامل التالية من حيث تأثير ها علي زيادة نسبة الكمية المطلوبة من المواد بشكل مبكر:

تأثير سلبي واضح	تأثير سلبي	لا يۇش	تأثير إيجابي	تأثير إيجابي واضح	العامل	مسلسل
					البرمجة الدقيقة للمشروع.	1
					طول فترة السماح للدفع من قبل المورد.	2
					إمكانية فقدان المواد أو تلفها	3
					توفر القدرة المالية لدى المقاول	4
					توفر ساحات تخزين كافية لدي المقاول	5



21-المسئول عن طلب المواد في الشركة هو.

□ مدير الشركة □ مهندس المشروع □ قسم المشتريات □ قسم خاص بذلك
 □ غبر ذلك

22-بين مدى أهمية استخدام التقنيات التالية في طلب المواد من الموردين.

غير مهم جدا	غير مهم	لا أدري	مهر	مهم جذا	الطريقة	مسلسل
					الإنترنت <u>.</u>	1
					البريد الإليكترونيe-mail.	2
					الفاكس.	3
					الاتصال بالهاتف	4
					الذهاب شخصيا .	5

5-3 تخزين المواد(Storage) :

23- ما هي المساحة المتوفرة لديكم من المخازن المغطاة:

□ 200 م2 فأقل □ 100-501 م2 □ 1000-501 م2 □ 2000-1001 م2

□ أكبر من 2000 م2

- 24- ما هي المساحة المتوفرة لديكم من المخازن غير المغطاة:
- □ 1000م2 فأفل □ 1000-6001 م2 □ 1000-0001 م2 □ 1000-6001 م2
 - □ أكبر من 10000 م2
 - 25- يتم الاحتفاظ بالمواد المختلفة في مخازن الشركة على النحو التالي:

مطلقا	ئادرا	أحيانا	غالبا	دائما	المادة	مسلسل
					حديد التسليح	1
					خشب الطوبار .	2
					إسمنت.	3
					رمل.	4
					حصمة	5
					حجارة البلوك	6
					بلاط.	7
					دها <u>ن.</u>	8
					حجر صخر للواجهات.	9
					مواسير (صرف صحي+مياه).	10
					بلاط إسمنتي.	11
					حجر جبهة ِ	12
					مادة طبقة الأساس(Base course).	13
					مستلزمات كهربائية.	14
					مستلزمات صحية.	15



26- بين مدى تأثير العوامل التالية علي زيادة الفاقد في المخازن:

يؤثر بدرجة قليلة جدا	يۇئر بدرجة قليلة	يۇئر بدرجة متوسطة	يۇئر بدرجة كبيرة	يؤثر بدرجة كبيرة جدا	العامل	مىيلىيل
					تخزين خاطئ للمواد (عدم ترتيبها حسب النوع).	1
					تكدس المواد وعدم كفاية مخزن الموقع	2
					التعليمات غير كافية عن كيفية التخزين.	3
					استخدام أماكن غير مناسبة للتخزين (رطبة- معتمة).	4
					عدم وجود أشخاص مؤ هلين في إدارة المخازن .	5

4-5 كيفية اعتماد الموردين How to qualify suppliers

27- حدد درجة أهمية العوامل التالية في تقييم واعتماد الموردين:

غير مهم جذا	غير مهم	لا أدري	e fe	مهم جذا	العامل	مسلسل
					منافسة الأسعار المقدمة منهم.	1
					قصر الفترة الزمنية لتسليم المواد .	2
					جودة المواد المقدمة	3
					الامتيازات وزيادة فترة السماح للدفع	4
					سمعة المورد في السوق.	5
					توفر شبكة فروع جيدة للموردين في المناطق المختلفة.	6

28- تتم مواجهة المشاكل التالية أثناء تقيم الموردين على النحو التالي:

مطلقا	ئادرا	أحيانا	غالبا	دائما	المادة	مسلسل
					وجود مورد وحيد للمادة المطلوبة.	1
					عدم توفر المعلومات المطلوبة عن الموردين.	2
					عدم صدق المعلومات المتوفرة.	3
					عدم وجود عقوبات تفرض ضد الموردين في حالة الخلل بالاتفاق.	4
					عدم القدرة علي إلزام المورد بتنفيذ العقد	5
					الظروف السياسية السائدة	6

المتسارات

5-5 الفاقد في مواد البناء

29- حدد مدى أهمية معرفة الفاقد في المواد للعوامل التالية:

غير موافق بشدة	غير موافق	محايد	موافق	مو افق بشدة	الأهمية	مسلسل
					يـساعد فـي معرفـة الكميـات المطلوبـة بالـضبط لتنفيـذ المشروع.	1
					يزيد في فرصة الحصول على تمويل للمشروع.	2
					يساعد على إعداد جدول كميات دقيق <u>.</u>	3
					يساعد المقاولين على تسعير العطاءات بشكل دقيق.	4
					تحسين فرصة الفوز بالعطاء.	5
					معرفة المتطلبات الحقيقية للمشروع المراد تنفيذه والمتمثلة في المواد والزمن والتكاليف	6
					يساهم في نجاح المشروع وتحقيق الأرباح.	7
					يـساهم فـي تحـضير جـدول زمنـي دقيـق يـشمل مـوارد المشروع	8
					يساعد المقاول في إعداد جداول زمنية دقيقة لتوريد المواد	9

6-5 العناصر التي تسبب زيادة الفاقد في الموقع

30- بين مدي تأثير العناصر التالية علي زيادة الفاقد:

الفاقد في مواد البناء						
يؤثر بدرجة قليلة	يؤثر بدرجة قليلة	يوئٹر بدرجة تقصيطة	يوأثل بدرجة كبيرة	يۇٹر بدرجة كبيرة	العناصر التي تسبب زيادة الفاقد في الموقع	مسلسل
				J	(أ) في الموقع:	
					تلف مواد البناء في الموقع.	1
	-				القص الغير مناسب لمواد البناء.	2
					وجود مواد غير ضرورية في الموقع.	3
					إنتاج كميات أكثر مما هو مطلوب أو قبل الحاجة إليها.	4
					المواد بها عيوب من التصنيع.	5
					السرقة أو التخريب.	6
					رداءة مواد البناء.	7
					نقص المواد المطلوبة بسبب الإغلاق والطلب المتزايد عليها.	8
					نقص الرقابة على المواد في الموقع	9
					عدم تخزين المواد بشكل صحيح إ	10
					زيادة في حجم العناصر الإنشائية أثناء التنفيذ	11
					استخدام كميات أو أطوال أو أقطار زائدة من المواد دون الحاجة	12



	البناء	لي مواد	الفاقد ف			
يؤثر بدرجة قليلة	يؤثر بدرجة قليلة	يۇڭر بدرجة متاسطة	يؤثل بدرجة كبيرة	يۇئر بدرجة كبيرة 214	العناصر التي تسبب زيادة الفاقد في الموقع	مسلسل
					لذلك ِ	
					(ب) مناولة المواد:	
					سوء مناولة المواد داخل الموقع.	13
					النقل المتكرر لمواد البناء داخل الموقع	14
					التعليمات غير كافية عن كيفية مناولة المواد داخل الموقع	15
	1				(ج) النقل:	T
					نقل المواد بطريقة غير صحيحة مما يؤدي إلى تلفها.	16
					عدم تخزين المواد في مكان قريب من موقع الإنشاء .	17
				ŋ	(د) إدارة الموقع:	
					الافتقار إلى خطة لإدارة المواد والوقت.	18
					الافتقار إلى نظام لضبط الجودة	19
					سوء تخطيط موقع العمل (إهمال الموقع- صعوبة التنقل).	20
					عدم كفاءة طاقم المقاول في التنفيذ.	21
					تزويد فريق العمل في المتبروع بمعلومات ضئيلة.	22
					عدم كفاءة المقاول في الرقابة على سير عمل المتنزوع.	23
					نقص المهنيين والمشرفين في طاقم المقاول.	24
					البطء في اتخاذ القرار من قبل المقاول .	25
					التأخير عن بدء العمل بالمتسروع.	26
				1	(و) الإشراف على الموقع:	07
					ضعف المراقبة و الإسراف وناخر تعليمات المسرف.	27
					طاقم الاستثناري الذي ثم تعيينه في المسروع غير مؤهل	28
					تأخر مهندس الاستشاري في الرد علي نساؤ لات المفاول .	29
					تأخر مهندس الاستشاري في الموافقة على تنفيد أو تقديش أو فحص . أعمال المقاول.	30
					سوء الاتصال والتنسيق بين المالك و مهندس الاستشاري و المقاول	31
					تأخر المالك في تسليم المشروع للمقاول.	32
					إيقاف العمل من طرف المالك	33
					الأوامر التغييرية.	34
					عدم تعاون المالك مع المقاول والتأخر في الدفعات وحل مطالباته.	35

المنارات المستشارات

Annex 2

The Questionnaire (English Version)



Section one: Company Profile:

1- Year of company establishment _____

2- Company work field:

• Water and sewerage works: □ Main □ Secondary □ Unspect	ecialized
• Water and sewerage works. \Box Man \Box Secondary \Box Onsp	ecialized
Roads works: □ Main □ Secondary □ Unspective	ecialized

3- Company classification according the contracting union for the following fields:

- Water and sewerage works:
 □ First class
 □ Second class
 □ Third class

4- Managerial position of the respondent:
 □ Director/Deputy director
 □ Project manager
 □ Site engineer
 □ Other way

5- Average of employees' number during the last five years_____

6- Average of employees' number (in the year) during the last five years according to their scientific certificates:

•	MSc. And higher	\Box Existing	Number
•	BSc.:	Existing	Number
•	Community Collage:	Existing	Number
•	Baccalaureate-Tawjihi	\Box Existing	Number
•	Under baccalaureate:	\Box Existing	Number

7- Average of technical employees' number during the last five years according to the engineering specialization:

Civil engineer		Architect engineer_	Electr	rical
engineer	_Mechanical	engineer	other specialist engin	neer
Technical				

8- Number of executed projects during the last five years:

• \Box 10 and below \Box 11-20 \Box 21-30 \Box 31-40 \Box 41-50 \Box More than 50

9- Total amount of executed projects during the last five years (in million dollars):

- \Box 1.5 and lesser \Box 1.6-3 \Box 3.1-4.5 \Box 4.6-60 \Box 6.1-7.5 \Box More than 7.5
- 10- The person in charge of managing construction materials in construction projects:

Company Director	Project Manager	□ Site Enginee
\Box specific section	□ Others. Determine	



Section tow: Application of construction materials management tools and techniques in construction projects:

11. To which extent you evaluate the necessary and degree of usage of the next techniques in construction materials management:

		N	lecessi degree	ty e	Usage degree			
No.	Techniques	Necessary	Somehow necessary	Unnecessar y	Usually	Occasionall y	Rarely	
	Establishing Data base							
1	Establishing categorized materials							
	database							
2	Creating local suppliers database.							
3	Creating international suppliers							
	database.							
4	Creating materials price database.							
	Updating database							
1	Updating the database of local							
	suppliers.							
2	Updating the database of international							
	suppliers.							
3	Updating the database for materials							
	price when change occurs.							
4	Using internet for knowing the new							
	materials and their prices.							

12. To which extent you evaluate the necessary, degree of usage, and method of use

of the next techniques in construction materials manageme	nt:
---	-----

		Nece	ssity d	egree	Usa	age deg	gree		Method	l of use	e
No	Techniques	Necessary	Somehow necessary	Unnecessary	Usually	Occasionally	Rarely	By using a computerized	Recording by using a form	Recording without form	Without recording
1	Providing a list of materials in project that includes for example (material name, material number unit price)										
2	Daily recording of using materials in the project.										
3	Providing material cards at site store that contain for example (input-output- balance).										
4	Providing materials purchase order including for example (order number-material description-required quantity -price).										
5	Recording the received materials on site, the record										



		Necessity degree			Usage degree			Method of use				
No	Techniques	Necessary	Somehow necessary	Unnecessary	Usually	Occasionally	Rarely	By using a computerized	Recording by using a form	Recording without form	Without recording	
	shows for example (delivery											
	number-supplier name-											
	material description-											
	quantity).											
6	Reporting the situation of											
	the report above (supplier name											
	order number-quantity input-											
	quantity output-balance).											
7	Reporting the problems for examples (wastage and											
	breakage-thief and loss-shortage											
	in delivery).											
8	Following up the prices in the											
	market and recording the											
L	variation of prices.											
9	Following up the closure											
	variations in the `market											
Cast	·		··· •		•				1 - 1- 2124	<u> </u>	1	

<u>Section three: The effect of closure in Gaza strip on materials availability and cost:</u>

13. To which extent you evaluate the effect of closure on the following factors:

No	Factor	Very high degree affect	High degree affect	Mid degree affect	Little degree affect	Very little degree affect
1	The closure causes a noticeable increase in the main material prices specially (cement, aggregate, and steel).					
2	The closure causes an increase of other material prices (painting, electrical, sewage material).					
3	The closure increases the subcontractors' rates.					
4	The closure causes a delay of a project completion.					
5	The closure causes an increase of total project cost.					

Section four: Computer applications in material management systems in construction projects:

14. using of your company of software for supporting the construction materials management.

 \Box Not use. \Box Use spreadsheet-based software (Ms Excel) \Box Use special software



	e	-	•	Ũ
Word	\square V. Good	\square Good	\Box Acceptable \Box Low	\Box V. Low
Excel	\square V. Good	\square Good	\Box Acceptable \Box Low	\Box V. Low
Access	\square V. Good	\square Good	\Box Acceptable \Box Low	□ V. Low
Ms-project	\Box V. Good	\square Good	\Box Acceptable \Box Low	□ V. Low

15. Rank the following software according your extent of handling them in the job:

16. To which extent you evaluate the next obstacles effect on the usage of construction materials management:

No.	Obstacle	Very high degree	High degree effect	Mid degree effect	Little degree effect	Very little degree
1	Non- realization of importance of construction materials management system by the contractor.					
2	Absence of understanding of construction materials management system.					
3	Inability implementation of the system.					
4	Shortage of user friendly of construction materials management system.					
5	Shortage of qualified persons in using a construction materials management system.					
6	Simplicity of manual managing a construction material.					
7	The high cost of a construction materials management system.					
8	The thinking that implementing the system wastes the time of project supervisors.					

17. I am ready to get and use user friendly and inexpensive cost of construction material management system.

 \Box strongly agree \Box Agree \Box Neutral \Box Disagree \Box strongly disagree.



Section five: Implementation of construction materials management systems on construction projects

1-5 Important of implementation a construction materials management system

18. To which extent you evaluate the effect of the following benefits when implement a construction materials management system:

No	Benefit	Very high degree effect	High degree effect	Mid degree effect	Little degree effect	Very little degree effect
1	Reducing the costs of project materials.					
2	Better handling of materials.					
3	Reducing duplication of materials orders.					
4	Materials are timely available on site with the right quantity.					
5	Improving labour productivity.					
6	Complying to time schedule.					
7	Complying to enhancement of quality control.					
8	Improving follow up and monitoring of construction materials.					
9	Better relationships with suppliers.					
10	Waste reduction.					
11	Reducing the space for materials on site.					
12	Obtaining better price for the construction materials.					
13	Improving cash flow.					

19. To which extent you evaluate the importance of implementing materials

management to reduce the following problems:

No	Problem	Very important	important	No defense	Not important	Never Important
1	Materials are not available.					
2	Materials are not available with required quantity.					
3	Late delivery to the site.					
4	Slow response from the consultant engineer about submittals.					
5	Deliver wrong materials.					
6	Deliver materials with wrong dimensions.					
7	Deliver materials with wrong quantities.					
8	Increase materials quantity in storages.					
9	Burglary, theft and vandalism					
10	Destroyed materials when deliver.					



2-5 Ordering materials

20-To which extent you evaluate effect of the following factors on increasing quantity

of materials early:

No	Factor	Clear positiv e effect	Positiv e effect	Don't affect	Negati ve affect	Clear negativ e affect
1	Good project schedule.					
C	Long of supplier period for contractor					
2	payment.					
3	Damage or waste of materials.					
4	Contractors cash capability.					
5	Sufficient storages available.					

21-The person who responsible about ordering materials:

 \Box Company manager \Box site engineer \Box procurement department

 \Box specific section \Box others. Determine

22. Determine the importance degree of the following techniques to order materials from suppliers:

No	Problem	Very important	important	I don't Know	Not important	Never Important
1	Internet.					
2	E-mail.					
3	Fax.					
4	Telephone.					
5	Personal meeting.					

3-5 Storage materials:

23- Available area of covered storages: □ 200m2and below □ 201-500m2 □ 501-1000m2 □ 1,001-2,000m2

 \Box more than 2,000m2

24- Available area of no covered storages:

□ 1,000m2and below □ 1,001-3,000m2 □ 3,001-6000m2 □ 6,001-10000m2 □ more than 10,000m2



25- The company stores the next materials as follow:

No	Material	Always	Often	Sometimes	Seldom	Never
1	Reinforcement steel.					
2	Shuttering timber.					
3	Cement.					
4	Sand.					
5	Aggregate.					
6	Blocks.					
7	Tiles.					
8	Painting.					
9	Stone.					
10	Pipes.					
11	Terrazzo tiles.					
12	Curb stone.					
13	Base course.					
14	Electrical materials.					
15	Sanitary materials.					

NO.	Factor	Very high degree affect	High degree affect	Mid degree affect	Little degree affect	Very little degree affect
1	In proper materials storage.					
2	Inadequate stacking and insufficient storage on					
	site					
3	Insufficient instructions about storage and stacking					
4	Using unsuitable places for store materials					
5	Inappropriate storage leading to damage or					
	deterioration					

26. To which extent these factors affect increasing waste in storage:



<u>4-5.How to qualify suppliers:</u>

27. To which extent you evaluate the important degree of the following factors in qualifying suppliers:

NO.	Factor	very important	important	Don't know	Not important	Little important
1	Price competitiveness.					
2	Short period to deliver materials to the contractor					
3	Quality of materials					
4	Increase payment period					
5	The capability of supplier in the market					
6	Having many branches in different geographical areas.					

28. To which extent you are facing the following problems when qualifying suppliers:

No.	problem	Always	Often	Someti mes	Seldom	Never
1	Sole supplier.					
2	Lack of required information about suppliers.					
3	Wrong information about suppliers.					
4	Lack of penalty measures against defaulted suppliers.					
5	Inability to enforce contract condition on suppliers.					
6	Prevailing political conditions.					

5-5 materials waste:

29- To which extent you evaluate the important degree on knowing materials waste for the following factors:

No.	Importance	Strongly agree	Agree	Neutral	Disagree	Strongly disagree
1	Help to determine the exact required quantities.					
2	Increase the chance for obtaining the project finance.					
3	Help for preparing accurate bill of quantities.					
4	Help contractors to price tenders more accurately.					
5	Help the contractor to have a better chance to win the tender.					
6	Knowing the real requirements for the project (materials- time-cost)					
7	Help to finish the project successfully and have profits.					
8	Help in preparing a good schedule program including project resources.					

5-6. Factors increase waste on the construction site:

30. Explain the degree of effect for the following factors which increase waste on the construction site:

(1) Very high degree affect (2)High degree affect (3)Mid degree affect (4)Little degree affect (5)Very little degree affect



No.	Factors causes crease waste on site	E	2	ts' D	egre	e 5
	(A): On site:	1	2	5	-	5
1	Materials damage on site.					
2	Improper cutting of materials.					
3	Existence of unnecessary materials on site.					
4	Overproduction/Production of a quantity greater than required					
	or earlier than necessary.					
5	Manufacturing defects.					
6	Burglary, theft and vandalism.					
7	Poor quality of materials.					
8	Lack of materials (due to closure).					
9	Lack of on site materials control.					
10	Poor storage of materials.					
11	Over sizing structural elements during execution.					
12	Using excessive quantities of materials.					
	(B): Handling:					
13	Improper handling of materials on site.					
14	Duplication of transporting material on site.					
15	Insufficient instructions about handling materials on site.					
	(C): Transportation:					
16	Improper materials.					
17	storing materials in far away stores					
	(D): Site management and practices:					
18	Lack of material and time waste management plan.					
19	Lack of a quality management system aimed at waste					
	minimization.					
20	Poor site layout.					
21	Poor qualification of the contractor's technical staff assigned					
	to the project.					
22	Providing project team with insufficient information.					
23	Ineffective control of the project progress by the contractor.					
24	Shortage of technical professionals in the contractor's					
	organization.					
25	contractors slowness in taking decisions.					
26	Delay in project commencement.					
	(E): Site supervisor:					1
27	Poor control of supervision and delay in giving instructions.					
28	Poor qualification of consultant engineer's staff assigned to					
	the project.					
29	Slow response from the consultant team to contractor					
	inquiries.					
30	Delay in performing inspection and testing by the consultant					
	team.					
31	Poor coordination and communication among the consultant,					
	the owner and the contractor.					
32	Owner's delay in handing over the site to the contractor.					
33	Suspension of work by the owner.					
34	Change orders.					
35	Poor cooperation of the owner towards settling contractors					
	payments and claims					



Annex 3

Review of Construction Materials Management Software Packages

(CMMS)

Figure A3.1: A sample of CMMS Files.

Figure A3.2: A sample of "Main Menu" sheet.

Figure A3.3: A sample of "Company and project information" sheet

Figure A3.4: A sample of "Materials Pool" sheet.

Figure A3.5: A sample of "Activity Pool" sheet.

Figure A3.6: A sample of "Project Activities" sheet.

Figure A3.7 A sample of "Activities Materials Quantities" sheet.

Figure A3.8: A sample of "Materials Needed" sheet.

Figure A3.9: A sample of "Materials to Order between Two Dates" sheet.

Figure A3.10: A sample of "Materials Spaces and Cash Requirements" sheet.

Figure A3.11: A sample of "Earned value S-curve calculation" sheet.

Figure A3.12 A sample of "Earned value S-curve".

Figure A3.13: A sample of "Materials to order and deliver on site" sheet.

Figure A3.14: A sample of "Materials Must be purchased at" sheet.

Figure A3.15: A sample of "Materials Spaces Assignment" sheet.

Figure A3.16 A sample of "Materials Prices Assignment" sheet.

Figure A3.17: A sample of "Materials Cumulative Spaces & Prices at P.O.D." sheet.

Figure A3.18: A sample of" Purchase order" sheet.

Figure A3.19: A sample of "Materials Must be on the site at" sheet.

Figure A3.20: A sample of "Materials Card" sheet.

Figure A3.21: A sample of "Waste control" sheet.

Figure A3.22: A sample of "Help and Method of Use" sheet.





Figure A3.1: a sample of CMMS files.





Figure A3.2: a sample of "Main Menu "sheet.



Figure A3.3 a sample of "Company and project information" sheet



					Microsoft Excel - F
🚽 اكتب سؤالاً للتعليمات			تعليمات	أدوات بيانات إطار	تحرير عرض إدراج تنسيق
🎽 🖬 🖪 🔒 🖪 🖪 🖏 🖤 🛍	🔏 🖻 🔁 • 🟈 🤊 • 🍽 • 🧕	$\Sigma \rightarrow \frac{1}{2} \downarrow \frac{1}{4} \downarrow	• 💿 🚽		
• 1• • B I	U 토 플 클 쿄 ① M • 9	% , *.0 .00 €≢ €≢ []]	• 🙆 • 🛕 • 💂		
A1 ▼ fx					
	UE	FUHI	J K L M	N U P	U R S I
	BACK TO M. MEN	U .			
Phone :	ENTER				
Fax :					
- MATERIAL	POOL				
	15 (50)				
M. Code Material Material	Name Unit Unit Price	Unit Space			
↓ ▶ N \ Inter Face / main menu /	com@pro.inf) Mat.Pool / Act.Pool	/ Pro.Act / Qua.of Mat Of Ac	/ Mat To Order / 2Date	Mat.To Order 🔏 Mat.	Ass / Ear V. S-Curve Cal / Ea
1		,	,	A contraction	
NUM					

Figure A3.4: a sample of "Materials Pool" sheet.



									Mi	crosoft Excel	- Part1 💶 🗗
🚽 اكتب سؤالاً للتعليمات						لجار تعليمات	ات او	ىيق أد وات بيان	ج تنم	<u>حرير عرض إدر</u>	_ 🕙 مِلف ت
🞽 🖬 🕒 🗿 🖾 🕼 🎌 🛍 🖁 🖧 • 🟈 🔊 •	(H + 8	$\sum_{i} \Sigma * \frac{A}{Z} \downarrow \frac{Z}{A} \downarrow$		🚯 %lo 🗸 🔞	÷						
	भग 🛨 🔤	🖞 % , 號 .	.00	: 🖅 🖽 + 🖄	• <u>A</u>	• 🕫					
A1 7× 7× B C	D	E	F	G	н	1	J	К	L	М	N O
_											
	BAC	SK TO W. WENG									
Final Fax :	9	ENTER	1		BAC	K TO IWAT.POOL					
ACTIVITY POOL											
				Mai	terial	s Per One Hi	uit Oi	f Activity			
Act.Code Activity Description	Unit	Mat. 1 Name	Mat1 Quan. / Unit	Mat. 2 Name	Mat2 Quan. /Unit	Mat. 3 Name	Mat3 Quan. /Unit	Mat. 4 Name	Mat4 Quan. /Unit	Mat. 5 Name	Mat5 Quan. /Unit
		-									
82											
1 1 1 1 1 1		-									
▶ ♥ \ Inter Face / main menu / com@pro.inf / Mat.Pool)	Act.Poc	ol / Pro.Act / Q	ua.of N	lat Of Ac 🖌 Mat	: To Or	, der / 2Date Mat	.To O	rder / Mat.Ass	/ Ear \	/.S-Curve Cal 🖌	Ear 🔇 🚺
NUM			_		_						

Figure A3.5: a sample "Activity Pool" sheet.



26														Micro	soft E	xcel -	Part1	_ &
	غليمات	🚽 اكتب سؤالاً للت							تعليمات	إطار	بيانات	أ <u>د</u> وات	تنسيق	إدراج	عرض	تحرير	ا ملف	🖹 - 8
	📬 🔒	🖪 🗿 🖪 💁 🎌 📖 👗 🖻 📇 • 🕩 🔊 •	(2 +	🔍 Σ 🔹	21 31 10	1 -13	%10 - 🤇											
Aria	(• 1•• B <i>I</i> U ≡ ≡ ≡ ⊒ ⊡ 1	Þ¶ -		00. 0.+		1 🖽 🗕 👌	• A	-									
-	A1	✓ f _x						_										
A	С	D	E	F	G	Н	1	J	К	L	M	N	0	P	Q	R	S	Т
2			RACK	TO BE REFUIL	2													
4			Сиси				-											
5	Phone : Fax :			ENTER				BACK	IO ACT.POOL									
	or production of the					PRO	JECT A	CTIV	TTIES									
1	1				1					2								
0		ENTER PROJECT SATRY DATE																
1										n								
2	Act. Code	Activity Description	Unit	Duration	Date	Finish Date	Quantity	Unit	Total									
3							<u>.</u>											
5																		
6									<u> </u>									
5																		
9	8																	
1																		
2 3																		
4																		
5									8									
2	8			8														
			-															
2																		
+					-													
5	3							<u> </u>										
,																		
3	-				-													
0																		
1					-													
3		Inter Eace / main menu / com@nro.inf / Mat Deal	Act		ct / Ours	of Mot O	FAC / M	t To C	rder / 2D-+	D Mot 7	[n Order	/ 64-+	Acc /	Early C	Curve	cal / r	ar 💌	
	• • • (1	NUM	(ACL	FOOL Y PLOTA	ιι _Α Qua	or Mat O	IAC A Ma		iller i ZDat	e Mat, I	io orde	X Mat	.ASS <u>X</u> 1	car v. S	curve	cal X t	ar 💽	
	9-12 🔞	3.0						منام	الله الد د	Micro	soft Even	- Part1		ma	ster 13	emntu 🖻		المر

Figure A3.6: a sample of "Project Activities" sheet.



										Microsoft	Excel - F	art 1 📃 🕻
اكتب سؤالأ للتعليمات						تعليمات	إطار	أدوات بيانات	تنسيق	عرض إدراج	تحرير	_ 🕙 مِلف
🞽 🖬 🖪 🔒 🗃	i 💁 🖑 🎇 i 🕺 🖻 🔀 • 🟈 i 🤊 • 🖗 ·	• 🔒 Σ	+ A↓ Z↓ A	1 - 01%	0							
l 💽	1·• B I U ≣ ≣ ≣ ⊠ I M	• 🛒 %	••.0 .00 •.€ 00. •	🗱 🗱 🛄 🗸	<u>ی</u> -	<u>A</u>						
A1 🔻	fx											
В	C	D	E	F	G	Н		J	K	L	M	N
		846										
	25-Photo			1								
Phone : Fax :			ENTER									
		_										
ACT	TIVITIES MATERIAL QUANTITIES											
17. 19 <u></u> 17.				81								
Activity	Activity Description	Unit	total		a 0	A	ctiviti	es Material Pi	opor	tions		
Code			Quantity	Mat1	Qua.	Mat2	Qua.	Mat3	Qua.	Mat4	Qua.	Mat5
									g		8 8	
				t.								
				<u>.</u>							8 8	
		-			-							
		8.8		<u>.</u>					8		8 3	
		-						·				
									-	12		
						- 1 1			Ι,			-1
► ► \ Inter Face /	main menu (com@pro.inf (Mat.Pool (Act	.Pool / Pro	o.Act \Qua.o	of Mat Of Ac /	Mat T	o Order 🖌 2Date	e Mat.T	o Order / Mat.A	ss (Ear V. S-Curve	e Cal / Ea	r < 🔟
NUM							0.010	NAR WARANTAN			and the second	-
9:12 🔞 😵 🧶						🛚 🖌 - الرسام	Micros	oft Excel - Part1 🔀		master 1	3 empty 🚞	ابدا

Figure A3.7: a sample of "Activities Materials Quantities" sheet.



2												dicros	oft Ex	icel - F	art1	FX
:	الأللتعليمات	👻 اکتب سز					_	تعليمات	ايات إطار	<u>د</u> وات ہی	تنسيق أ	إدراج	عرض	تحرير	💌 ملف]_ 8 ×
	3 🖬 🔒		$\Sigma + \frac{A}{Z}$	Z 🛄	-19	%10 - 🕜	-									
Arial			% *	€.0 .00 3		🖽 • 🖄	• 🔺 - 🕫									
A	M4 B	f* C	D	E	F	G	н	1	J	К	L M	0	P	Q	S	т
1 2					2						-					
3			BACK TO	M. MENU	÷						-	1				
5	Phone :		EN	TER		•						•				
7	rax :															
9		MATERIALS NEEDED														
10				1000000000	,		1				-+4				<u> </u>	1-12 I
12	Act. Code	Activity Description	Unit	Total Dur.	Total Quan.	Satart Date	Finish Date	Days No.	Act. Dates	Mat.1 M Name	otal Quan.to	Mat.2 Name	total	Quan.to order	Mat.3 Name	iat3. Qu iotal c
13		<u>,</u>									uan		Quan			
15 16																
17 18																
19 20																
21																
23																E
25																
26																
28																
30 31																
32 33																
34 35																-
36											-					
38												1				
40																
41 42																
43 44																
45 46																
47		Eace / main menu / com@nro inf / Mat Dool / Act Dool /	Dro Act	/ Out of	Mat Of	Ac \ Mat	To Order	/ 2Date /	Mat To Or	for / M-	at Acc / F			al / 5-		
	NUM	Pace & mainmenu & comeptorini & MacPool & ActiPool &	FIU.ACL	V Qualiti	Mat OI	AC Y Mai	. To order	V Share I	nati tu Oft	иен Д Ма	ICASS & Ed	n v. 5-0	curve c	a K Ea		::: [] 🔊 جاهز
e +9	18 😻 😼 🧐						۰ الرسام	19	Microsoft Ex	cel - Part1	3	ma	ster 13 e	mpty 🚞		ابدًا

Figure A3.8: a sample of "Materials Needed" sheet.



••••••••••••••••••••••••••••••••••••	26			Microsoft Excel - Part1 🗔 🗃
Al	بمات 🗧	🚽 اكتب سؤالاً للتعلي		🗗 _ 🕙 مِلف تحرير عرض لِدراج تنسيق أدوات بيائِات لِطار تِعليمات
A1 A A1 A C A B10-10 M MEDU Proce: B10-10 M MEDU Proce: B10-10 M MEDU MATERIAL TO ORDER BETWEEN TWO DATES FR/MADATE Model Material Description Model Motorial To OATE Model Motorial To OATE Model Motorial To OATE Model Motorial To OATE To OATE To OATE Model Motorial To OATE Motorial To OATE Motorial			alo 🗸 👩 🔡 🔡 Arial	• 1•• B / U 三 三 三 冠 引 H • 明 % • % 梁 建 建 田 • ③• A •
C J K L M N 0 P 0 R 3 T U V W X ECX TO 01 JISUU Prom: 2 DATE MATERIAL TO ORDER IMATERIAL TO ORDER BETWEEN TWO DATES IRONDATE TO DATE TO DATE Material Description Material Description Material Description Material Description Material Description To Order Image: Description Material Description Materi	A1			
	A B	9 C	1	JK LM NOPQRSTUVWX-
	3		BACK TO M. MENU	
Automatical conduction of the second cond	4 5		CHITCO	
2 DATE MATERIAL TO ORDER MATERIAL TO ORDER DETWEEN TWO DATES ROM DATE TO DATE TO DATE TO DATE TO DATE	6 Fax:		ENTER	
A DE MATERIAL TO ORDER MATERIAL TO ORDER BETWEEN TWO DATES FROM DATE TO DATE TO DATE TO DATE TO DATE TO DATE TO Order TO O	7			
IMATERIAL TO ORDER BETWEEN TWO DATES TROM DATE Mail Mail Mail Code Image: Code	8	<u> </u>		
MATERIAL TO ORDER BETWEEN TWO DATES Bit Not. Material Description Material Index Material Description Material Description Material Index Material Description Material Description Material Description Index Material Description Material Description Material Description Inter Face (main menu / com@prouint / Mat.Pool / Act.Pool / Pro.Act / Quadified Of Act / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear Material Description NM Material Description Material Description Material Description Material Description	10			
ROMDATE TO DATE TO DATE TO DATE TO DATE TO Code Material Description Material To Order To Order To Order To T	11	MATERIAL TO ORDER BETWEEN TV	VO DATES	
Nate Naterial Description Material To Order Image: To Date Image: To Order Image: To Order Image: To Date Image: To Order Image: To Order Image: To Date Image: To Order Image: To Order Image: To Date Image: To Order Image: To Order Image: To Date Image: To Order Image: To Order Image: To Date Image: To Order Image: To Order Image: To Date Image: To Order Image: To Order Image: To Order Image: To Order Image: To Orde	12	FROMDATE		
Material Description Material To Order Image: Code Image: Code Image: Code Im	13	TODATE		
Code Material Description To Order To	Ma	at.	Material	
H Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / S	15 CO	de Material Description	To Order 💽	
Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Pro.Act / Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear & Not Material Com@pro.inf / Mat.Pool / Pro.Act / Qua.of Mat Of Ac / Mat Pool /	16			
Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Quaof Mat Of Ac / Mat No / Com@pro.inf / Mat.Pool / Pro.Act / Quaof Mat Of Act.Pool / Pro.A	18			
* * * * * * * * * * * * * * * * * * *	19			
Num All Other Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Curve Cal /	20			
A Model and the formation of the form	22			
A Minimum Component / Mat.Pool / Act.Pool / Pro.Act / Qua of Mat Of Ac / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear / Dual 1000000000000000000000000000000000000	23			
A Marcel End A Marcel End Ford Act Pool (Act Pool / Pro.Act / Qua.of Mat Of Act / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear <) >) >) >) >) >) >) >) >) >)	25			
A 1 Med A 20	26			
NUM	27			
 NUM Num 	29			
 + → N Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order) 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear Mum 	30			
A 1 A 20 A	32			
A 1 A 20 A	33			
1 → Pi Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear	34			
A 1 A 20 A	36			
A 14 A 20 A 2	37			
A 14 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A	39			
A 14 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A	40			
A 14 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A	41			
A 14 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A 2 A	43			
د به با / Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear الاسالة به المالية به بالاسلام المالية بي بي بالاسلام المالية بي	44			
۲ → ۲ / Inter Face / main menu / com@pro.inf / Mat.Pool / Act.Pool / Pro.Act / Qua.of Mat Of Ac / Mat To Order / 2Date Mat.To Order / Mat.Ass / Ear V. S-Curve Cal / Ear () کار ایجاد انفر	46			
A LA BALLE AND ALL AN	47			
تاهز المرابع المرابع	14 + H	\Inter Face / main menu / com@pro.inf / Mat.Pc	ol / Act.Pool / Pro.Act /	Qua.of Mat Of Ac 🖌 Mat To Order 👌 2Date Mat.To Order 🖉 Mat.Ass 🧹 Ear V. S-Curve Cal 🏑 Ear 🚺 💿 🔊
A 16 (2) 2, 0		NUM		تاهز
	e -9:10	339		الد سام 🖌 🙀 Microsoft Excel - Part 🔯 🛛 master 13 empty 😁 🚺 👬

Figure A3.9: a sample of "Materials to Order between Two Dates" sheet.



Figure A3.10: a sample of "Materials Spaces and Cash Requirements" sheet.



2										Microsoft Excel - Part1 🔳 🗗 🔀
1	ليمات	🚽 اكتب سؤالاً للتعا							يمات	🗙 🖻 _ 🐴 مِلْف تحِرير عِرض لِدراج تنسيق أدِوات بيانِات لِطَار تِعا
10) 📂 🕞	🔒 🛃 🚉 • 💞 🤊 • Σ • ≙↓ 🛛 %1	• - 0	A A	rial	1	- 1	BI	U I E E	🛯 🔤 💽 भा • 🛒 % । 🐄 🖧 🛊 📰 🖬 • 🔗 • 🗛 • 📘
	A1									
1	A C	D	E	F	G	н		J	К	L AS AT AU AV AW AX AY AZ
2	•									
3		· · · · · · · · · · · · · · · · · · ·		BACK TO	M. MENU	L				
5	Phone :			ENT	ER					
6	Fax :									
7						EARI	VED VA.	L. S-C	CURVE CA	LC.
8		PRO IECT SATRT DATE			1					
10		PROJECT SMIKI DATE								
11		ritorov rivisi bare							1	
12	Act. Code	Act Description	Unit	Duration	Satart Date	Finish Date	Quantity	Unit Price	Total	
13		-	-							
15										
16										
18					-		-			
19										
20			-							
22										
23										
24			5 8	6	-					
26										
27			_	<u></u>						
29										
30										
31			-							
33			1 8							
34	-		-							
36										
37					-					
38			-							
40										
41				-						
43										
44										· · · · · · · · · · · · · · · · · · ·
14 4	· · · ·	Inter Face / main menu / com@pro.inf / Mat.Pool	/ Act	Pool / Pro	Act / Q	ua.of Mat	Of Ac /	Mat To	Order / 2D	ate Mat. To Order 🖌 Mat. Ass 👌 Ear V. S-Curve Cal 🖉 Ear 🚺 🔊 📗
		NUM								جاهز
e	-4:11 🔞	080						ام	الرسا - 1۰ 👹	Microsoft Excel - Part1 🕱 🛛 🛛 master 13 empty 🗁 🦾 🛵 👭
									~	

Figure A3.11: a sample of "Earned value S-curve calculation" sheet.





Figure A3.12: a sample of "Earned value S-curve".



4							М	licrosoft Excel
	 اكتب سؤالاً للتعليمات 			بات	، إطار تعلي	أدوات بيانات	; تنسيق	رير <u>عر</u> ض إدراج
1			- 1	BIUE		वि भा 🚽 🔤	1%	*.0 .00 E E
			and the second s				- 91 Ke	100 000 1 -2 -2
1	B C D	E	G	H 1	J	K L	M	N O
C								de se de se
-								
-								
	Phone :	Enter						
	Fax:							
-	 LEAD TIME FOR ORDERING & DEL 	IVERY MATERIALS						
1	Enter Daily Materials Assignment Period	From Date ??						
	Not > 2 Month Period	To Date ??						
1								
	Enter Lead time for deliver mat. & Le	ead time for order mat						
-		Lead time for Lead	time					
	Material Name	deliver for o	ders					
	<u>.</u>							
	2							
9								
1	Mat.Days Before - O.S & P.O / Mat. Must Be Pu	rch. At 🖌 Mat. Spaces At P.	D.D / Mat. Prices	At P.O.D 🖌 comul	. Pr & Sp /P	. 0 / Mat. M	lust Be Or	n Site At 🖌
4			The summary sector is a state of the state o	the same water part of the last of the second second with the second second second second second second second				
•	NUM							

Figure A3.13: a sample of "Materials to order and deliver on site" sheet.





Figure A3.14: a sample of "Materials Must be purchased at' sheet.





Figure A3.15: a sample of "Materials Spaces Assignment" sheet.





Figure A3.16: a sample of 'Materials Prices Assignment" sheet.





Figure A3.17: a sample of "Materials Cumulative Spaces & Prices at P.O.D" sheet.




Figure A3.18: a sample of "Purchase order" sheet.





Figure A3.19: a sample of "Materials Must be on the site at" sheet.





Figure A3.20: a sample of "Materials Card" sheet.



26															Mic	rosoft Ex	cel - Pa	rt4 💶	
1	لأ للتعليمات	🚽 اکتب سؤا								فليمات	اِطْار ت	بيانات	<u>د</u> وات	سيق أ	راج تند	<u>عر</u> ض اٍد	ف ت <u>حرير</u>	🛯 مل	_ 8 ×
: 🗅	💕 🖬 🖪		• 🎯 🎝 • Σ • Å	%10 - 🕜 💾	Arial		-	17 - 1	BIU		콜콜	4	¶ + ⁰	P %	°, *.0		律 田・	- 👌 -	A - ,
1	D4	▼ fx B	ack To M. Menu														2		
1	A B		C		D	E	F	G	Н	<u> </u>	J	К	L	М	N	0	Р	Q	R
2																			-
3					Back	To M. M	enu	83											
5	Phone :						-												
7	Fax :	10 																	
8			WASTE CONTROL	6															
9																			
11		1									-	1					1		
12	Act.		Activity Description		Unit	Total	Total	Satart	Finish	No.	Act. Dates	Mat.1	total	to	Earned	Consumed	Waste %	Mat.2	total
13	Code					Dur.	Quan.	Date	Date			Name	Quan	order	Quantity	Quantity		Name	Quan
14												-							
16																			
17																			
19												1							
20											-								
22																			
23																			
25																			
20																			
28											-								
30											¢.								
31												-							
33																			
34												-							
36																			
37												-				-			
39																			
40											-								
42]							
43																			
45																			
46																			
48		to Control /							L,		-	ļ							~
114 4	P PI Was	te Control/							1.	s.)		U.							2
		9	1.01.2.19	n. u 1916		1		aî -		nun MA	í		Date	an I			-	5	جاهر
÷ *			tembre 14 M	Part4 🎫			에다 램			all2 🔡			Fall	뀀		master 13 er	npy 🖵	1	¥! 💶

Figure A3.21: a sample of "Waste control" sheet.



www.manaraa.com

A sample of Activities List



Act.	Activity Description	Unit
No.		
0100	General Works	
0101	Demolition Works	1.s
0200	Earth Work	
0201	Excavation	m3
0203	Fill with compacted clean sand	m3
0300	Concrete Works	
0301	Plain concrete B150	m3
0302	R. Concrete B300 for foundation	m3
0303	R. Concrete B250 for Ground beams and steps	m3
0304	R. Concrete B300 for Necks and Columns	m3
0305	R. Concrete B250 for Slaps 25 cm thick.	m3
0400	Block Works	
0401	Block works for walls 20cm thick.	m2
0402	Block works for internal walls 15cm thick.	m2
0403	Block works for internal walls 10cm thick.	m2
0500	Plastering Works	
0501	3-face internal plastering	m2
0502	External plastering on walls	m2
0600	Tilling Works	
0601	Pre cast terrazzo tiles 25*25 cm for floors	m2
0602	White ceramic tiles20*30cm for walls	m2
0603	White ceramic tiles 20*20cm for floor	m2
0700	Painting Works	
0701	Internal Painting-Poliside for Ceiling	m2
0702	Internal Painting-Supercryle for walls	m2
0800	Carpentry Works	
0900	Metal and Aluminum Works	
1000	Electrical Works	
1100	Mechanical Works	
1200	Isolation Works	
1300	Special Works	
1400	Finishing Works	

Figure A4.1 a sample of Activities List



System Evaluation Questionnaire (In Arabic)



استبيان تقيم برنامج الحاسوب

(CMMS) "Construction Materials Management Software"

السادة شركة /

نشكر لسيادتكم للمساهمة في أوقاتكم وجهودكم في تطبيق برنامج الحاسوب CMMS الخاص بإدارة مواد البناء في مشاريع التشييد والمقدم لسيادتكم لتجريبه في مشروع حقيقي. يرجى من سيادتكم تعبئة هذا الاستبيان، وذلك من أجل التحقق من فعالية البرنامج المذكور. إن رأيكم وملاحظ اتكم على البرنامج مهمة جدا.

ولكم جزيل الشكر

الباحث م. إياد عبد القادر الحداد



أولا/ التقنيات التي يوفرها البرنامج:

بين مدى موافقتك على التقنيات التي يوفر ها البرنامج والخاصة بإدارة مواد البناء في مشاريع التشييد لديكم:

موافق بدرجة ضعيفة جدا	موافق بدرجة ضعيفة	موافق بدرجة متوسطة	موافق	موافق جدا	التقنيات	الرقم
					تقرير عن المساحات المطلوبة لمواد المشروع في أي فترة زمنية.	1
			-		تقرير عن كميات المواد المطلوبة لأنشطة المشروع في أي فترة زمنية.	2
			-		تقرير عن تكاليف المواد المطلوبة لأنشطة المشروع في أي فترة زمنية.	3
					تقرير عن المواد المطلوب عمل أمر شراء لها في أية فترة زمنية.	4
					نقرير عن المواد التي يجب أن تتوفر في موقع العمل	5
					المنحني S-Curve للتكلفة الفعلية للمشروع" Earned Value"	6
					تقرير مراقبة الفاقد في المواد لكافة أنشطة المشروع.	7
					توفير بيان لكيفية توزيع كل مادة على مدار المشروع على الأنشطة المختلفة	8

المنسارات المستشارات

ثانيا/التصميم والتركيب:

بين مدى موافقتك على الخصائص والميزات التالية المتعلقة بتصميم وتركيب البرنامج.

موافق بدرجة ضعيفة جدا	موافق بدرجة ضعيفة	موافق بدرجة متوسطة	موافق	مو افق جدا	الخصائص/الميز ات	الرقم
					البرنامج مرن ويمكن تحديث البيانات فيه بسهولة.	1
					استخدامه بصورة عامة سهل	2
					يوفر الوقت والجهد	3
					طريقة إدخال البيانات سهلة وواضحة.	4
					التقارير والمخرجات واضحة وسهلة القراءة والفهم.	5
			-		طريقة فرز البيانات سهلة.	6
					الجمل والأرقام الواردة فيه مختصرة ومفيدة وحجمها مناسب ومقروءة.	7
					يمكن إستخراج المعلومات منه بسهولة.	8
					طريقة الاستخدام المرفقة مفهومة.	9
					يوفر سهولة وارتياح في التعامل معه لأنة يعمل ضمن بيئة Ms Excel.	10
					التدريب عليه سهل ولا يحتاج لوقت كبير ، كما انه لا يحتاج لموظف محترف للتعامل معه	11
					يمكن تطبيقه في معظم المشاريع بأنواعها في قطاع غزة (مناسب لمقاولي قطاع غزة).	12
					يساهم في تطوير إدارة مواد البناء في مشاريع التشييد في قطاع غزة.	13

المنسارات المستشارات

ثالثا / أذكر الصعوبات التي واجهتك في استخدام البرنامج.

رابعا / إذا كان لديك انتقادات أو تعليقات على البرنامج، يرجى ذكرها.

•••••	 	 	 	

خامسا / في رأيك، ما هي مميزات البرنامج.



System Evaluation Questionnaire (English Version)



Questionnaire about evaluation of

Construction Materials Management Software (CMMS)

Sire/.....

I strongly thank you for your contribution of time and effort to apply and test the CMMS software in a real project.

Please fill this questionnaire which aims to verify the construction materials management software (CMMS).

The researcher Eng. Eyad A.Q.haddad



First / The construction materials management tools and techniques

Clarify your extent of agree with performance with the performance of the followings construction materials management tools and techniques which are provided by the

CMMS software

No	Techniques	Strongly agree	Agree	Intermediately agree	Weakly agree	Very weakly agree
1	Needed storage area report for project materials at any					
-	date.					
2	Required quantities of materials report project activities					
_	at any date.					
3	Actual cost report for project activities at any date.					
4	Required materials report to be purchased at any date.					
5	Required materials, to be available on site.					
6	Earned value S-Curve.					
7	Waste monitoring report for activity materials.					
8	Provide a card for any material and its assignment on					
	projects` activities.					



Second / The design and structure

Clarify your extent of agreeing with the following features of CMMS design and structure.

No	Techniques	Strongly agree	Agree	Intermediately agree	Weakly agree	Very weakly agree
1	The software is flexible, and the data can be updated easily.					
2	In general, it is easy to use.					
3	It saves time and effort.					
4	Method of entering data is easy and clear.					
5	The reports and outputs are clear, and easy to read and understand.					
6	Method of sorting data is easy.					
7	Text and numbers shown are concise, and their sizes are suitable and readable.					
8	The information can be inquired easily.					
9	The method of use is understandable.					
10	It is easy to handle as it is developed within Excel environment.					
11	Training to use the CMMS is easy and it does not need much time. In addition, it does not need a professional user to deal with it.					
12	It can be applied for most of Gaza strip projects. (It is suitable for Gaza strip contractors).					
13	It contributes in improving the construction materials management practice in Gaza strip.					



Fourth / If you have any criticism or comment on the software, please state them.
Fifth / According to your opinion, what are the CMMS advantages?





Recording Incoming of Construction Materials Device



Figure A7: Recording incoming of construction materials device



147

The Average waste percentage of construction material in different countries



Table A8: The Average waste percentage of construction material in different countries.

	Palestine	Palestine	Palestine	Égypt	Australia	Korea	Netherlands	Hong	Kong	UK	USA
	Al	Enshassi	Dahman	Garas	Ekanayake	Chen	Bassink	Poon e	t Study	Poon	Chen
Material	Mogany	Study	Study	Study	and Ofrie	etal	and	(200	01a)	etal	etal
Wateria	study	(1996)	(2004)	(2003)	Study	Study	Brauwers			Study	Study
	(2006)				(2004)	(2002)	Study	Public	Privet	(2004)	(2002)
							(1996)				
Concrete	5.4	NA	4.7	4	NA	1.5	3	4-5	3-5	5	7.5
Steel	5.4	3.6	5.4	5	NA	NA	NA	1-8	3-5	5	NA
Reinforcement											
Timber Board	8.7	NA	10.4	13	13.8	16.7	NA	15	5	NA	10
Block	5.4	NA	4.6	6	NA	3	NA	4-8	6	NA	3.5
Tile	4.4	NA	3.8	5	NA	2.5	10	4-10	6-8	2	6.5
Cement	4.4	NA	5.3	5	19.6	NA	10	4-20	1-7	10	NA
Sand	10.5	NA	8.8	9	NA	NA	NA	NA	NA	NA	NA
Aggregate	8.9	NA	5.7	NA	NA	NA	NA	NA	NA	NA	NA
Paints	4	NA	NA	NA	NA	NA	NA	NA	NA	2	NA
Gypsum	4	NA	NA	NA	NA	NA	NA	6-10	5	NA	NA
Water Pipes	1.7	NA	NA	NA	NA	NA	NA	1-5	1	NA	NA
Sewage Pipes	2.1	NA	NA	NA	NA	NA	NA	1-5	2	NA	NA
Crushed Rock	8.5	NA	NA	NA	NA	NA	NA	NA	NA	NA	NA

Help and Method of Use



Help and Method of Use

1. الجزء الأول Input Data & Basic Information": Part 1 الجزء الأول 1

هذا الجزء مخصص لعمليات الإدخال الرئيسية وبعض الحسابات الأساسية، ويتكون من واجهة واجهة المشروع وعدد 11 واجهة عمل مقسمة إلى أربع واجهات عمل مخصصة للمدخلات و6 واجهات عمل للحسابات الأساسية على النحو التالي:

واجهات العمل وتنقسم إلى:

1.1 واجهة التحكم Main Menu ويتم من خلالها إدارة عملية الانتقال بين واجهات المصنف الواحد والمصنفات المختلفة. وهناك 4 واجهات مدخلات وهي على التوالي:

1.2 واجهات العمل الخاصة بالمدخلات:

:Company and Project Information واجهة 1.2.1

ويتم في هده الواجهة إدخال البيانات التالية:

- بيانات الشركة وتتكون من اسم الشركة، المدينة، رقم الهاتف، رقم الفاكس الخاص بها.
- بيانات ومعلومات عن المشروع وتتكون من اسم المشروع، رقم المشروع، مكان المشروع، كلفة المشروع الإجمالية، تاريخ بداية المشروع، تاريخ نهاية المشروع.

1.2.2 واجهة Materials Pool!

ويتم في هذه الواجهة إدخال البيانات التالية:

- أسماء جميع المواد الداخلة في جميع أنشطة المشروع لمرة واحدة في خانة Material Name مع الحرص على عدم تكرار المواد الداخلة في أكثر من نشاط وبحيث لا يتجاوز عدد المواد عن 75 مادة كحد أقصى. وكمثال على ذلك مادة الرمل تدخل في أعمال القصارة والبلاط فنقوم بذكر مادة الرمل فقط لمرة واحدة.
 - كود وحيد لكل مادة في خانة Material Code ويمكن أن يكون رقم تسلسلي 1,2,3.
- وحدة قياس كمية المادة في خانة Unit مع ملاحظة أنه يمكن اختيار الوحدة من قائمة منسدلة أو كتابتها يدويا أو نسخها من ملف أخر.
- سعر الوحدة من تلك المادة في خانة Unit Price ومثال على ذلك الرمل يقاس بوحدة m3 ، وسعر الوحدة m3 من الرمل يساوي \$ 3 .
- المساحة التي تحتاجها الوحدة من تلك المادة في خانة Unit Space، ومثال عي ذلك يحتاج كوب
 الرمل m3 إلى 0.5 m2 من أجل تخزينه.

1.2.3 واجهة Activity Pool

ويتم في هذه الواجهة إدخال البيانات التالية:



- اسم ووصف لكل الأنشطة الرئيسية الوحيدة في مشروع البناء ولمرة واحدة في خانة Activity
 اسم ووصف لكل الأنشطة الرئيسية الوحيدة في مشروع البناء ولمرة واحدة في خانة Description
 مع الحرص على ذكر الاسم العام للنشاط مرة واحدة حتى ولو تكرر في أكثر من طابق أو مكان. ومثال على ذلك مبني يتألف من ثلاث طوابق، ولدينا أعمال أعمدة دور أرضي، دور ثاني، ودور ثالث و هكذا. ففي هذه الحالة في خانة Activity Description نذكر فقط أعمال خرسانة ودور ثالغمدة.
- إدخال كود وحيد لكل نشاط في خانة Activity Code بحيث لا يشترك نشاطين أو أكثر بنفس الكود مع ملاحظة أنة من المفضل أن يكون الكود مؤلف من أربع خانات 4 digit . 4 234 الخانتين رقم 1و2 تستخدمان للنشاط الفرعي،ومثال على ذلك نشاط بعنوان أعمال صب أعمدة والمستخدم يرمز لأعمال الباطون بكود أساسي 02، وأعمال الأعمدة بكود فرعي 06، فيصبح الكود الخاص بهذه الفعالية 0206.
- وحدة قياس النشاط في خانة Unit مع ملاحظة أنه يمكن اختيار الوحدة من قائمة منسدلة أو كتابتها يدويا أو نسخها من ملف أخر.
- أسماء المواد الداخلة في إنجاز كل نشاط في الخانات , 2Mat. Mat.1 Name, Mat.1 Name, Mat.1 Name, . مع ملاحظة انه لا يمكن
 Name.5Mat. Name, 4Mat. Name, 3Mat. Name,
 إدخال أكثر من خمس مواد لكل نشاط ، ففي حالة زيادة المواد اللازمة لكل نشاط عن خمس مواد نقوم
 بتجزئة النشاط لأكثر من نشاط فر عي كمخرج لهذه النقطة. يمكن اختيار أسماء المواد من قائمة منسدلة
 أو كتابتها يدويا أو نسخها من ملف أخر.
- Mat.1 Qua./Unit, Mat.2 كميات المواد الداخلة في إنجاز الواحدة من كل نشاط في الخانات التالية Mat.5 Qua./Unit. Mat.4 Qua./Unit, Mat.3 Qua./Unit, Qua. /Unit,
 وفي أغلب الأحيان تكون هذه الكميات ثابتة لكافة أعمال البناء مع إمكانية التعديل عليها عند حدوث أي تغير.

1.2.4. واجهة Project Activity!

ويتم في هذه الواجهة إدخال البيانات التالية:

- اسم ووصف لكل أنشط المشروع الوحيدة أو المتكررة في أكثر من طابق أو مكان. ومثال على ذلك،
 أعمال أعمدة دور أرضي، أعمال سقف دور أرضي، أعمال أعمدة دور أول، أعمال سقف دور أول.
 يمكن نسخ هذه القيم من برنامج Microsoft Project أو كتابتها يدويا.
- كود لهذه الأنشطة في خانة Activity Code مع الأخذ بعين الاعتبار أن جميع الأنشطة المتشابهة تأخذ نفس الكود و هو كود اسم النشاط العام الذي أدخلناه في واجهة Activity Pool . ومثال على ذلك في صفحة Activity Pool:

Activity Cod.	Activity Description
0260	أعمال خرسانة الأعمدة.

في صفحة Project Activity في



Activity Cod.	Activity Description	
0260	أعمال خرسانة الأعمدة للدور الأرضي	
-	-	
-	-	
0260	أعمال خرسانة الأعمدة للدور الأول.	

- وحدة قياس كل نشاط في خانة Unit مع ملاحظة أنه يمكن اختيار الوحدة من قائمة منسدلة أو كتابتها يدويا أو نسخها من ملف أخر.
 - Duration مدة إنجاز كل نشاط بالأيام في خانة
 - تاريخ بداية النشاط في خانة Start Date.
 - تاريخ نهاية النشاط في خانة Finish Date.
 - كمية كل نشاط في خانة Quantity.
 - سعر الوحدة لكل نشاط في خانة Unit Price.
 - يتم حساب التكلفة الكلية لكل نشاط تلقائيا وبشكل أوتوماتيكي في خانة Total.
- وفي النهاية يتم إدخال تاريخ بداية المشروع وتاريخ نهاية المشروع في كلا من الخانتين التاليتين على التوالي Enter Start Project Date, Enter Finish Project Date.

1.3 واجهات الحسابات الأساسية:

Quantity OF Material of Activity واجهة 1.3.1

فيها يتم سرد تلقائي وبشكل أوتوماتيكي لأسماء كل الأنشطة المدخلة في واجهة Project Activity وأكوادها ووحدات قياسها وكمياتها وأسماء المواد الداخلة في كل نشاط بشكل تلقائي. ومن ثم يتم حساب الكميات الكلية للمواد الداخلة في كل نشاط بضرب كمية كل مادة لواحدة النشاط بالكمية الكلية للنشاط، ونحصل بالتالي من هذه الواجهة على أسماء المواد الداخلة في كل نشاط والكمية الكلية لهذه المواد في الكمية الكلية للنشاط.

1.3.2 واجهة Materials Needed

فيها يتم سرد تلقائي وبشكل أوتوماتيكي لأسماء كل الأنشطة الموجودة في واجهة Project Activity وأكوادها ووحدات قياسها ومدة كل نشاط، كمية كل نشاط، تاريخ بداية وتاريخ نهاية كل نشاط، عدد أيام النشاط، التاريخ في كل يوم. ثم يتم سرد أسماء المواد الداخلة في كل نشاط ، الكميات الكلية لكل مادة في الكمية الكلية لكل نشاط. ومن ثم يتم توزيع الكمية الكلية لكل مادة على عدد أيام النشاط بالتساوي.

2 Date Materials to Order: واجهة 1.3.3

يتم في هذه الواجهة سرد تلقائي وبشكل أوتوماتيكي لأسماء المواد الداخلة في واجهة Materials Pool، وأكوادها، ويطلب منك البرنامج إدخال تاريخين لحصر كميات كل مادة من المواد والتي ستستهلك نظريا وحسب ما هو مخطط له بين هذين التاريخين في كل الأنشطة.



Materials Spaces and Cash requirements واجهة 1.3.4

فيها يتم سرد تلقائي وبشكل ألي لأسماء الأنشطة المذكورة في Project Activity، أكوادها، أسماء المواد الداخلة في كل نشاط، كمياتها المستهلكة نظريا وحسب ما هو مخطط له بين التاريخين المدخلين في صفحة 2Date Materials to Order.

Earned Value S-Carve Calculations واجهة 1.3.5

هي نسخة مطابقة لواجهة Project Activity مع عمل حسابات للدفعات المستحقة من المشروع كل 30 يوم من بداية المشروع وحتى نهايته.

1.3.6 واجهة S-Curve Diagram

و هو عبارة عن رسم توضيحي للدفعات المستحقة المتر اكمة كل 30 يوم من بداية المشروع وحتى نهايته

2. الجزء الثاني Materials Purchase Decision": Part 2

يستخدم هذا الجزء لإدارة طلبيات الشراء للمواد خلال فترات محدده من حيث كميات المواد التي يجب تحرير طلبات شراؤها عند تواريخ محدده والموارد المالية اللازمة لشرائها والمساحات اللازمة لتخزينها عند هذه التواريخ المحددة ومن ثم تحرير طلبات شراء لفترات محددة وكذلك تحديد التواريخ التي يجب أن تتواجد فيها كل كمية المواد في طلبات الشراء في موقع العمل. يتكون هذا الجزء من 7 صفحات. صفحة مدخلات و 6 صفحات حسابات تلقائية

صفحة المدخلات

Lead time for ordering and delivery materials: واجهة 2.1

و فيها يتم سرد تلقائي وبشكل ألي لأسماء المواد الداخلة في جميع أنشطة المشروع و يتم إدخال التالي في هذه الواجهة:

- عدد الأيام التي يجب أن تكون المادة متواجدة قيها في الموقع قبل تاريخ استخدامها في خانة Days
 Before On Site
 - عدد الأيام التي يجب أن تطلب المواد قبلها حتى قبل تاريخ تواجدها في الموقع.
 - مثال:

Mat. Name	Lead time for deliver	Lead time for order
رمل	1	3

هذا يعني أن الرمل يجب أن تطلب أي كمية منه قبل 3 أيام حتى يتواجد في الموقع قبل يوم من تاريخ استخدامه

 إدخال التاريخين الذين نريد إدارة طلبيات شراء المواد خلالهما في خانتي From Date, To Date بحيث لا تتجاوز الفترة الشهرين.



ملاحظة: يتم تقسيم مدة المشروع المدخل في Part 1 في خانة Pro Act. بين تاريخي البداية و النهاية إلى فترات كل فترة مدتها شهرين في هذه الصفحة لإدارة عمليات طلبات الشراء فيها.

2.2 واجهات الحسابات التلقائية: وهي على التوالي

Materials must be purchased at: واجهة 2.2.1

و فيها يتم سرد تلقائي للبيانات التالية:

Trace Terms تحديد أول و أخر تاريخ طلب شراء للمواد في فترة الشهرين المحددة في صفحة Mat. Days
 Before O.S & P.O

	P.O.D For Materials Assignment	From Date ???	01/03/2006	
	Period Is	To Date ???	15/04/2006	
سماء	ر تاريخ طلب شراء يوم بيوم و سرد تلقائي لأ	، أول تاريخ طلب شراء و آخر	، سرد تلقائي للتواريخ بين	ثم يتم

كل المواد الداخلة في المشروع ثم يتم حساب كمية كل مادة يجب شراؤها عند كل تاريخ طلب شراء.

2.2.2 واجهة Materials Spaces assignments!

البيانات التالية يتم سردها بشكل تلقائى:

- سرد تلقائي لأول تاريخ طلب شراء و آخر تاريخ طلب شراء لفترة الشهرين.
- سرد تلقائي لأسماء كل المواد الداخلة في المشروع ثم حساب المساحة اللازمة لكمية كل مادة يجب شراؤها عند كل تاريخ طلب شراء.

Materials Prices assignments واجهة 2.2.3

البيانات التالية يتم سردها بشكل تلقائى:

- سرد تلقائي للتواريخ بين أول تاريخ طلب شراء لفترة الشهرين المدخلة في صفحة Material Days before On site & purchase order
- سرد تلقائي لأسماء كل المواد المدخلة في المشروع في حساب التكلفة اللازمة لشراء كمية كل مادة يجب شراؤها عند تاريخ طلب الشراء

: Cumulative Prices & Spaces واجهة 2.2.4

البيانات التالية يتم سردها بشكل تلقائى:

- سرد تلقائي للتواريخ بين أول تاريخ طلب شراء و اخر تاريخ طلب شراء لفترة الشهرين المدخلة في
 - .Mat days before On site & Purchase order صفحة
- سرد تلقائي للمساحة الكلية التي يجب أن تتوافر لتخزين كل المواد التي يجب شراؤها عند كل تاريخ طلب شراء.



- سرد تلقائي للتكلفة الكلية التي يجب أن تتوافر لشراء كل المواد التي يجب شراؤها عند كل تاريخ طلب شراء.
 - المساحة التراكمية اللازمة عند كل تاريخ طاب شراء.
 - التكلفة التراكمية اللازمة عند كل تاريخ طلب شراء.

Purchase Order واجهة 2.2.5

و فيها يتم تقسيم الفترة الزمنية" الشهرين" والتي تم إدخالها في واجهة

Material Days before on site & purchase order إلي عدة فترات زمنية لطلب المواد خلالها. ثم يطلب منا البرنامج تحديد المدة الزمنية التي نريد تحرير طلبية الشراء للمواد خلالها بحيث تكون جزء من الفترة الزمنية الشهرين المحددة بشكل مسبق. ثم يقوم البرنامج بسرد أسماء وكميات المواد المطلوب شراؤها في هذه الفترة.

:Materials Must Be On Site At واجهة 2.2.6

البيانات التالية يتم سردها بشكل تلقائي:

- تحديد التواريخ التي يجب أن تكون فيه المواد في الموقع قبل استخدامها لفترة الشهرين المحددة في واجهة Mat days before On site & Purchase order
 - المواد في الموقع قبل استخدامها لفترة الشهرين نفسها في خانتي
- حساب كمية كل مادة يجب ان تتواجد في موقع المشروع عند كل تاريخ من تواريخ الفترة الزمنية

المحددة في الصفحة Material Days before On site & purchase order.

: Waste Control" Part 3 الجزء الثالث .3

ويستخدم هذا الجزء لإدارة الفاقد الحقيقي والفعلي لمواد البناء المستخدمة في المشروع، ويتم سرد تلقائي للبيانات التالية:

- وصف لجميع أنشطة المشروع، كود كل نشاط، وحدة القياس لكل نشاط، كمية كل نشاط، مدة كل نشاط، تاريخ بداية ونهاية النشاط، عدد الأيام لكل نشاط.
 - سرد تلقائي لجميع المواد وكمياتها الداخلة في تنفيذ كل نشاط.
 - توزيع كمية كل مادة من مواد النشاط على عدد أيام النشاط بالتساوي.

أما البيانات التي يجب إدخالها في هذا الجزء هي على النحو التالي:

الكمية المنفذة فعليا لكل مادة من مواد النشاط في كل يوم من أيام هذا النشاط وتوضع في خانة Earned
 quantity

Earned quantity = Today quantity – Yesterday quantity



Yesterday quantity الكميات المنفذة فعليا لغاية اليوم السابق. Today quantity الكميات المنفذة فعليا حتى نهاية هذا اليوم.

الكمية المستهلكة فعليا من كل مادة من مخازن الموقع واللازمة لإنجاز النشاط وتوضع في خانة
 Consumed quantity . ويتم حساب هذه القيمة بشكل يدوي بالمعادلة التالية:

Consumed quantity = quantity on site (Q) + IN quantity – Left quantity

- Quantity on site (Q)
 - IN quantity الكمية الداخلة لموقع المشروع في هذا اليوم.
 - د التي غادرت الموقع في هذا اليوم.

بعد ذلك يقوم البرنامج بحساب نسبة الفاقد %Waste Percentage الفعلية بالشكل التالي:

Waste % = (Consumed quantity – Earned quantity) X 100 / Errand quantity

على المستخدم مقارنة النسبة الفعلية مع النسبة النظرية للفاقد والتي يمكن أن نجدها في المراجع ولدى ذوي الخبرة في هذا المجال. الملحق رقم 8 يوضح بعض النسب النظرية للفاقد في عدة دول.

4. الجزء الرابع "Materials Control "

ويتم في هذا الجزء تحديد كميات وتواريخ استخدام كل مادة من مواد المشروع لكل نشاط من أنشطة المشروع. حيث يقوم المستخدم باختيار اسم المادة من خانة Choose Material Name ومن ثم يقوم البرنامج بإظهار تلقائي للبيانات التالية:

- أسماء الأنشطة التي تدخل المادة المختارة في انجاز ها.
 - تواريخ استخدام هذه المادة في كل نشاط.
 - كمية استهلاك هذه المادة عند كل تاريخ.